SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549 -----FORM 8-K CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 Date of Report (Date of earliest event reported) JULY 24, 2000 PHILLIPS-VAN HEUSEN CORPORATION (Exact name of registrant as specified in its charter) DELAWARE (State or other jurisdiction of incorporation) 1-724 -----(Commission File Number) 13-1166910 (IRS Employer Identification No.) 200 MADISON AVENUE, NEW YORK, NEW YORK 10016 _____ (Address of principal executive offices) (Zip Code) (212) 381-3500 -----(Registrant's telephone number, including area code)

Item 2. Acquisition or Disposition of Assets.

On July 24, 2000, Phillips-Van Heusen Corporation, a Delaware corporation ("PVH"), Cluett, Peabody & Co., Inc., a Delaware corporation ("CP"), and Cluett Peabody Resources Corporation, a Delaware corporation executed and delivered the Trademark License Agreement, dated July 24, 2000 (the "License Agreement"), pursuant to which, CP granted to PVH an exclusive license to use trademarks related to CP's "Arrow" clothing line in the United States and its territories and possessions in connection with PVH's manufacture, advertising, marketing, promotion, distribution, offer to sell and sale of men's and boys' dress shirts, sports shirts (including knit, woven and all forms of fleece), pants, shorts and sweaters in certain specified channels of trade, including the Internet. Pursuant to the License Agreement, CP also granted to PVH the right to operate websites that use URLs that incorporate the licensed trademarks. The initial term of the License Agreement expires on June 30, 2007 and PVH has the option to renew the License Agreement for two additional five year terms, provided it meets certain conditions.

In connection with the License Agreement, on July 24, 2000, the transactions contemplated by the Purchase and Sale Agreement (the "Purchase and Sale Agreement") among PVH, CP, Cluett, Peabody Canada, Inc., a Canadian corporation ("CP Canada"), Arrow Factory Stores Inc., a Delaware corporation ("AFS"), Cluett Designer Group, Inc., a Delaware corporation ("CDG"), Consumer Direct Corporation, a Delaware corporation ("CDC"), and Cluett, Peabody Holding Corp., a Delaware corporation ("Holding"), and the Share Purchase Agreement (the "Share Purchase Agreement" and, together with the Purchase and Sale Agreement, the "Agreements") among PVH CP, CDG and Bidermann Tailored Clothing Inc., a Delaware corporation ("BTC"), were closed. Each of the parties to the Agreements other than PVH and CDG is, and prior to the closing CDG was, a wholly-owned subsidiary of Cluett American Corp. ("Cluett American"). Copies of the Agreements are attached as exhibits hereto and are incorporated herein by this reference.

Pursuant to the Share Purchase Agreement, all of the outstanding shares of capital stock of CDG were sold by CP and BTC to PVH. The assets of CDG principally consist of the license for the design, sourcing and distribution in the United States of dress shirts bearing the Kenneth Cole trademarks and CDG's existing inventory of such shirts. Pursuant to the Purchase and Sale Agreement, (i) all of the outstanding shares of capital stock of C.A.T. Industrial S.A. de C.V., a corporation organized under the laws of Honduras ("CAT"), were sold to PVH by CP, AFS, CDG, CDC and Holding, (ii) PVH purchased all of CP's and AFS's right, title and interest in and to certain of the other assets (principally inventory) of CP and AFS used primarily by CP and AFS in their Arrow label and private label men's and boy's dress and sport shirts operations and their Arrow factory outlet operations in the United States and (iii) PVH purchased all of CP Canada's right, title and interest in and to Arrow label dress and sport shirt inventory used in CP Canada's United States "Career Apparel" operations. CAT operates a shirt manufacturing plant. PVH intends to continue operating the plant for the same purpose

The aggregate consideration paid by PVH pursuant to the Purchase and Sale Agreement and the Share Purchase Agreement was approximately \$46.9 million in cash, calculated as set forth in, and subject to adjustment pursuant to, Article 3 of each of the Purchase and Sale Agreement and the Share Purchase Agreement. PVH used available cash and the proceeds of loans under its existing credit facility to complete these transactions. Cluett American has irrevocably and unconditionally guaranteed the payment of any amount payable by CP or BTC to PVH under each Agreement which is not paid by CP or BTC when due, including pursuant to the purchase price adjustment and indemnification provisions of the Agreements.

Item 7. Financial Statements, Pro Forma Financial Information and Exhibits

(a) Financial statements of businesses acquired.

The required financial statements are not included herein and will be filed within 60 days of the filing of this Report, as permitted by Item 7.

(b) Pro forma financial information.

The required PRO FORMA information is not included herein and will be filed within 60 days of the filing of this Report, as permitted by Item 7.

(c) Exhibits

- 1. Purchase and Sale Agreement among Cluett, Peabody & Co., Inc., Cluett, Peabody Canada, Inc., Arrow Factory Stores Inc., Cluett Designer Group Inc., Consumer Direct Corporation, Cluett, Peabody Holding Corp. and Phillips-Van Heusen Corporation, dated as June 12, 2000
- 2. Share Purchase Agreement among Cluett, Peabody & Co., Inc., Cluett Designer Group Inc., Bidermann Tailored Clothing Inc. and Phillips-Van Heusen, dated as of June 28, 2000.
- 3. Joint Press Release issued by Phillips-Van Heusen Corporation and Cluett American Corp., dated July 24, 2000, announcing completion of the licensing of the U.S. Arrow trademark and the sale of Cluett Designer Group Inc. to Phillips-Van Heusen Corporation.

The exhibits and schedules to the agreements attached as Exhibits to this Report are described in the Table of Contents of the applicable agreement and have been omitted from this Report pursuant to Item 601(b)(2) of Regulation S-K. Registrant agrees to furnish supplementally to the Securities and Exchange Commission a copy of any exhibit or schedule to either of the above-listed agreements upon request by the Securities and Exchange Commission.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

PHILLIPS-VAN HEUSEN CORPORATION

By: /s/ EMANUEL CHIRICO

Emanuel Chirico, Executive Vice President and Chief Financial Officer

Dated: August 8, 2000

PURCHASE AND SALE AGREEMENT

DATED AS OF JUNE 12, 2000

among

CLUETT, PEABODY & CO., INC.,

CLUETT, PEABODY CANADA, INC.,

ARROW FACTORY STORES INC.,

CLUETT DESIGNER GROUP, INC.,

CONSUMER DIRECT CORPORATION,

CLUETT, PEABODY HOLDING CORP.,

and

PHILLIPS-VAN HEUSEN CORPORATION

TABLE OF CONTENTS

		Payt	;
ARTICLE		NS	,
	1.1	Definitions	
ARTICLE	2.		
	PURCHASE	AND SALE OF ASSETS	5
	2.1	Assets to be Transferred	
	2.2	Excluded Assets	ò
	2.3	Shares to be Transferred	7
	2.4	Other Transactions	7
	2.5	Assignment of Assets	3
	2.6	Transfer and Delivery of the Purchased Shares	3
	2.7	Obtaining Permits and Licenses	3
ARTICLE	3.		
	CONSIDERA	TION)
	3.1	Purchase Price)
	3.2	Inventory	
	3.3	CAT Purchase Price Adjustment	
	3.4	Allocation of Purchase Price13	3
ARTICLE	4.		
	LIABILITI	ES, OBLIGATIONS AND INDEMNITIES14	1
	4.1	Liabilities and Indemnity	
	4.2	Indemnification Procedure	7
	4.3	Other Indemnification Provisions18	3
ARTICLE	5.		
	EMPLOYEES	AND EMPLOYEE BENEFITS)
	E 1	Employee Polations and Ponofits	٠

ARTICLE 6.

ARTICLE

ARTICLE

REPRESENT	ATIONS AND WARRANTIES OF CP
6.1	Organization and Corporate Power22
6.2	Due Authorization; No Breach
6.3	Real Property
6.4	Personal Property24
6.5	Title and Condition of Assets other than Real Property24
6.6	Consents
6.7	Compliance With Laws
6.8	Permits and Licenses
6.9	Environmental Matters
6.10	Contracts
6.11	Tax Matters
6.12	CAT Stock
6.13	Labor Matters
6.14	Finders; Brokers
6.15	Litigation, Claims and Proceedings28
6.16	Affiliate Transactions
6.17	1999 CAT Agreement
6.18	Survival of Representations and Warranties29
7.	
	TATIONS AND WARRANTIES OF BUYER
	ATIONS AND WARRANTIES OF BUYER
REPRESENT	Corporate Existence29
REPRESENT	
REPRESENT 7.1 7.2	Corporate Existence
REPRESENT 7.1 7.2 7.3	Corporate Existence
REPRESENT 7.1 7.2 7.3 7.4	Corporate Existence.29Due Authorization; No Breach.30Consents.30Finders; Brokers.31Availability of Funds.31
REPRESENT 7.1 7.2 7.3 7.4 7.6	Corporate Existence. 29 Due Authorization; No Breach. 30 Consents. 30 Finders; Brokers. 31
REPRESENT 7.1 7.2 7.3 7.4 7.6 7.7	Corporate Existence.29Due Authorization; No Breach.30Consents.30Finders; Brokers.31Availability of Funds.31No Other Representations or Warranties.31
REPRESENT 7.1 7.2 7.3 7.4 7.6 7.7 7.8	Corporate Existence.29Due Authorization; No Breach.30Consents.30Finders; Brokers.31Availability of Funds.31No Other Representations or Warranties.31Survival of Representations and Warranties.31
REPRESENT 7.1 7.2 7.3 7.4 7.6 7.7 7.8 8. COVENANTS	Corporate Existence
REPRESENT 7.1 7.2 7.3 7.4 7.6 7.7 7.8 8. COVENANTS 8.1	Corporate Existence. 29 Due Authorization; No Breach. 30 Consents. 30 Finders; Brokers. 31 Availability of Funds. 31 No Other Representations or Warranties. 31 Survival of Representations and Warranties. 31 Conduct of the Business. 32
REPRESENT 7.1 7.2 7.3 7.4 7.6 7.7 7.8 8. COVENANTS 8.1 8.2	Corporate Existence
REPRESENT 7.1 7.2 7.3 7.4 7.6 7.7 7.8 8. COVENANTS 8.1 8.2 8.3	Corporate Existence
REPRESENT 7.1 7.2 7.3 7.4 7.6 7.7 7.8 8. COVENANTS 8.1 8.2	Corporate Existence

	8.6 8.7 8.8 8.9 8.10 8.11 8.12 8.13	Letter of Credit 33 Accounts Receivable 34 1999 CAT Agreement 35 Returned Inventory 35 U.S. AFS Cash Registers 36 Break-Up Fee 36 Cluett Designer Group 36 Trademark License Agreement 37
ARTICLE	9.	
	CONDITIONS 9.1 9.2 9.3	S OF CLOSING
ARTICLE	10.	
	10.1 10.2	ON; SURVIVAL
ARTICLE	11.	
	CLOSING 11.1 11.2 11.3 11.4	Closing
ARTICLE	12.	
	EXPENSES / 12.1 12.2 12.3	AND POST CLOSING OBLIGATIONS
ARTICLE	13.	
	BULK SALES	S LAW

Page

		P -	age
	13.1	Waiver	.44
ARTICLE	14.		
	14.1	CONFIDENTIALITY	.44
	14.2	Confidentiality	.44
ARTICLE	15.		
	NOTICES		. 45
	15.1	Notices	. 45
ARTICLE	16.		
	MISCELLANE	:0US	. 46
		Binding Effect; Assignment	
		Exhibits and Schedules	
	16.3	Specific Performance	. 47
		Counterparts	
		Headings; Interpretation	
		Waiver	
		Severability	
		Governing Law and Forum	

EXHIBITS

v

SCHEDULES

Schedule 2.1(a)	-	•
Schedule 2.1(c)	-	Staple Dress Shirts
Schedule 2.1(d)	-	Contracts
Schedule 3.2(a)	-	Average Sales Price of Goods Sold at Discounted Prices
Schedule 3.3	-	Working Capital Closing Date Statement
Schedule 4.1(a)	-	Material Assumed Liabilities
Schedule 5.1(b)		
Schedule 6.5 ´		
Schedule 6.6	-	CP Required Consents
Schedule 6.7	-	
Schedule 6.8	-	Permits and Licenses
Schedule 6.9	-	Environmental Matters
Schedule 6.10	-	Contracts in Default
Schedule 6.12	-	CAT Stock
Schedule 6.13	-	Employment Contracts
Schedule 6.15	-	Litigation, Claims and Proceedings
Schedule 6.16	-	
Schedule 7.2	-	Due Authorization; No Breach
Schedule 7.3	-	Buyer's Required Consents
		.,

PURCHASE AND SALE AGREEMENT

This Purchase and Sale Agreement (this "AGREEMENT"), dated as of June 12, 2000, is among CLUETT, PEABODY & CO., INC., a Delaware corporation ("CP"), CLUETT, PEABODY CANADA, INC., a Canadian corporation ("CP CANADA"), ARROW FACTORY STORES INC., a Delaware corporation ("FS"), CLUETT DESIGNER GROUP, INC., a Delaware corporation ("CDG"), CONSUMER DIRECT CORPORATION, a Delaware corporation ("CDC"), CLUETT, PEABODY HOLDING CORP., a Delaware corporation ("CP HOLDING" and, collectively with FS, CDG and CDC, the "MINORITY CAT SHAREHOLDERS") and PHILLIPS-VAN HEUSEN CORPORATION, a Delaware corporation ("BUYER"). CP, FS, CP Canada and Buyer are referred to collectively as the "PARTIES," and individually as a "PARTY."

WITNESSETH:

WHEREAS, CP and FS are engaged, among other things, in the business of manufacturing and selling dress shirts and sportswear with the Arrow label and private label dress and sport shirts and in the operation of retail stores in the United States, such stores (as opposed to the corporate operations in support thereof) collectively referred to as "U.S. ARROW FACTORY STORES" or "U.S. AFS", and CP Canada is engaged, among other things, in the business of selling in the United States apparel in the "Career Apparel" channel of distribution (I.E., apparel bearing the name or logo of corporate entities or other organizations) ("U.S. CAREER APPAREL"). The aforementioned businesses of CP, FS and CP Canada are collectively referred to as the "U.S. BUSINESS";

WHEREAS, CP and the Minority CAT Shareholders own all of the outstanding shares of C.A.T. Industrial S.A. de C.V., a corporation organized under the laws of Honduras ("CAT") which is engaged in the manufacture of mens' apparel (the "HONDURAN BUSINESS") at a factory in Honduras (the "HONDURAN FACILITY") leased by CAT (the "HONDURAN FACILITY LEASE");

WHEREAS, concurrently with this Agreement, CP, Buyer and Cluett Peabody Resources Corporation, a Delaware corporation ("RESOURCES"), are entering into the Trademark License Agreement (defined below);

WHEREAS, CP Canada owns the U.S. Career Apparel inventory being purchased by Buyer hereunder;

WHEREAS, Buyer wishes to purchase, and CP and the Minority CAT Shareholders are willing to sell, all of the outstanding shares of capital stock of CAT (the "CAT SHARES") on the terms and conditions of this Agreement; and

WHEREAS, Buyer wishes to purchase, and CP, CP Canada and FS are willing to sell, all of CP's and FS's right, title and interest in and to certain of the other assets of CP and FS used primarily by CP and FS in the conduct of the U.S. Business and all of CP Canada's right, title and interest in and to the U.S. Career Apparel inventory owned by CP Canada, and as part of such purchase and sale of such assets, Buyer is willing to assume certain of the obligations and liabilities of CP, CP Canada and FS related to such assets, all in the manner and subject to the terms set forth herein.

NOW, THEREFORE, for and in consideration of the mutual covenants and agreements hereinafter set forth, the parties hereby agree as follows:

ARTICLE 1.

DEFINITIONS

1.1 DEFINITIONS. The following terms are defined in the Section indicated:

Term	Section
1999 CAT Agreement	8.4
Agreement	Introduction
Allocation	3.4(a)
Alternative Proposal	8.3
•	
Amount of Royalties	8.13(a)
Ancillary Agreements	2.4
Arbiter	3.2(d)
Arrow Inventory	2.1(c)
Assets	2.1
Assignment and Assumption Instruments	11.2(b)
Assumed Liabilities	4.1(a) and 4.1(b)
BTC	8.12
Buyer	Introduction
Buyer Indemnified Parties	4.1(b)
Buyer Plans	5.1(c)
Buyer's 401(k) Plan	5.1(h)
CAT	Recitals
CAT Employees	5.1(a)
CAT Shares	Recitals
on onares	NCOTERTS

Term	Section
CDC CDG Asset Purchase Agreement CDG Share Purchase Agreement Claim Closing Closing Date Code Confidentiality Agreement Contingent Period Contracts Corporate Employees CP CP Canada CP Employees CP Holding CP Indemnified Parties CP's 401(k) Plan CP's Cost CP's Knowledge Current Arrow Inventory Damages Encumbrances Environmental Claim Environmental Laws Escrow Agreement Estimated Inventory Purchase Price Excepted Event Excluded Assets Excluded Claims Excluded Claims Excluded Liabilities Fair Market Value Final Inventory Purchase Price FS	Introduction 8.12 8.12 4.2 11.1 11.1 3.4(a) 14.2 5.1(a) 5.1(a) Introduction Introduction 5.1(a) Introduction 4.1(c) 5.1(g) 3.2(a) 6.3(c) 2.1(c) 4.1(b) 2.1 6.9(b) 6.9(b) 6.9(d) 2.4(e) 3.2(a) 8.13(a) 2.2 6.15 4.1(e) 3.2(a) 3.2(d) Introduction

Term	Section
Hazardous Materials Honduran Business Honduran Environmental Laws Honduran Facility Honduran Facility Lease Hosiery Agreement HSR Act Indemnified Party Indemnifying Party Interests Inventory Closing Date Statement Inventory Escrow IRS Issuing Bank Items July 9 Count KCPL KCPL Shirt License Agreement Liability Minority CAT Shareholders Non-Current Arrow Inventory Non-Retained Employees Party Pension Plan Permits Pre-Closing Mervyn's Event Pre-Closing Sears Event Purchase Price Real Property Release Resources	6.9(b) Recitals 6.9(d) Recitals 8.9(d) Recitals 8.12 6.6 4.2 4.2 2.5 3.2(b) 3.2(a) 3.4(a) 8.6 16.2 3.2(c) 8.12 8.12 4.1(a) Introduction 2.1(c) 5.1(a) Introduction 5.1(f) 6.8 8.13(a)
Retained Employees Returned Inventory Sales Plans	5.1(a) 8.9
Sales Figure	5.1(i)

Term	Section
Securities Act	7.5
Sourcing Commitments	8.6
Tax	6.11(d)
Tax Authority	6.11(d)
Tax Returns	6.11(d)
Termination Costs	5.1(a)
Trademark License Agreement	2.4(b)
Transferred Employees	5.1(a)
Transitional Employees	5.1(b)
Trim	3.2(a)
U.S. AFS	Recitals
U.S. AFS Employees	5.1(a)
U.S. AFS FF&E	2.1(b)
U.S. Arrow Factory Stores	Recitals
U.S. Business	Recitals
U.S. Career Apparel	Recitals
U.S. Environmental Laws	6.9(a)
WARN	5.1(j)
Working Capital Closing Date Statement	3.3(a)

ARTICLE 2.

PURCHASE AND SALE OF ASSETS

2.1 ASSETS TO BE TRANSFERRED. Subject to the terms and conditions of this Agreement, and except as otherwise expressly provided in Section 2.5 hereof, at the Closing (as defined in Section 11.1 hereof), CP, CP Canada and FS shall sell, assign, transfer, convey and deliver to Buyer, and Buyer shall purchase, acquire and accept from CP, CP Canada and FS, all of CP's, CP Canada's and FS's right, title and interest in and to each of the following assets used by CP, CP Canada and FS in the conduct of the U.S. Business, in each case except as otherwise expressly provided in Section 2.2 hereof, free and clear of all mortgages, liens, pledges, security interests, charges, claims, restrictions and encumbrances of any nature whatsoever ("ENCUMBRANCES"), with such changes, deletions or additions thereto as may occur from the date hereof to the Closing in accordance with the terms and conditions of this Agreement and any Ancillary Agreement (as defined herein) executed in connection herewith (collectively, the "ASSETS"):

- (a) all U.S. Arrow Factory Store leasehold interests in real property leased by CP or FS and set forth in Schedule 2.1(a), including all buildings, structures and other improvements located on such real property and any additions, fixtures, replacements and alterations thereto between the date hereof and the Closing Date;
- (b) all tangible personal property and interests therein located in the U.S. Arrow Factory Stores, including all furniture, fixtures and equipment, but not including any computer software (the "U.S. AFS FF&E");
- (c) all raw materials, work in process, Trim and finished goods inventory relating to the U.S. Business (including, without limitation, the private label, U.S. Career Apparel and U.S. AFS businesses) that (i) in the case of staple dress shirts (as set forth in Schedule 2.1(c)) and U.S. Career Apparel inventory, can reasonably be expected to be sold within the next nine months based on prior sales history over the previous twelve (12) months of similar products adjusted for current customers' sales plans and (ii) in the case of fancy dress shirts and sportswear, relate to seasons subsequent to Spring 2000 and goods for open orders for the Spring 2000 season, in each case including, without limitation, any of such items in transit from manufacturing facilities of CP and its affiliates or third parties or warehouses of CP and its affiliates (such finished goods inventory, the "CURRENT ARROW INVENTORY") and non-current raw materials, non-current work in progress, non-current Trim and non-current finished goods inventory relating to the U.S. Business, including, without limitation, any of such items in transit from manufacturing facilities of CP and its affiliates or third parties or warehouses of CP and its affiliates (such finished goods inventory, the "NON-CURRENT ARROW INVENTORY" and, together with the Current Arrow Inventory, the "ARROW INVENTORY");
- (d) all rights under all material contracts, leases, licenses, commitments, sales orders, purchase orders, invoices and other agreements relating to the U.S. Business (the "CONTRACTS") as listed on Schedule 2.1(d); and
- (e) originals or copies of all records, files, supplier lists and other relevant data relating to the Assets, the operations of the U.S. Business that support the Assets and the Assumed Liabilities, provided that CP, CP Canada and FS shall retain originals of any such items which relate primarily to the Excluded Assets or the Excluded Liabilities.
- 2.2 EXCLUDED ASSETS. The parties to this Agreement expressly understand and agree that notwithstanding anything herein or in the Ancillary Agreements to the contrary, CP, CP Canada and FS are not hereunder selling, assigning, transferring or conveying to Buyer, and the Assets shall not include, any of the assets, rights and properties of CP, CP Canada or FS not expressly set forth in Section 2.1 (the "EXCLUDED ASSETS").

- 2.3 SHARES TO BE TRANSFERRED. Subject to the terms and conditions of this Agreement, at the Closing (as defined in Section 11.1 hereof), CP and the Minority CAT Shareholders shall sell, assign, transfer, convey and deliver to Buyer and/or its designees free and clear of all Encumbrances, and Buyer shall purchase, acquire and accept the CAT Shares and all of the certificates representing the CAT Shares, and such transfer shall be registered in the CAT shareholders' registration book.
- 2.4 OTHER TRANSACTIONS. In addition to the transactions contemplated above, the following acts or transactions shall also occur at or before the Closing:
 - (a) Buyer, FS and CP each shall execute and deliver a Transition Services Agreement substantially in the form attached hereto as Exhibit 2.4(a);
 - (b) Buyer, Resources and CP each shall execute and deliver a Trademark License Agreement substantially in the form attached hereto as Exhibit 2.4(b), as such form is revised pursuant to Section 8.13 hereto (the "TRADEMARK LICENSE AGREEMENT"), and all documents required thereby;
 - (c) FS, CP Canada and CP each shall execute and deliver a Bill of Sale and Assignment substantially in the form attached hereto as Exhibit 2.4(c);
 - (d) Buyer shall execute and deliver an Assumption Agreement substantially in the form attached hereto as Exhibit 2.4(d);
 - (e) Buyer, CP and The Bank of New York, as escrow agent, shall each execute and deliver an Escrow Agreement substantially in the form attached hereto as Exhibit 2.4(e) (the "ESCROW AGREEMENT");
 - (f) CP, CP Canada and FS shall deliver UCC-3 partial releases and/or termination statements and such other instruments as are required to release all Encumbrances of the lenders and other creditors of CP, CP Canada and FS on the Assets; and
 - (g) CP shall execute and deliver, and shall use its reasonable efforts to cause its subcontractors, contractors and warehousemen to deliver, Waivers and Consents substantially in the form attached hereto as Exhibit 2.4(g).

The agreements set forth in paragraphs (a) through (g) of this Section 2.4 are hereinafter referred to as the "ANCILLARY AGREEMENTS."

2.5 ASSIGNMENT OF ASSETS. (a) To the extent that any Asset to be sold, assigned, transferred or conveyed to Buyer, or any claim, right or benefit arising thereunder or

resulting therefrom (the "INTERESTS"), cannot be sold, assigned, transferred or conveyed without the approval, consent or waiver of, or filing with, the issuer thereof or the other party thereto or any third person (including a government or governmental unit), the Parties will cooperate jointly to obtain such approval, consent or waiver or make such filing. To the extent that such sale, assignment, transfer or conveyance or attempted sale, assignment, transfer or conveyance of any such Interest would constitute a breach thereof or a violation of any law, decree, order, regulation or other governmental edict, this Agreement shall not constitute a sale, assignment, transfer or conveyance thereof or an attempted sale, assignment, transfer or conveyance thereof.

- (b) Notwithstanding anything to the contrary contained herein, CP, CP Canada and FS are not obligated to sell, assign, transfer or convey to Buyer, and Buyer is not obligated to purchase hereunder, any of CP's, CP Canada's and FS's rights and obligations in and to an Interest without first having the necessary approvals, filings, consents or waivers required to effect the sale, assignment, transfer or conveyance of such Interest.
- (c) To the extent that any of the approvals, consents, filings, waivers, or other authorizations referred to in Section 2.5(a) hereof have not been obtained as of the Closing, CP, CP Canada or FS, as the case may be, shall, during the remaining term of such Interest, use all reasonable efforts, to (i) obtain the consent of any such third person (including a government or governmental unit), (ii) cooperate with Buyer in any reasonable and lawful arrangement designed to provide the benefits of such Interest to Buyer and (iii) enforce, at the request of Buyer and for the benefit of Buyer, any rights of CP, CP Canada or FS arising from such Interest against such issuer thereof or the other party or parties thereto (including the right to elect to terminate any such Interest in accordance with the terms thereof upon the written request of Buyer). The out-of-pocket expenses incurred by CP, CP Canada and FS and Buyer in connection with any such efforts pursuant to this Section 2.5(c) shall be shared equally by CP and Buyer.
- 2.6 TRANSFER AND DELIVERY OF THE PURCHASED SHARES. CP and the Minority CAT Shareholders shall transfer and deliver to Buyer share certificates representing the CAT Shares duly endorsed, or accompanied by stock powers duly executed, with all necessary stock transfer stamps attached thereto and canceled, or such other assignments, deeds, share transfer forms, endorsements, notarial deeds of transfer or other instruments or documents, duly stamped where necessary, as required by the jurisdiction of organization of CAT. CP and the Minority CAT Shareholders hereby waive any preferential right they may have under applicable law to acquire the CAT Shares owned by any other person.
- 2.7 OBTAINING PERMITS AND LICENSES. Buyer shall use all reasonable efforts to obtain as of the Closing or as soon thereafter as may be practicable all permits and licenses required by any governmental agency with respect to the Assets. CP, CP Canada and FS, as the case may be, shall provide any cooperation reasonably requested by Buyer to assist Buyer in obtaining such permits and licenses.

ARTICLE 3.

CONSIDERATION

follows:

3.1 PURCHASE PRICE. Buyer shall pay CP, CP Canada and FS as

- (a) Buyer shall pay CP, on behalf of itself and FS, \$1.00 for the Assets (other than the Assets referred to in Section 2.1(c));
- (b) Buyer shall pay CP, on behalf of itself and CP Canada, the Final Inventory Purchase Price (as defined in Section 3.2(a)) for the Arrow Inventory;
- (c) Buyer shall pay CP \$7 million in consideration of the execution and delivery of this Agreement and the Ancillary Agreements; and
- (d) Buyer shall pay CP, on behalf of itself and the Minority CAT Shareholders, \$1.00 for the CAT Shares, subject to adjustment as provided in Section 3.3.

All payments made to CP pursuant to this Section 3.1 (all such payments, collectively, the "PURCHASE PRICE") shall be made by wire transfer of immediately available funds to an account designated in writing by CP at least five business days prior to the Closing Date. All payments made to CP pursuant to this Section 3.1 shall be made at Closing, except as expressly provided otherwise in Section 3.2.

3.2 INVENTORY. (a) Ten days prior to the Closing Date CP shall deliver to Buyer its good faith estimates, prepared on a basis consistent with CP's and CP Canada's past practices, of: (i) CP's Cost of the Current Arrow Inventory (excluding Current Arrow Inventory located in the U.S. Arrow Factory Stores); (ii) the Fair Market Value of the Non-Current Arrow Inventory (excluding Non-Current Arrow Inventory located in the U.S. Arrow Factory Stores); (iii) CP's Cost of raw materials to be used for any current merchandise; (iv) 50% of CP's Cost of all other raw materials; (v) CP's Cost of all trim (including, without limitation, packing, labels, buttons, zippers and hang tags (collectively, "TRIM")) to be used in connection with any current merchandise; (vi) CP's Cost of raw material plus 100% of cut labor and overhead standard cost, 50% of sew labor and overhead standard cost and 100% of southbound freight to CAT (as reflected in their standard costing sheets) of work in process; and (vii) the value of the Arrow Inventory located in the U.S. Arrow Factory Stores as calculated in accordance with generally accepted accounting principles applied in a manner consistent with the preparation of the audited financial statements of Cluett American Corp. for the fiscal year ended December 31, 1999; in each case as of the close of business on the Closing Date. The aggregate estimate referred to in the preceding sentence is referred to as the "ESTIMATED INVENTORY PURCHASE PRICE"; it being understood and agreed that all Trim not referred to in clause (v) of the preceding sentence shall be valued at \$0. Buyer shall pay CP (on behalf of itself, FS and CP Canada) at Closing 95% of the

Estimated Inventory Purchase Price. The balance of the Estimated Inventory Purchase Price (the "INVENTORY ESCROW") shall be deposited in escrow by Buyer in accordance with the Escrow Agreement. For purposes hereof, "CP'S COST" shall mean CP's or CP Canada's cost as calculated in accordance with generally accepted accounting principles consistent with CP's past practice. For purposes hereof, "FAIR MARKET VALUE" shall mean the lower of CP's Cost or the average selling price of similar goods sold to Off-Price Retail Outlets (as defined in the Trademark License Agreement) in the past six months; provided that staple dress shirts and U.S. Career Apparel inventory included in the Non-Current Arrow Inventory that can reasonably be expected to be sold within the next twelve months based on prior sales history over the previous twelve (12) months of similar products adjusted for current customers' sales plans shall be deemed to have a Fair Market Value equal to 90% of CP's Cost thereof; and provided further that if the Non-Current Arrow Inventory (excluding any Non-Current Arrow Inventory located in the U.S. Arrow Factory Stores) constitutes more than 13% of the aggregate value of the Arrow Inventory (excluding any Arrow Inventory located in the U.S. Arrow Factory Stores) as reflected on CP's and CP Canada's books and records, the Fair Market Value of the portion of the Non-Current Arrow Inventory in excess of such 13% threshold (other than staple dress shirts and U.S. Career Apparel referred to in the preceding proviso) shall be reduced by 10% for purposes hereof (it being agreed that the Fair Market Value of such portion of the Non-Current Arrow Inventory prior to the application of such 10% discount shall be deemed to be the average Fair Market Value of all of the Non-Current Arrow Inventory (excluding any Non-Current Arrow Inventory located in the U.S. Arrow Factory Stores)). Set forth in Schedule 3.2(a) is the average sales price of goods sold at Discounted Prices (as defined in the Trademark License Agreement) from January 1, 2000 to April 30, 2000. The price of goods sold to Off-Price Retail Outlets will be calculated by Style Code and consistent with the average sales price of goods sold at Discounted Prices, summarized by season and product line from January 1, 2000 to April 30, 2000, as set forth in Schedule 3.2(a). Notwithstanding anything to the contrary in the foregoing, Arrow Inventory in transit will be treated pursuant to this Section 3.2(a) based on the date of receipt and the condition upon receipt. The Parties agree to cooperate in resolving any disputes with third party vendors with respect to such Arrow Inventory.

(b) As soon as practicable following the Closing Date and in any event within 60 calendar days thereof, CP shall prepare and deliver to Buyer a statement of each of the items referred to in Section 3.2(a)(i) through (vii) as of the close of business on the Closing Date (including the notes thereto, the "INVENTORY CLOSING DATE STATEMENT"), together with the work papers used by CP (and, subject to the execution and delivery by Buyer of a customary accountants' access letter, including all work papers used by CP's independent public accountants) in the preparation thereof. There shall be attached to the Inventory Closing Date Statement an annex setting forth in reasonable detail the computation of each of the items referred to in Section 3.2(a)(i) through (vii).

(c) The Inventory Closing Date Statement shall be based upon a full physical count of all Assets referred to in Section 2.1(c) as of the close of business on July 9, 2000, which

shall be conducted by CP and observed and participated in by Buyer and its representatives (the "JULY 9 COUNT"), as adjusted for items sold after the July 9 Count, items received after the July 9 Count and items manufactured after the July 9 Count. Buyer and CP shall share equally the cost of the July 9 Count. As promptly as practicable following the July 9 Count, CP shall deliver to Buyer copies of all work papers used by CP (and, subject to the execution and delivery by Buyer of a customary accountants' access letter, including all work papers used by CP's independent public accountants) in connection with the July 9 Count. CP shall, and shall cause CP's independent public accountants to, be available to Buyer and its representatives to answer questions regarding the July 9 Count or such work papers.

(d) Unless Buyer delivers written notice to CP on or prior to the 45th day following Buyer's receipt of the Inventory Closing Date Statement and the work papers used in the preparation thereof of Buyer's objection to any amount set forth in the Inventory Closing Date Statement, specifying in reasonable detail all disputed amounts and the basis therefor, Buyer shall be deemed to have accepted and agreed to the Inventory Closing Date Statement. If Buyer so notifies CP, CP and Buyer shall have 15 business days thereafter to resolve their differences, and each of Buyer and CP shall make a good faith effort to resolve their differences. If any amount remains in dispute, all such amounts remaining in dispute shall be submitted for resolution to a "Big 5" accounting firm mutually agreed upon by Buyer and CP which has not had a material relationship with either Buyer and its subsidiaries or CP and its subsidiaries within the two years preceding the appointment. If Buyer and CP cannot agree on the arbiter, they shall request that their respective independent certified public accountants jointly designate the arbiter, and such appointment shall be binding and conclusive on the parties hereto. The arbiter designated by Buyer and CP or their respective independent certified public accountants, as the case may be, is hereinafter referred to as the "ARBITER". Within 30 business days after the Arbiter's acceptance of its appointment under this Section 3.2(d) as Arbiter, each of CP and Buyer shall submit to the Arbiter (and each other) a single presentation setting forth its proposed resolution of the dispute and the basis therefor and the resulting computation of the Inventory Closing Date Statement, and each of the items referred to in Section 3.2(a)(i) through (vii) (collectively, the "FINAL INVENTORY PURCHASE PRICE"). Within 20 business days after the earlier to occur of (1) the Arbiter's receipt of the presentations referred to in the preceding sentence or (2) the expiration of the 30 business day period referred to in the preceding sentence, the Arbiter shall accept either Buyer's or CP's proposed resolution of the dispute and the resulting computation of the Inventory Closing Date Statement and each of the items referred to in Section 3.2(a)(i) through (vii), and shall notify each of Buyer and CP of its decision in writing. Such decision shall be final, binding and conclusive on all parties hereto. The party whose proposed resolution is not accepted shall pay all of the Arbiter's fees and expenses with respect to its engagement under this Section 3.2(d). Each party agrees to execute, if requested by the Arbiter, a reasonable engagement letter with the Arbiter with respect to its engagement under this Section 3.2(d).

(e) If the Final Inventory Purchase Price as determined above equals or exceeds the Estimated Inventory Purchase Price, the Inventory Escrow shall be delivered to $\sf CP$

and Buyer shall pay CP by wire transfer of immediately available funds to an account designated in writing by CP an amount equal to any excess, together with interest on such excess at an annual rate equal to the reference rate from time to time of The Chase Manhattan Bank from and including the Closing Date to but not including the date of payment. If the Final Inventory Purchase Price is less than the Estimated Inventory Purchase Price but greater than 95% thereof, the portion of the Inventory Escrow equal to any such excess shall be delivered to CP and the balance shall be delivered to Buyer. If the Final Inventory Purchase Price's is equal to 95% of the Estimated Inventory Purchase Price, the Inventory Escrow shall be delivered to Buyer. If the Final Inventory Purchase Price is less than 95% of the Estimated Inventory Purchase Price, the Inventory Escrow shall be delivered to Buyer, and CP shall pay Buyer an amount equal to such difference between 95% of the Estimated Inventory Purchase Price and the Final Inventory Purchase Price by wire transfer of immediately payable funds to an account designated in writing by Buyer, together with interest on such amount at an annual rate equal to the reference rate from time to time of The Chase Manhattan Bank from and including the Closing Date to but not including the date of payment. All Net Investment Proceeds in the Inventory Escrow Fund (as those terms are defined in the Escrow Agreement) shall be disbursed to Buyer and CP, respectively, in the same proportion that the Inventory Escrow is disbursed. Any amounts payable pursuant to this Section 3.2(e) in addition to the Inventory Escrow Fund shall be made on the third business day following (i) the 30th day after the delivery to Buyer of the Inventory Closing Date Statement and the work papers used in the preparation thereof, to the extent that there shall have been no dispute between CP and Buyer with respect to any or all of the items referred to in Section 3.2(a)(i) through (vii) and the Inventory Closing Date Statement, and/or (ii) if a dispute with respect to any or all of the items referred to in Section 3.2(a)(i) through (vii) and the Inventory Closing Date Statement is settled by the parties without resort to arbitration, the date of such settlement and/or (iii) the issuance of the Arbiter's written decision under Section 3.2(d).

3.3 CAT PURCHASE PRICE ADJUSTMENT. (a) As soon as practicable following the Closing Date and in any event within 60 calendar days thereof, CP shall prepare and deliver to Buyer a working capital statement of CAT in Honduran Lempiras as of the close of business on the Closing Date ("WORKING CAPITAL CLOSING DATE STATEMENT") and the work papers used in the preparation thereof. The Working Capital Closing Date Statement will be prepared consistent with the unaudited March 31, 2000 CAT pro forma balance sheet set forth in Schedule 3.3 in regards to current assets and current liabilities, excluding intercompany obligations.

(b) Unless Buyer delivers written notice to CP on or prior to the 45th day following Buyer's receipt of the Working Capital Closing Date Statement and the work papers used in the preparation thereof of Buyer's objection to any amount set forth in the Working Capital Closing Date Statement, specifying in reasonable detail all disputed amounts and the basis thereof, Buyer shall be deemed to have accepted and agreed to the Working Capital Closing Date Statement. If Buyer so notifies CP, CP and Buyer shall have 15 business days thereafter to resolve their differences, and each of them shall make a good faith effort to resolve their differences. If any amount remains in dispute, all such amounts remaining in dispute shall be

submitted for resolution to the Arbiter. Within 30 business days after the Arbiter's acceptance of its appointment hereunder as Arbiter, each of CP and Buyer shall submit to the Arbiter (and each other) a single presentation setting forth its proposed resolution of the dispute and the basis therefor and the resulting computation of the Working Capital Closing Date Statement. Within 20 business days after the earlier to occur of (1) the Arbiter's receipt of the presentations referred to in the preceding sentence or (2) the expiration of the 30 day period referred to in the preceding sentence, the Arbiter shall accept either Buyer's or CP's proposed resolution of the dispute and the resulting computation of the Working Capital Closing Date Statement, and shall notify each of Buyer and CP of its decision in writing. Such decision shall be final, binding and conclusive on all parties hereto. The party whose proposed resolution is not accepted shall pay all of the Arbiter's fees and expenses with respect to its engagement under this Section 3.3(b). Each party agrees to execute, if requested by the Arbiter, a reasonable engagement letter with the Arbiter with respect to its engagement under this Section 3.3(b).

(c) If current assets minus current liabilities as shown on the Working Capital Closing Date Statement is less than negative 250,000 Honduran Lempiras, CP shall pay Buyer an amount equal to any shortfall by wire transfer of immediately payable funds to an account designated in writing by Buyer, together with interest on such amount at an annual rate equal to the reference rate from time to time of the Chase Manhattan Bank from and including the Closing Date to but not including the date of payment. Any amount payable shall be made on the third business day following (i) the 30th day after the delivery to Buyer of the Working Capital Closing Date Statement and the work papers used in the preparation thereof, to the extent that there shall have been no dispute between CP and Buyer with respect thereto, or (ii) if a dispute with respect to the Working Capital Closing Date Statement is settled by the parties, the date of such settlement or (iii) the issuance of the Arbiter's written decision under 3.3(b).

3.4 ALLOCATION OF PURCHASE PRICE. (a) Buyer, CP, CP Canada and FS agree that they shall allocate the Purchase Price among the Assets as of the Closing Date on Internal Revenue Service ("IRS") Form 8594, in accordance with Section 1060 of the Internal Revenue Code of 1986, as amended (the "CODE"), and the Treasury regulations promulgated thereunder. CP shall provide Buyer with a copy of CP's , CP Canada's and FS's proposed fair market value allocation (the "ALLOCATION"), which shall be based on the Inventory Closing Date Statement as finally determined, as promptly as reasonably practicable. The Allocation shall be based upon (i) the allocation provided for in Section 3.1 and (ii) to the extent further allocation is required, the fair market value of the Assets as of the Closing Date; provided that the portion of the Allocation relating to inventory shall be based on the valuation thereof pursuant to Section 3.2. In the event that Buyer and CP are unable to agree on the Allocation with respect to Assets other than inventory within 90 days of the date on which CP provides Buyer with a copy of the Allocation, a third-party appraiser jointly selected by Buyer and CP, the cost of which shall be borne equally by Buyer, on the one hand, and CP on the other hand, shall resolve all items with respect to the Allocation to which there is a dispute between the parties.

- (b) Buyer and CP shall timely file with the appropriate Tax authorities copies of the agreed upon IRS Form 8594 and shall use the Allocation in the preparation of all Tax Returns (as defined in Section 6.11(e)), including any attachments thereto, and for all other Tax purposes. In the event any party hereto receives notice of an audit in respect of the Allocation, such party shall notify the other party in writing as to the date and subject of such audit as promptly as reasonably practicable.
- (c) If any Tax Return filed by Buyer or CP relating to the transactions contemplated hereby is challenged by the Tax authority with which such Tax Return was filed on the basis of the Allocation, as finally adjusted, the party filing such Tax Return shall keep the other party generally apprised of its decisions and the current status and progress of all administrative and judicial proceedings, if any, that are undertaken at the election of such party with respect thereto.
- (d) Any adjustment to the Purchase Price or the amount of Assumed Liabilities shall result in an appropriate adjustment to the Allocation and the IRS Form 8594 described above.

ARTICLE 4.

LIABILITIES, OBLIGATIONS AND INDEMNITIES

4.1 LIABILITIES AND INDEMNITY. (a) Effective as of the Closing, Buyer shall absolutely and irrevocably assume and be solely liable and responsible for, and from and after the Closing Date shall pay, perform and discharge when due, any and all of the Liabilities arising under, relating to or in connection with the Assets (collectively, the "ASSUMED LIABILITIES"). Set forth on Schedule 4.1(a) hereof are all of the material Assumed Liabilities.

The term "LIABILITY" shall mean and include any direct or indirect indebtedness, liability, claim, loss, damage, deficiency, cost of environmental investigation or remediation, obligation or responsibility, of whatever kind or nature, whether known or unknown, fixed or unfixed, choate or inchoate, liquidated or unliquidated, secured or unsecured, accrued or unaccrued, absolute, contingent or otherwise.

(b) Notwithstanding anything to the contrary contained in Section 4.1(a) hereof, the term "ASSUMED LIABILITIES" shall not mean or include, or be deemed to mean or include, and CP hereby agrees to indemnify and hold harmless each of Buyer and its affiliates and their respective directors, shareholders, partners, officers, employees, agents, consultants, representatives, successors, transferees and assigns (hereinafter sometimes collectively referred to as "BUYER INDEMNIFIED PARTIES") against (i) any Liabilities which are satisfied or discharged prior to the Closing or any Liabilities (other than Liabilities of CAT) that by their terms were due

and payable prior to Closing and (ii) any Excluded Liabilities (as defined below) and any liability, loss, damage (excluding consequential damages (including, without limitation, lost business profits)) and other out-of-pocket costs (including reasonable fees and expenses of outside counsel) incident to proceedings or investigations or the prosecution or defense of any claim (collectively, "DAMAGES") incurred by any of them relating to any Excluded Liability. Notwithstanding the foregoing, Buyer agrees to cooperate with CP and to take all other reasonable actions to avoid or minimize Damages relating to Section 4.1(e)(vii) hereof.

- (c) Buyer shall indemnify and hold harmless CP, CP Canada, FS, the other Minority CAT Shareholders and their respective affiliates and the directors, shareholders, officers, employees, agents, consultants, representatives, successors, transferees and assigns of any thereof (hereinafter sometimes collectively referred to as "CP INDEMNIFIED PARTIES"), against all Damages incurred by any of them relating to any Assumed Liability. It is agreed and understood that claims under this Section 4.1(c) or 4.1(d) may be brought by CP Indemnified Parties at any time and shall not be limited as to dollar amount.
- (d) Buyer shall also indemnify and hold harmless the CP Indemnified Parties against any Damages which are caused by or arise out of any breach of any covenant or agreement of Buyer hereunder or under any Ancillary Agreement.
- (e) Except for the Assumed Liabilities or as may be expressly agreed upon in any Ancillary Agreement, Buyer is not assuming, and is not responsible for, any Liability of any of CP, CP Canada and FS, whether or not related to the Assets ("EXCLUDED LIABILITIES"), including without limitation, the following:
 - (i) Except as expressly provided otherwise in Section 8.8, Liabilities of CP, CP Canada, FS or CAT under any bond, note, debenture or similar instrument or any other indebtedness for borrowed money or Liabilities of CAT to CP or any of its affiliates;
 - (ii) Any cash overdrafts;
 - (iii) Liabilities of CP, CP Canada and FS related to Excluded Assets;
 - (iv) Liabilities of CP, CP Canada and FS under this Agreement or any Ancillary Agreement;
 - (v) Liabilities of CP, CP Canada and FS to the extent arising out of or as a consequence of (A) injury or death of any person as a consequence of any event occurring prior to the Closing Date, (B) damage to the property of any third party as a consequence of any event occurring prior to the Closing Date or (C) workers' compensation claims relating to events which occurred prior to the Closing Date;

- (vi) Liabilities arising from any claim, action, suit, investigation or proceeding (whether initiated prior to or after Closing) relating to CP's, CP Canada's or FS's ownership of the Assets, CP's, CP Canada's or FS's conduct of the U.S. Business or CP's or the Minority CAT Shareholders' ownership of any CAT Shares (excluding Liabilities relating to the conduct of the Honduran Business), in each case prior to the Closing, including without limitation, claims with respect to defective products or services, alleged improper sales practices, warranty claims, or claims for any loss, damage or cost arising out of any property damage or personal injury, in each case to the extent due to the use of any product manufactured or sold by or services furnished by CP, CP Canada and FS prior to the Closing Date;
- (vii) Liabilities under Environmental Laws (as defined in Section 6.9(d) hereof) concerning any Asset, in each case to the extent arising out of or as a consequence of a violation of Environmental Law or a Release of Hazardous Material occurring or existing prior to the Closing;
- (viii) Liabilities relating to the U.S. Business resulting from or relating to the failure of CP, CP Canada or FS to qualify to do business as a foreign corporation in any jurisdiction;
- (ix) Any obligations or liabilities arising out of or relating to all claims and causes of action under United States federal, state and/or municipal civil rights and/or employment law statutes including, but without limitation, Title VII of the Civil Rights Acts of 1964, the Americans with Disabilities Act, the Age Discrimination in Employment Act, the Fair Labor Standards Act, the Occupational Health & Safety Act, or the National Labor Relations Act, in each case to the extent arising out of acts and events that occurred prior to the Closing Date;
- - (xi) Liabilities set forth on Schedule 6.10; and
 - (xii) Excluded Claims (as defined in Section 6.15).
- (f) CP shall also indemnify and hold harmless each Buyer Indemnified Party against any Damages (i) which are caused by or arise out of any breach of any covenant or agreement of any of CP, FS, CP Canada and the Minority CAT Shareholders hereunder or under any Ancillary Agreement or (ii) which are caused by or arise out of any breach of any representation or warranty of any of CP, FS, CP Canada and the Minority CAT Shareholders hereunder. No claim for breach of a representation or warranty shall be asserted under Section

- 4.1(f)(ii) after the expiration of the survival period of the relevant representation or warranty under Section 6.18. No Buyer Indemnified Party shall be entitled to indemnification for claims for breaches of representations and warranties under Section 4.1(f)(ii) unless and until the aggregate amount of such claims exceeds \$100,000 and the Buyer Indemnified Parties may claim indemnification only for the portion of such claims which in the aggregate are in excess of said \$100,000. The maximum amount recoverable by Buyer Indemnified Parties for indemnification for breaches of representations and warranties under Section 4.1(f)(ii) shall be (x) in respect of breaches relating to Assets referred to in Section 2.1(c), the purchase price therefor as finally determined under Section 3.2, and (y) in respect of breaches relating to the CAT Shares or the Honduran Business and all other breaches not referred to in clause (x) above, \$1,000,000. It is agreed and understood that claims under Section 4.1(f)(i) may be brought by Buyer Indemnified Parties at any time and shall not be limited as to dollar amount. No claim for indemnification for breaches relating to inventory may be made to the extent duplicative of the valuation of such inventory pursuant to Section 3.2. No claim for indemnification for breaches relating to CAT may be made to the extent duplicative of the working capital calculation made in accordance with Section 3.3.
- (g) Any indemnification amounts payable to Buyer Indemnified Parties or CP Indemnified Parties hereunder shall be paid to them by wire transfer of immediately available funds to a bank account designated by them in writing.
- (h) The indemnification obligations of CP and Buyer set forth in this Section 4.1 shall constitute the sole and exclusive remedy of the Buyer Indemnified Parties and the CP Indemnified Parties for the recovery of any Liabilities or Damages, except for Liabilities or Damages which are caused by or arise out of the fraud of the other party.
- (i) All Buyer Indemnified Parties hereby waive any rights and remedies that any of them may otherwise have against any CP Indemnified Party under any Environmental Law, including without limitation any claims for contribution under the Comprehensive Environmental Response Compensation and Liability act or any similar Environmental Laws.
- 4.2 INDEMNIFICATION PROCEDURE. The obligation of a party (the "INDEMNIFYING PARTY") to indemnify any person or entity (the "INDEMNIFIED PARTY") under Section 4.1 hereof is conditioned upon receiving from the Indemnified Party written notice of the assertion or institution of a claim arising from or related to any liability set forth in Section 4.1 hereof (a "CLAIM") or of the occurrence of an event which the Indemnified Party reasonably believes could lead to the assertion of a Claim, specifying in reasonable detail the nature and amount of such Claim, promptly after the Indemnified Party becomes aware of such Claim or event; provided, however, that the failure of the Indemnifying Party to receive such notice on a timely basis shall relieve the Indemnifying Party of its obligation to indemnify hereunder only if and to the extent that such failure is prejudicial to its ability to defend such Claim. Subject to the terms hereof, the Indemnifying Party shall have the absolute right, in its sole discretion and at its

sole expense, to elect to defend, settle or otherwise protect against any Claim with legal counsel of its own selection reasonably satisfactory to the Indemnified Party, provided, however, that no Claim may be settled by the Indemnifying Party without the consent of the Indemnified Party, which consent shall not be unreasonably withheld. The Indemnified Party shall have the right, but not the obligation, to participate, at its own expense, in the defense of any Claim through counsel of its own and the fees and expenses of such counsel will be at the expense of such Indemnified Party unless (i) the Indemnifying Party specifically authorized the employment of such counsel and specifically agreed to pay such counsel's fees, (ii) based on the advice of counsel, there is a conflict of interest between the position of the Indemnifying Party on the one hand and the Indemnified Party on the other hand, or (iii) the Indemnifying Party fails to assume the defense or fails to contest such action in good faith, in any which case, if the Indemnified Party notifies the Indemnifying Party that it elects to employ separate counsel, the Indemnifying Party will not have the right to assume the defense of such Claim on behalf of the Indemnified Party and the reasonable fees and expenses of such separate counsel shall be borne by the Indemnifying Party. The Indemnified Party shall, and shall cause its affiliates to, at all times cooperate in all reasonable ways with, make its relevant files and records available for inspection and copying by, and make (subject to assertion of attorney-client and other applicable privileges) its employees available or otherwise render reasonable assistance to the Indemnifying Party in connection with its defense of any Claim. Subject to the next sentence, if the Indemnified Party, without the prior consent of the Indemnifying Party (which shall not be unreasonably withheld or delayed), makes any settlement with respect to any Claim, the Indemnifying Party shall be discharged from all obligations under Section 4.1 hereof with respect to such Claim. In the event the Indemnifying Party does not undertake the defense against, settlement of or protection against any Claim in accordance with this Section 4.2, the Indemnified Party shall have the right, but not the obligation, to defend, contest, assert cross claims or counterclaims or otherwise protect against the same, to make any compromise or settlement thereof, with the consent of the Indemnifying Party which shall not be unreasonably withheld or delayed, and to recover from the Indemnifying Party and be indemnified by the Indemnifying Party for the entire cost thereof, including, without limitation, legal expenses, disbursements and all amounts paid as a result of such Claim or the compromise or settlement thereof.

4.3 OTHER INDEMNIFICATION PROVISIONS. The Parties shall make appropriate adjustments for insurance proceeds actually recovered by or on behalf of an Indemnified Party in respect of an indemnifiable liability in determining the amount of Damages pursuant to any Claims asserted under this Article 4. All indemnification payments made pursuant to this Article 4 shall be considered adjustments to the Purchase Price for federal income tax purposes.

ARTICLE 5.

EMPLOYEES AND EMPLOYEE BENEFITS

5.1 EMPLOYEE RELATIONS AND BENEFITS. (a) The parties hereto intend that there shall be continuity of employment with respect to all employees of CAT (the "CAT EMPLOYEES"). CP will submit a list of all dedicated Arrow corporate employees in the sales, marketing, product development, operations, merchandise and administrative areas in its New York office and all dedicated Arrow corporate employees in the Hong Kong and Smyrna, Georgia offices, excluding employees in the human resources and MIS departments ("CORPORATE EMPLOYEES"), and all U.S. Arrow Factory Stores employees ("U.S. AFS EMPLOYEES" and, together with the Corporate Employees, the "CP EMPLOYEES") to Buyer on the date hereof and at least fifteen (15) business days prior to the Closing Date. At any time prior to the Closing Date, Buyer may notify CP of any CP Employee that Buyer would not intend to hire on or after the Closing Date; PROVIDED, HOWEVER, that such notice shall be given no later than the Monday which is at least five business days prior to the Closing Date. Commencing on or prior to the Closing Date, on each Monday after the Closing Date and through sixty (60) days following the Closing Date (the "CONTINGENT PERIOD"), Buyer will provide CP with a list of all CP Employees it does not wish to hire and whose services Buyer will no longer need as of the end of such week. The CP Employees that Buyer does not hire are referred to herein as the "NON-RETAINED EMPLOYÉES." For all Non-Retained Employees, CP will be responsible for the payment of all severance benefits, if any, and other costs related to any termination of the Non-Retained Employees' employment with CP and FS ("TERMINATION COSTS"). At any time during the Contingent Period, but no later than the last day of the Contingent Period, Buyer shall offer employment, which shall commence and be effective (assuming the employee accepts the offer) on either the first day following the Contingent Period or, if Buyer offers employment to any CP Employee prior to the end of the Contingent Period, the date such employee accepts such offer, all CP Employees who are offered employment by Buyer or are retained on the last day of the Contingent Period ("RETAINED EMPLOYEES"). CAT Employees and Retained Employees shall collectively be referred to as "TRANSFERRED EMPLOYEES."

(b) CP will use its reasonable efforts to continue to employ employees of the Arrow Austell Cutting Facility, Arrow Austell Distribution Center and the Arrow Albertville Sewing Plant as identified on Schedule 5.1(b) and employees in the human resources and MIS departments (the "TRANSITIONAL EMPLOYEES"), and such employees will perform certain services for Buyer on terms and conditions set forth in the Transition Services Agreement. Prior to Closing, CP shall provide to the Arrow Austell Cutting Facility and the Arrow Albertville Sewing Plant employees the required notice under WARN (as defined in Section 5.1(j)) and any other applicable law and to otherwise comply with any such statute with respect to any "plant closing" or "mass layoff" (as defined in WARN) or similar event affecting such employees. CP will be responsible for Termination Costs, if any, with respect to the Transitional Employees.

(c) Each Retained Employee shall participate in employee benefit plans, agreements, programs, policies and arrangements of Buyer that are no less favorable in the aggregate than those that cover similarly situated employees of Buyer ("BUYER PLANS").

(d) CP shall or shall cause the applicable plan to retain responsibility for and continue to pay all medical, life insurance, disability and other welfare plan expenses and benefits for each Transferred, Non-Retained Employee and Transitional Employee with respect to claims incurred by such employees or their covered dependents subject to and in accordance with the terms of the applicable plan for payment of such expenses and benefits prior to and on and after the Closing Date, except as provided in the following sentence. Expenses and benefits with respect to claims incurred by Transferred Employees or their covered dependents on or after the date they become employees of Buyer shall be the responsibility of Buyer. For purposes of this paragraph, a claim is deemed incurred when the services that are the subject of the claim are performed; in the case of life insurance, when the death occurs, in the case of long-term disability benefits, when the disability occurs and, in the case of a hospital stay, when the employee first enters the hospital.

(e) With respect to any Buyer Plan that is a "welfare benefit plan" (as defined in Section 3(1) of ERISA) or any Buyer Plan that would be a "welfare benefit plan" (as defined in Section 3(1) of ERISA) if it were subject to ERISA, Buyer shall (i) cause there to be waived any pre-existing condition applicable to the Transferred Employees, to the extent waived under CP's similar welfare benefit Plan, (ii) give effect for the year in which the Closing Date occurs, in determining any deductible and maximum out-of-pocket limitations, to claims incurred and amounts paid by, and amounts reimbursed to, such Transferred Employees with respect to similar CP Plans, as in effect immediately prior to the Closing Date and (iii) recognize all credited service to the same extent such service was recognized under similar CP Plans immediately prior to the Closing Date.

(f) Upon the termination of employment with CP with respect to each Transferred Employee, CP shall cause the participation by each such employee in the Cluett Retirement Plan (the "PENSION PLAN") to cease. Upon the termination of employment with CP with respect to each Transferred Employee, CP shall cause the accrued benefit of each such Transferred Employee in the Pension Plan to fully vest. No assets or liabilities with respect to the Transferred Employees shall be transferred as a result of this Agreement from the Pension Plan to any plan or arrangement established by Buyer or any other employer for the benefit of the Transferred Employees (other than pursuant to an eligible rollover distribution). Benefits payable to the Transferred Employees under the Pension Plan through the termination of employment of each such employee shall be payable to such Transferred Employees pursuant to the terms of, and at the time and in the amounts provided under the Pension Plan based upon such Transferred Employees' years of service with, and compensation received from, CP through the date of termination of their employment with CP (including periods of employment with any other employer which is taken into account under the Pension Plan). No period of service by any Transferred Employee with Buyer or any other employer and no compensation earned by any Transferred Employee for services performed for Buyer or any other employer shall be taken into account under the Pension Plan as a result of this Agreement except as expressly provided above.

- (g) Upon the termination of employment with CP with respect to each Transferred Employee, CP shall (i) cause the account balance of each Transferred Employee in the Cluett Savings Plan ("CP'S 401(k) Plan") to fully vest, (ii) cause the participation by each Transferred Employee in the CP's 401(k) Plan to cease and (iii) make distributions to such Transferred Employees in accordance with the terms of the CP's 401(k) Plan and applicable law.
- (h) Buyer shall take all steps necessary to permit each such Transferred Employee who has received an eligible rollover distribution (as defined in Section 402(c)(4) of the Code) from the Pension Plan or CP's 401(k) Plan, if any, to roll such eligible rollover distribution into an account under the Buyer's Associate Investment Plan (the "BUYER'S 401(k) PLAN"). Periods of employment by Transferred Employees with CP and FS for which credit was given under the CP's 401(k) Plan shall be taken into account for purposes of eligibility and vesting (but not benefit accrual) under the Buyer's 401(k) Plan and any Buyer Plan that is a "pension plan" (as defined in section 3(2) of ERISA).
- (i) With respect to the Cluett Peabody Account Executive Sales Bonus Plan and the Cluett Peabody Specialty Store Sales Bonus Plan (the "SALES PLANS"), CP and Buyer agree to prorate all responsibilities, liabilities and obligations under the Sales Plans for the period commencing on January 1, 2000 and ending on December 31, 2000 based on the number of days participants in the Sales Plans are employed by CP and Buyer, respectively, during the year; provided, however, that all Transferred Employees who participate in the Sales Plans shall, only for purposes of the determination of CP's and Buyer's respective obligations under the Sales Plans, be deemed employed by Buyer commencing on the Closing Date.
- (j) Buyer agrees to provide any required notice under the Worker Adjustment and Retraining Notification Act of 1988, as amended ("WARN"), and any other applicable law and to otherwise comply with any such statute with respect to any "plant closing" or "mass layoff" (as defined in WARN) or similar event affecting any Transferred Employees and occurring on or after the date such person becomes a Transferred Employee. Buyer shall indemnify and hold harmless the CP Indemnified Parties with respect to any liability under WARN or other applicable law arising from the actions (or inactions) of Buyer or its affiliates with respect to the termination of the Transferred Employees by Buyer on after the Closing Date.
- (k) Nothing herein, expressed or implied, shall confer upon any employee or former employee of CP, Buyer or any of their affiliates (including, without limitation, the Transferred Employees), any rights or remedies (including, without limitation, any right to employment or continued employment for any specified period) of any nature or kind whatsoever, under or by reason of this Agreement.

ARTICLE 6.

REPRESENTATIONS AND WARRANTIES OF CP

CP hereby represents and warrants to Buyer that:

6.1 ORGANIZATION AND CORPORATE POWER. Each of CP and FS is a corporation duly organized, validly existing and in good standing under the laws of Delaware and CP Canada is a corporation duly organized, validly existing and in good standing under the laws of Canada. CP, CP Canada and FS have all requisite power and authority to own, lease and operate the Assets owned, leased and/or operated by them and CP and the Minority CAT Shareholders have all requisite power and authority to own the CAT Shares. CAT is a corporation duly organized, validly existing and in good standing under the laws of Honduras, with all requisite power and authority to own, lease and operate the assets and properties owned, leased and/or operated by it and to conduct the Honduran Business as it is now being conducted. CAT does not, directly or indirectly, (i) own, of record or beneficially, any outstanding voting securities or other equity interests in any other person or (ii) otherwise control any other person. CP, CP Canada, FS and the other Minority CAT Shareholders each has full corporate power and authority to enter into and perform this Agreement and each Ancillary Agreement to which it is a party.

6.2 DUE AUTHORIZATION; NO BREACH. The execution, delivery and performance by each of CP, CP Canada, FS and the other Minority CAT Shareholders of this Agreement and each Ancillary Agreement to which it is a party and the consummation of the transactions contemplated hereby and thereby have been approved by its Board of Directors and no further corporate action is required to be taken by it and its affiliates in order to execute, deliver and perform this Agreement and the Ancillary Agreements to which it is a party and to transfer the CAT Shares and Assets to Buyer. This Agreement is a valid and legally binding obligation of each of CP, CP Canada, FS and the other Minority CAT Shareholders, enforceable against each of them in accordance with its terms subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to the effect of general principles of equity (regardless of whether enforcement is considered in a proceeding at law or in equity), and each Ancillary Agreement and other agreement or instrument contemplated by this Agreement, when executed and delivered by each of CP, CP Canada, FS and the other Minority CAT Shareholders that are parties thereto in accordance with the provisions thereof, will be a valid and legally binding obligation of each such party, enforceable against each such party in accordance with its terms subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to the effect of general principles of equity (regardless of whether enforcement is considered in a proceeding at law or in equity). All persons who have executed this Agreement on behalf of CP, CP Canada, FS and the other Minority CAT Shareholders or who will execute on behalf of CP, CP Canada, FS and the other Minority CAT Shareholders, any agreement or instrument contemplated by this Agreement, have been duly authorized to do so by all necessary corporate action. Neither the execution and

delivery of this Agreement, any Ancillary Agreement and the other agreements and documents to be executed or delivered pursuant hereto, nor the consummation of the transactions contemplated hereby and thereby, will (i) violate, or conflict with, any provision of the articles of incorporation or by-laws (or other governing documents) of CP, CP Canada, FS or the other Minority CAT Shareholders, (ii) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event of default which with notice or lapse of time or both would become a default) under, or result in the termination (or grant a right of termination) of, cancellation, amendment or accelerate (or grant the right to accelerate) the performance required by, or result in the creation of any lien, security interest, charge or encumbrance upon any of the CAT Shares or Assets under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, agreement, lease, Permit or other material instrument to which CP, CP Canada, FS or the other Minority CAT Shareholders is a party or by which any of them or any of the Assets or CAT Shares are bound or (iii) violate, or conflict with, any order, writ, injunction, arbitration award, judgment or decree of any court, governmental body or arbitrator applicable to CP, CP Canada, FS or the other Minority CAT Shareholders or any applicable statute, law, rule or regulation, except, in the case of clause (ii) or (iii), as would not have a material adverse effect on the U.S. Business, the Honduran Business, the CAT Shares and the Assets taken as a whole.

6.3 REAL PROPERTY. (a) True and complete copies of the Honduran Facility Lease and the U.S. Arrow Factory Store leases (the "REAL PROPERTY") have previously been delivered to Buyer.

(b) CAT does not own any real property or interests in real property. CAT does not lease or sublease any real property other than pursuant to the Honduran Facility Lease. (i) The Honduran Facility Lease is a valid agreement, in full force and effect and constitutes a valid and binding obligation of CAT and, to CP's Knowledge (as defined below), of any other party thereto, and is legally enforceable against CAT and, to CP's Knowledge, any other party thereto, (ii) CAT has not received any written notice from any other party to the Honduran Facility Lease of the termination thereof or alleging a material default thereunder by CAT, (iii) there is no default or event which, with notice or lapse of time or both, would constitute a material default on the part of CAT (nor, to CP's Knowledge, on the part of any other party thereto) under the Honduran Facility Lease, and (iv) CAT has not transferred, assigned, hypothecated, pledged or encumbered any of its rights or interest thereunder.

(c) CP and FS do not lease or sublease any real property included in the Assets other than pursuant to the U.S. Arrow Factory Store Leases. (i) Each Arrow Factory Store Lease is a valid agreement, in full force and effect and constitutes a valid and binding obligation of CP and/or FS that is a party thereto and, to CP's Knowledge, of any other party thereto, and is legally enforceable against each of CP and FS that is a party thereto and, to CP's Knowledge, any other party thereto, (ii) CP has not received any written notice from any other party to the Arrow Factory Store Leases of the termination thereof or alleging a material default thereunder by any of CP and/or FS that is a party thereto, (iii) there is no default or event which, with notice or lapse of

time or both, would constitute a material default on the part of any of CP and/or FS that is a party thereto (nor, to CP's Knowledge, on the part of any other party thereto) under the Arrow Factory Store Leases, and (iv) neither CP nor FS has transferred, assigned, hypothecated, pledged or encumbered any of their rights or interest thereunder. For the purposes of this Agreement, "CP'S KNOWLEDGE" shall mean any fact or set of facts of which any director or officer of any of CP, CP Canada, FS and the Minority CAT Shareholders has actual knowledge.

- 6.4 PERSONAL PROPERTY. All of the fixtures, plants, buildings, structures and improvements on the Real Property and all machinery, equipment and other tangible personal property included in the Assets are being purchased by Buyer "AS IS, WHERE IS."
- 6.5 TITLE AND CONDITION OF ASSETS OTHER THAN REAL PROPERTY. CP, CP Canada and FS have good and marketable title to, or hold by valid and existing lease or license, all of the Assets (not including the Real Property, which is covered in Section 6.3) and will transfer same to Buyer at the Closing. Except for Encumbrances created by, or arising as a result of the ownership of the Assets by, Buyer, the Assets (not including the Real Property, which is covered in Section 6.3) are free and clear of all Encumbrances, except Encumbrances set forth on Schedule 6.5 hereto.
- 6.6 CONSENTS. Schedule 6.6 hereto sets forth all actions, approvals, permits, consents or authorizations required to be obtained by any of CP, CP Canada, FS and the Minority CAT Shareholders in order to consummate the transactions contemplated hereby, other than the expiration of any applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder (the "HSR ACT").
- 6.7 COMPLIANCE WITH LAWS. Except (i) as set forth in Schedule 6.7 hereto and (ii) as to Environmental Laws, which are covered by Section 6.9 hereof, CP, CP Canada, FS, CAT and the Minority CAT Shareholders are not, with respect to the conduct of the Honduran Business as presently conducted or the ownership or use of the Assets as presently used or the ownership of the CAT Shares, in default under or in violation of any United States federal, state or local or Honduran statute, law or regulation which would give rise to a material Liability or which is reasonably likely to materially and adversely affect the Assets or the CAT Shares or their ability to own the Assets or the CAT Shares or operate and use the Assets as presently used.
- 6.8 PERMITS AND LICENSES. Schedule 6.8 attached hereto sets forth all governmental licenses, permits and other governmental authorizations (collectively "PERMITS") which are issued to, held or used by CP, CP Canada, FS, CAT and the Minority CAT Shareholders in respect of the use of the Assets or the conduct of the Honduran Business as presently conducted, including the dates of issuance and expiration thereof and, except as set forth on Schedule 6.8 hereto, there are no other governmental licenses, permits, franchises or authorizations which are material to the operation of either the U.S. Business or the Honduran Business, in each case as presently conducted. Except as set forth on Schedule 6.8 hereto and for

environmental matters, which are exclusively addressed in Section 6.9, within the past 24 months, CP, CP Canada, FS, CAT and the Minority CAT Shareholders have not received any written warning, notice of violation or probable violation, notice of revocation or other written communication from or on behalf of any governmental entity, which violation has not been corrected or otherwise settled, alleging (i) any material violation of any Permit listed on Schedule 6.8, (ii) that CP, CP Canada, FS, CAT and/or the Minority CAT Shareholders need a Permit to use the Assets as presently used or to conduct the Honduran Business as presently conducted which Permit is not currently held by CP, CP Canada, FS, CAT and/or the Minority CAT Shareholders or (iii) any current violation, with respect to the use of the Assets as presently used or the conduct of the U.S. Business or the Honduran Business as presently conducted, of any United States federal, state or local or Honduran laws, statutes or regulations.

- 6.9 ENVIRONMENTAL MATTERS. Except as disclosed on Schedule 6.9, and except as would not, individually or in the aggregate, reasonably be expected to result in a material adverse effect on the Assets or the Honduran Business:
- (a) CP, CP Canada and FS are in material compliance with all U.S. Environmental Laws (as defined below) affecting the U.S. Business. For purposes of this Agreement, "U.S. ENVIRONMENTAL LAWS" shall mean any and all currently applicable United States federal, state or local laws, statutes or regulations now in effect relating to the protection of the Environment.
- (b) CP, CP Canada and FS have not received any written notice of any Environmental Claim (as hereinafter defined) and, to CP's Knowledge, no such Environmental Claim is threatened. For purposes of this Agreement, "ENVIRONMENTAL CLAIM" means any notice, claim, demand, action, suit, complaint, proceeding or other communication by any person alleging liability or potential liability (including, without limitation, liability or potential liability for investigatory costs, cleanup costs, governmental response costs, natural resource damages, property damage, personal injury, fines or penalties) relating to (x) the Release of any Hazardous Materials (as hereinafter defined) at any Arrow Factory Store, or (y) circumstances forming the basis of any violations by CP, FS, CP Canada or CAT in respect of the Assets of any U.S. Environmental Laws or in respect of the Honduran Business of any Honduran Environmental Law. For purposes of this Agreement, "RELEASE" shall mean any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping or disposing into the Environment; "HAZARDOUS MATERIALS" means any and all hazardous or toxic substances, wastes, materials or chemicals, petroleum (including crude oil or any fraction thereof) and petroleum products, friable asbestos and friable asbestos-containing materials, pollutants and polychlorinated biphenyls, in each case regulated pursuant to, as applicable, U.S. Environmental Laws or Honduran Environmental Laws (as defined in Section 6.9(d)(i) hereof); and "ENVIRONMENT" shall mean soil, surface waters, ground waters, land stream sediments, surface or subsurface strata and ambient air.

- (c) No Hazardous Materials have been Released by CP or FS or, to CP's Knowledge, any other person at any Arrow Factory Store in material violation of, or in a manner that would reasonably be expected to give rise to material liability relating to the Assets under, any U.S. Environmental Laws.
- (d) The Honduran Facility is not currently being used by CP, CAT or the Minority CAT Shareholders to, nor to CP's Knowledge, has the Honduran Facility ever been used to, make, store, handle, treat, dispose of, generate or transport Hazardous Materials in violation of any Honduran Environmental Laws (as defined below) nor, to CP's Knowledge, has any use of the Honduran Facility by any person resulted in the generation of Hazardous Materials. The Honduran Facility is currently being operated, and to CP's Knowledge has at all times, been operated in compliance with Honduran Environmental Laws in all material respects. "HONDURAN ENVIRONMENTAL LAWS" (and, together with U.S. Environmental Laws, the "ENVIRONMENTAL LAWS") shall mean any and all currently applicable Honduran federal, state or local laws, statutes or regulations now in effect relating to the environment. To CP's Knowledge, there has never been a Release, or threatened Release, of Hazardous Materials on, from or migrating onto the Honduran Facility. There are no above or below ground storage tanks presently in use, or to CP's Knowledge, formerly used for the storage of any Hazardous Materials at the Honduran Facility.
- (e) To CP's Knowledge, there have been no reports, studies, analyses, tests or monitorings pertaining to (i) hazardous activities conducted by CP, CAT or the Minority CAT Shareholders or by any other person at, in or on the Honduran Facility, or (ii) Hazardous Materials in, on, removed from or under, the Honduran Facility, or (iii) compliance with any Honduran Environmental Laws. There have been no written notices received by CP or the Minority CAT Shareholders, or since July 13, 1999 by CAT or, to CP's Knowledge, by CAT prior to July 13, 1999, from any governmental body with respect to the Release of Hazardous Materials in, on or under the Honduran Facility.
- (f) The representations and warranties contained in this Section 6.9 are the only representations and warranties being made by CP herein with respect to the Environment and no other representation or warranty contained in this Agreement or any Ancillary Agreement, express or implied, is being made with respect thereto.
- 6.10 CONTRACTS. Except as set forth in Schedule 6.10, with respect to the Contracts listed in Schedule 2.1(d), (i) each Contract is a valid agreement, in full force and effect and constitutes a valid and binding obligation of CP, CP Canada or FS, as the case may be, and, to CP's Knowledge, of any other party thereto, and is legally enforceable, (ii) none of CP, CP Canada and FS has received any written notice from any other party to a Contract of the termination thereof or alleging a material default thereunder by CP, CP Canada or FS and (iii) there is no material default or event which, with notice or lapse of time or both, would constitute a material default on the part of CP, CP Canada or FS (nor, to CP's Knowledge, on the part of any other party thereto) under any Contract.

6.11 TAX MATTERS. (a) CAT has filed or caused to be filed all Tax Returns with the D.E.I. (Direccion Ejecutivo de Ingresos) of the Government of Honduras that are or were required to be filed by it pursuant to applicable Honduran law, and CAT has paid, or made provision for the payment of, all Taxes that have become due and payable by it, regardless of whether or not shown on such Tax Returns. For tax periods beginning on or after July 13, 1999 and, to CP's Knowledge, for tax periods prior to July 13, 1999, CAT is not, and has never been, (i) required to file Tax Returns in any Taxing jurisdiction other than Honduras, and (ii) liable for Taxes in any Taxing jurisdiction other than Honduras. For tax periods beginning on or after July 13, 1999 and, to CP's Knowledge, for tax periods prior to July 13, 1999, CAT is not, has never been, owns no interest in, and has never owned an interest in, a "United States real property holding corporation" within the meaning of Section 897 of the Code, a "personal holding company" within the meaning of Section 542 of the Code, a "foreign personal holding company" within the meaning of Section 552 of the Code, a "passive foreign investment company" within the meaning of Section 1297 of the Code, a "FSC" within the meaning of Section 922 of the Code, or a "foreign investment company" within the meaning of Section 1246 of the Code. For tax periods beginning on or after July 13, 1999 and through the Closing and, to CP's Knowledge, for tax periods prior to July 13, 1999, CAT has no "earnings and profits" within the meaning of Section 986 of the Code.

- (b) Since January 1, 1998 there has been no audit commenced against CAT regarding Taxes of any type or nature.
- (c) None of the Assets is property that is required to be treated as owned by any other person pursuant to section 168(f)(8) of the Internal Revenue Code of 1954, as amended, as in effect immediately prior to the enactment of the Tax Reform Act of 1986, is "tax-exempt use property" within the meaning of section 168(h) of the Code or is "tax-exempt bond financed property" within the meaning of section 168(g) of the Code. There are no Encumbrances for Taxes on any Assets or any of the CAT Shares that arose in connection with any failure (or alleged failure) to pay any Tax or otherwise, except for Encumbrances for Taxes described on Schedule 6.5.
- (d) For purposes of this Agreement, (i) the term "TAX" or "TAXES" shall mean all Honduran or United States federal, state and local and all foreign income, profits, franchise, gross receipts, payroll, sales, employment, use, property, excise, value added, net worth, intangible, privilege, business, license, transfer, estimated, stamp, alternative or add-on minimum, environmental, withholding and any other taxes, Honduran social security, Honduran RAP FOSOVI, duties, assessments or other similar governmental charges, together with all interest, penalties and additions imposed with respect to such amounts, (ii) the term "TAX RETURNS" shall mean any return (including any consolidated combined or unitary return), declaration, estimated, installment, report, claim for refund or information return or statement relating to Taxes which is required to be filed with any governmental agency or other Tax authority, including any schedule

or attachment thereto, and including any amendment thereof and (iii) the term "TAX AUTHORITY" shall mean any authority having jurisdiction over Taxes.

6.12 CAT STOCK. All of the CAT Shares are duly authorized, validly issued and outstanding, registered, fully paid and nonassessable and are owned, of record and beneficially, by CP and the Minority CAT Shareholders free and clear, except for the obligation to make the payments referred to in Section 8.8 hereof, of all liens, claims, charges, security interests, options or other legal or equitable encumbrances. Schedule 6.12 sets forth (i) the authorized capital stock of CAT and (ii) the number of shares of each class of such capital stock that is outstanding and the name of the holder of record thereof. The CAT Shares constitute all of the outstanding shares of capital stock of CAT. There are no outstanding options, warrants, commitments or other rights of any kind relating to the sale, issuance, redemption, registration, purchase or voting of any shares of capital stock of CAT or any outstanding securities or other obligations convertible into, exchangeable for or evidencing the right to put, subscribe for, register, purchase or otherwise acquire any shares of capital stock of CAT.

6.13 LABOR MATTERS. CAT is not a party to or bound by any labor contract. Neither CP nor FS is a party to or bound by any labor contract affecting the U.S. AFS Employees or Corporate Employees. The CAT Employees, U.S. AFS Employees and/or the Corporate Employees are not represented by any union or other employee organization. There has not been, there is not presently pending or existing and, to CP's Knowledge, there is not threatened (a) any strike, slowdown, picketing, work stoppage or lockout in respect of any CAT Employees, U.S. AFS Employees and/or Corporate Employees or (b) any organizational activity against or affecting CAT. CAT is not a party to any employment agreement except as set forth on Schedule 6.13 and neither CP nor FS is a party to any employment contract being assumed by Buyer.

6.14 FINDERS; BROKERS. With the exception of fees and expenses payable to Bear, Stearns & Co. Inc., which shall be CP's sole responsibility, CP is not party to any agreement with any finder or broker, or in any way obligated to any finder or broker for any commissions, fees or expenses in connection with the origin, negotiation, execution or performance of this Agreement or the Ancillary Agreements.

6.15 LITIGATION, CLAIMS AND PROCEEDINGS. Except as set forth in Schedule 6.15 attached hereto, there are no judgments, orders, writs or injunctions of any foreign or U.S. federal, state or local court or governmental authority presently pending or, to CP's Knowledge, threatened against any of CP, CP Canada, FS and the Minority CAT Shareholders relating to the Assets or the CAT Shares or by which the Assets or CAT Shares are or would be bound, and there are no lawsuits, actions, arbitrations, claims, governmental proceedings or notices of violation presently pending or, to CP's Knowledge, threatened to which any of CP, CP Canada, FS, the Minority CAT Shareholders and CAT is a party (as plaintiff, defendant or otherwise) which relate to the Assets or the CAT Shares or the Honduran Business, except for routine litigation, claims or proceedings (including, without limitation, product liability and warranty claims or litigation, and

workers compensation claims) in which the amount in controversy does not exceed \$10,000 for any individual matter or \$100,000 in the aggregate for any related matters (the "EXCLUDED CLAIMS").

- 6.16 AFFILIATE TRANSACTIONS. Except as set forth on Schedule 6.16, CAT is not a party to any agreement or arrangement with CP or any of CP's affiliates which shall survive the Closing.
- $6.17\ 1999\ CAT\ AGREEMENT.$ CP has made all of the payments due under Section 2.2(b) of the $1999\ CAT\ Agreement$ (as defined below) as the same have become due and payable.
- 6.18 SURVIVAL OF REPRESENTATIONS AND WARRANTIES. (a) None of the representations and warranties made by CP, CP Canada, FS and/or the Minority CAT Shareholders shall survive the Closing, except as provided in this Section 6.18. All representations and warranties made by CP in Article 6 (other than those contained in Sections 6.4 and 6.11) shall survive until the 12-month anniversary of the Closing and shall thereupon expire together with any right to indemnification for breaches thereof (except to the extent a claim for indemnification for breach thereof has been made prior to the expiration of such period). The representations and warranties contained in Section 6.11 shall survive the Closing for a period of 60 days following the expiration of all applicable statutes of limitation (after giving effect to any extensions or tollings thereof) and shall thereupon expire together with any right to indemnification for breach thereof (except to the extent a claim for indemnification has been made prior to the expiration of such period). The representations and warranties contained in Section 6.4 shall not survive the Closing.
- (B) EXCEPT AS SET FORTH IN THIS AGREEMENT AND IN OTHER CERTIFICATES DELIVERED PURSUANT TO THIS AGREEMENT, NONE OF CP AND ITS AFFILIATES MAKES ANY REPRESENTATION OR WARRANTY TO BUYER WHATSOEVER, EXPRESS OR IMPLIED, WITH RESPECT TO THE U.S. BUSINESS, THE HONDURAN BUSINESS, THE ASSETS OR THE ASSUMED LIABILITIES.

ARTICLE 7.

REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer hereby represents and warrants to CP that:

7.1 CORPORATE EXISTENCE. Buyer is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation. Buyer has all requisite power and authority to own, lease and operate the Assets, and to own the CAT Shares. Buyer is duly authorized, qualified or licensed to do business as a foreign corporation and in good

standing in every jurisdiction wherein, by reason of the character of the Assets, the failure to be so

qualified would have a material adverse effect on the ability of Buyer to consummate the transactions contemplated hereby. Buyer has full corporate power and authority to enter into and perform this Agreement and each Ancillary Agreement to which it is a party.

7.2 DUE AUTHORIZATION; NO BREACH. (a) The execution and performance by Buyer of this Agreement and each Ancillary Agreement to which it is a party and the transactions contemplated hereby and thereby have been approved by its Board of Directors, and no further corporate action is required to be taken by it in order to execute, deliver and perform this Agreement. Each of this Agreement and the Ancillary Agreements to which it is a party is a valid and legally binding obligation of Buyer, and each agreement or instrument contemplated by this Agreement, when executed and delivered by Buyer in accordance with the provisions hereof, will be a valid and legally binding obligation of Buyer in each case enforceable against Buyer in accordance with its terms subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to the effect of general principles of equity (regardless of whether enforcement is considered in a proceeding at law or in equity). All persons who have executed this Agreement on behalf of Buyer or who will execute on behalf of Buyer any agreement or instrument contemplated by this Agreement, have been duly authorized to do so by all necessary corporate action. Except as set forth on Schedule 7.2 hereto, neither the execution and delivery of this Agreement, the Ancillary Agreements and all other agreements and documents to be executed or delivered hereunder, nor the consummation of the transactions contemplated hereby and thereby, will (i) violate, or conflict with, any provision of Buyer's certificate of incorporation or by-laws or other governing documents, (ii) violate, or conflict with, or result in a breach of any provisions of, or constitute a default (or an event of default which with notice or lapse of time or both would because a default) under, or result in the termination (or grant a right of termination) of, cancellation, amendment or accelerate (or grant the right to accelerate) the performance required by, or result in the creation of any lien, security interest, charge or encumbrance upon any of the properties or assets of Buyer under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, agreement, lease, Permit or other material instrument to which Buyer is a party or by which it or any of the Assets are bound or (iii) violate, or conflict with, any order, writ, injunction, arbitration award, judgment or decree of any court, governmental body or arbitrator applicable to Buyer or any applicable statute, law, rule or regulation, except, in the case of clause (ii) or (iii), as would not have a material adverse effect on the ability of Buyer to perform its obligations hereunder or any Ancillary Agreement to which it is a party.

7.3 CONSENTS. Buyer is not subject to any order, judgement or decree which would prevent the consummation of the transactions contemplated hereby. No claim, legal action, suit, arbitration, governmental investigation, action, or other legal or administrative proceeding is pending or, to the knowledge of Buyer, threatened against Buyer which would enjoin or delay the transactions contemplated hereby. Except as set forth in Schedule 7.3 hereto, no consent,

approval, order or authorization of, license or permit from, notice to or registration, declaration or filing with, any governmental authority or entity, domestic or foreign, or of any third party, is or has been required on the part of Buyer in connection with the execution and delivery of this Agreement or the Ancillary Agreements or any of the transaction documents, or the consummation of the transactions contemplated hereby and thereby, except for such consents, approvals, orders or authorizations of, licenses or permits, filings or notices the failure of which to obtain or make would not have a material adverse effect on the ability of Buyer to consummate the transactions contemplated hereby.

- 7.4 FINDERS; BROKERS. Buyer is not party to any agreement with any finder or broker, or in any way obligated to any finder or broker for any commissions, fees or expenses in connection with the origin, negotiation, execution or performance of this Agreement or the Ancillary Agreements.
- 7.5 PURCHASE FOR INVESTMENT. Buyer is aware that the CAT Shares have not been registered under the Securities Act of 1933, as amended (the "SECURITIES ACT"), or under any state or foreign securities laws. Buyer is not an underwriter, as such term is defined under the Securities Act, and is purchasing such shares solely for investment, with no present intention to distribute any such shares to any person, and Buyer will not sell or otherwise dispose of shares except in compliance with the registration requirements or exemption provisions under the Securities Act and the rules and regulations promulgated thereunder, or any other applicable securities laws.
- 7.6 AVAILABILITY OF FUNDS. Buyer has (i) in hand or (ii) the right to borrow under legally valid and binding definitive agreements all funds necessary to enable Buyer to consummate the transactions contemplated by this Agreement.
- $7.7\ \text{NO}$ OTHER REPRESENTATIONS OR WARRANTIES. Except for the representations and warranties contained in this Section 7, neither Buyer nor any other person makes any other express or implied representation or warranty on behalf of Buyer.
- 7.8 SURVIVAL OF REPRESENTATIONS AND WARRANTIES. None of the representations and warranties contained in this Article 7 shall survive the Closing, except for those representations and warranties contained in Sections 7.1, 7.2, 7.4 and 7.5, which shall survive until the 12-month anniversary of the Closing.

ARTICLE 8.

COVENANTS

- 8.1 CONDUCT OF THE BUSINESS. From the date hereof until the Closing, or termination of this Agreement in accordance with Article 10 hereof, CP and FS will:
- (a) operate the U.S. Business and the Honduran Business only in the usual and ordinary course of business consistent with past practice and do all acts and things as may be necessary to preserve, protect and maintain intact the U.S. Business and the Honduran Business as a going concern;
- (b) refrain from taking any action which reasonably could be expected to render any representation or warranty of CP contained herein untrue or incorrect in any material respect (except to the extent a representation or warranty is qualified by materiality, in which case CP will refrain from taking any action which would render such representation or warranty untrue or incorrect) as of the Closing;
- (c) comply in all material respects with all laws applicable to the U.S. Business, the Honduran Business and the Assets;
- (d) refrain from making any disposition of any Assets other than in the ordinary course of business consistent with past practice; and
- (e) make timely all payments due under Section 2.2(b) of the 1999 CAT Agreement.
- 8.2 ACCESS TO BOOKS, RECORDS AND FACILITIES. CP, CP Canada and FS agree that prior to the Closing, they will permit Buyer and its representatives full access during normal business hours and working days and upon reasonable notice to all of their properties, books, contracts, records and employees relating to the Assets, the CAT Shares, CAT, the conduct of the Honduran Business or the operations of the U.S. Business that support the Assets and will furnish Buyer and its representatives during such period, upon reasonable notice, with all such financial, operating and other information concerning the Assets, the CAT Shares, CAT, the conduct of the Honduran Business and the operations of the U.S. Business that support the Assets as Buyer or its representatives may reasonably request.
- 8.3 BEST EFFORTS AND ALTERNATIVE PROPOSALS. Each Party and each of the Minority CAT Shareholders will use all reasonable best efforts to take all action and to do all things necessary, proper or advisable in order to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements (without payment of money, commencement of litigation, the assumption of any material obligation or the entering of any agreement to divest or hold separate any assets). CP agrees that (a) neither it nor any of its

affiliates shall, and it shall direct and use its reasonable best efforts to cause its officers, directors, employees, agents and representatives (including, without limitation, any investment banker, attorney or accountant retained by it or any of its subsidiaries) not to, initiate, solicit or encourage, directly or indirectly, any inquiries or the making or implementation of any proposal or offer to consummate any transaction similar to the transactions contemplated by this Agreement and the Ancillary Agreements involving the Assets or CAT Shares that would prevent, delay or impede the consummation of the transactions contemplated hereby (any such proposal or offer being hereinafter referred to as an "ALTERNATIVE PROPOSAL") or engage in any negotiations concerning, or provide any confidential information or data to, or have any discussions or otherwise cooperate in any way with, any person relating to an Alternative Proposal, or otherwise facilitate or encourage any effort or attempt to make or implement an Alternative Proposal and (b) upon execution and delivery of this Agreement, CP will (and shall direct and use its best efforts to cause its officers, directors, employees, agents and representatives to) immediately cease and cause to be terminated any existing activities, discussions or negotiations with any parties conducted heretofore with respect to any of the foregoing.

8.4 1999 CAT AGREEMENT. If Buyer notifies CP in writing that Buyer believes in good faith that CP is entitled to claim indemnification under Section 9 of the Share Purchase Agreement, dated June 23, 1999, among CP, Marck Frederick Goldstein, Fernando Valentin Herradon, Hector Ramos Paredes, Arnoldo Ivan Aparicio and Elena Padilla Sunseri (Mrs. Edgardo Dumas) (the "1999 CAT AGREEMENT") and that CP is entitled to exercise its right of set-off under Section 9.5 of the 1999 CAT Agreement, CP shall promptly make and use its reasonable best efforts to pursue in good faith such claim and to exercise such right of set-off. If CP receives after the date hereof any cash indemnification payment under Section 9 of the 1999 CAT Agreement, CP shall pay an amount in cash to Buyer equal to such payment less any reasonable out-of-pocket expenses or costs incurred by CP relating to or in respect of CP's claim for the recovery of such payment. CP represents and warrants that it has not made any claim for indemnification under the 1999 CAT Agreement prior to the date hereof and, to CP's Knowledge, has no basis for any such claim. CP will not make any claim for indemnification under the 1999 CAT Agreement without the written consent of Buyer, which consent shall not be unreasonably withheld.

8.5 NOTIFICATION. CP shall provide Buyer with written notice prior to Closing of any material damage or destruction of the Assets, material adverse change in the U.S. Business, the Assets or the Honduran Business the commencement of any lawsuit, action, arbitration or governmental proceeding to which any of CP and its affiliates is a party which seeks to prevent the consummation, or otherwise challenges the validity, of the transactions contemplated hereby, or breaches of Article 6 that constitute a failure to satisfy Section 9.3(b), in each case of which CP acquires actual knowledge prior to the Closing Date.

8.6 LETTER OF CREDIT. Buyer shall use all reasonable efforts to replace at Closing all outstanding letters of credit pertaining to the orders which CP, CP Canada and FS

have with third party manufacturers for the manufacture of goods for use in the U.S. Business ("SOURCING COMMITMENTS") to which any of CP, CP Canada and FS is a party. If Buyer is unable to replace at Closing any such outstanding letters of credit, in whole or in part, Buyer will use all reasonable efforts to provide at Closing "back-to-back" letters of credit to each of the banks that issued each of the letters of credit that are not being replaced (each, an "ISSUING BANK") in a form satisfactory to the relevant Issuing Bank (in its sole discretion) issued by a bank satisfactory to the relevant Issuing Bank (in its sole discretion), in an amount equal to 100% of the then undrawn stated amount of the outstanding letters of credit issued by such Issuing Bank that are not being replaced. The notice of reimbursement with respect to any drawdown on any such "back-to-back" letter of credit shall state the amount of the drawdown and the purchase orders to which it relates. If Buyer is unable to provide such "back-to-back" letters of credit, Buyer will reimburse CP, FS or CP Canada, as the case may be, for all amounts drawn down under outstanding letters of credit that are not being replaced and for which "back-to-back" letters of credit are not being provided within five business days after receipt by Buyer of reasonable evidence of such drawdown. Such evidence shall include the letter of credit number, the latest ship date, product description, product cost, price per unit and a copy of the funding message and such other documentation that Buyer may reasonably request. On the date hereof and at least ten business days prior to Closing CP will provide Buyer with a list of each outstanding letter of credit, stating the letter of credit number, the issuing bank, the name, address and country of the beneficiary, the purchase orders covered, the expiration date and the amount outstanding on the day the list is compiled with respect to each outstanding letter of credit. If Buyer offers to provide a "back-to-back" letter of credit to an Issuing Bank issued by Chase Manhattan Bank, N.A., Citibank, N.A. or The Bank of New York in customary form in an amount equal to 100% of the then undrawn stated amount of the relevant outstanding letter of credit, and the Issuing Bank states that any such issuing bank or such form is not satisfactory to the Issuing Bank, Buyer shall be deemed to have satisfied its obligations under the second sentence of this Section 8.6 with respect to such outstanding letter of credit.

8.7 ACCOUNTS RECEIVABLE. (a) After the Closing Date, CP and FS shall have the sole right and authority to collect, for their own account and sole benefit, all monies receivable in respect of the Assets for any period prior to the Closing Date, and Buyer shall have the sole right and authority to collect, for its own account and sole benefit, all monies payable in respect of the Assets for any period on or following the Closing Date. Buyer and CP shall jointly send a letter on the Closing Date to each customer of the U.S. Business announcing the transaction contemplated hereby and describing the customer's obligations to each of CP and Buyer, including specific instructions for the remittance of payments to each of CP and Buyer. The form and substance of such letter shall be agreed upon at or prior to the Closing.

(b) If CP, FS or any of their affiliates (or their successors) shall receive any such monies with respect to any period on or following the Closing Date, it shall hold all such monies in trust for the sole benefit of Buyer and its affiliates. If Buyer or any of its affiliates (or their successors) shall receive any such monies with respect to any period prior to the Closing Date, it shall hold all such monies in trust for the sole benefit of CP and its affiliates. On the first and fifteenth day of each calendar month (or, if such day is not a business day, the first business

day thereafter), CP and FS (or their successors) or Buyer and its affiliates (or their successors), as the case may be, shall cause the transfer and delivery to the other of any such monies which they (or their successors) may receive after the Closing Date. No Party will take any action that compromises the ability of the other Party and its affiliates (or their successors) to collect, for its own account and sole benefit, all monies payable to it as provided herein. Each of CP, FS and Buyer will use all reasonable efforts to collect its own receivables and to assist the other in the collection of their respective accounts receivable; provided, however, that nothing in this Agreement, shall be deemed to require any of them (i) to take any action that such Party believes, in its sole discretion, would be detrimental to such Party's business relationship with any customer or (ii) to incur any out-of-pocket expenses or allocate any significant employee resources to the collection of the other's accounts receivable.

(c) All disputed items and deductions taken from payments for invoices, arising from, but not limited to, markdown and other allowances, returns, NSF checks, vendor guide violations, shortages, discounts and price relating to pre-Closing accounts receivable and shipments are the responsibility of CP or FS, as the case may be, and CP and FS shall have sole authority to make decisions regarding the same. All disputed items and deductions taken from payments for invoices, arising from, but not limited to, markdown and other allowances, returns, NSF checks, vendor guide violations, shortages, discounts and price relating to post-Closing accounts receivable and shipments are the responsibility of Buyer, and Buyer shall have sole authority to make decisions regarding the same. Each of CP, FS and Buyer will identify to the other all such disputes and deductions that come to its attention but are the responsibility of the other and provide any documentation in its possession to the responsible entity. In the event that a deduction has been taken against the wrong Party, such Party shall provide all documentation provided by the customer to the responsible party. Settlement in the form of a cash reimbursement of deductions taken against the wrong Party shall be made on the first and fifteenth day of each calendar month (or, if such day is not a business day, the first business day thereafter).

8.8 1999 CAT AGREEMENT. Buyer agrees that it shall make timely on CP's behalf all payments required to be made by CP after the date hereof under Section 2.2(b) of the 1999 CAT Agreement. If such payment obligation is reduced as a result of CP exercising its right of set-off under the 1999 CAT Agreement as a result of a claim for indemnification thereunder made in accordance with Section 8.4 hereof, Buyer shall reimburse CP for any reasonable out-of-pocket expenses or costs incurred by CP relating to or in respect of CP's claim pursuant to Section 8.4 that resulted in such reduction.

8.9 RETURNED INVENTORY. Buyer and CP agree (a) to minimize all returns of Arrow Inventory sold by CP or FS to third parties prior to Closing and any such return needs to be authorized by CP, (b) if such returns cannot be avoided as a practical manner consistent with the past practice of the U.S. Business, Buyer agrees to purchase all such inventory which is returned to CP or FS within 180 days after Closing (the "RETURNED INVENTORY"). Buyer shall pay CP for the Returned Inventory which is Current Arrow Inventory an amount equal to fifty percent (50%) of Buyer's average selling price therefor.

8.10 U.S. AFS CASH REGISTERS. Buyer agrees that upon the expiration or termination of any U.S. Arrow Factory Store lease (as it may be extended or renewed from time to time) it shall deliver to CP, at CP's expense and for no other consideration, each cash register located in such U.S. Arrow Factory Store on the Closing Date.

8.11 BREAK-UP FEE. If Buyer does not obtain the rights of CDG under the KCPL Shirt License Agreement (including without limitation due to the issuance of a restraining order or termination of such agreement due to action by KCPL) or enter into a new shirt license agreement with KCPL in conjunction with the termination of the KCPL Shirt License Agreement in connection with the consummation of the transactions contemplated by the CDG Share Purchase Agreement or the CDG Asset Purchase Agreement (as defined in Section 8.12), CP shall wire transfer to a bank account designated in writing by Buyer \$7 million within five business days after such transactions have been abandoned by CP and Buyer in accordance with the terms of this Agreement, the CDG Share Purchase Agreement and/or the CDG Asset Purchase Agreement or Buyer is prevented from exercising CDG's rights under the KCPL Shirt License Agreement.

8.12 CLUETT DESIGNER GROUP. Each of CP, CDG and Buyer agrees to negotiate in good faith and to use its reasonable best efforts to execute and deliver, and/or to cause its relevant affiliates to execute and deliver, as soon as practicable after the date hereof (and in no event later than June 30, 2000) (a) a Purchase and Sale Agreement between CDG and Buyer (the "CDG ASSET PURCHASE AGREEMENT") providing for (i) either (1) the termination of the License Agreement dated as of August 1, 1997 between K.C.P.L., Inc. ("KCPL") and CDG, as amended (the "KCPL SHIRT LICENSE AGREEMENT") and the execution and delivery by Buyer and KCPL of a new shirt license agreement or (2) the assignment of CDG's rights and obligations under the KCPL Shirt License Agreement to Buyer, (ii) the sale to Buyer by CDG of all inventory of CDG related to the business conducted by it pursuant to the KCPL Shirt License Agreement and all rights under certain scheduled agreements, (iii) the assumption by Buyer of certain scheduled liabilities, (iv) the payment by Buyer to CDG of an amount in cash equal to CDG's cost (calculated in the manner agreed upon by the parties in the CDG Asset Purchase Agreement) of the inventory being sold, subject to adjustment based on a full physical count thereof, and (v) the escrow of 5% of the estimated purchase price referred to in clause (a)(iv) above, and (b) a Share Purchase Agreement among Bidermann Tailored Clothing, Inc. ("BTC"), CP and Buyer (the "CDG SHARE PURCHASE AGREEMENT") providing for (i) the sale of all outstanding shares of capital stock of CDG to Buyer, (ii) the payment by Buyer to CP and BTC of the same amount in cash referred to in clause (a)(iv) above, subject to the same adjustment and escrow arrangements referred to in clauses (a)(iv) and (a)(v) above, (iii) the transfer to affiliates of CDG of all assets of CDG not to be sold to Buyer pursuant to the CDG Asset Purchase Agreement and (iv) the indemnification of Buyer and its affiliates against all pre-closing liabilities of CDG not to be assumed by Buyer pursuant to the CDG Asset Purchase Agreement (such indemnification shall not be subject to any deductible or maximum amount). Each of the CDG Asset Purchase Agreement and the CDG Share Purchase Agreement shall (i) have attached as an exhibit thereto

an agreement (the "HOSIERY AGREEMENT") with terms consistent in all material respects with the term sheet attached hereto as Exhibit 8.12 and shall provide that the execution and delivery of the Hosiery Agreement by Buyer shall be a condition precedent to the obligations of CP, BTC and CDG to close under the CDG Asset Purchase Agreement and the CDG Share Purchase Agreement, (ii) provide that none of the representations and warranties of Buyer, CP, BTC and CDG contained therein shall survive the closing thereof and (iii) have a provision identical to Section 10.1(b) hereof. Neither the CDG Asset Purchase Agreement nor the CDG Share Purchase Agreement shall have a provision similar to Section 9.3(c) hereof. If all consents and agreements required to be obtained from KCPL and its affiliates under the KCPL Shirt License Agreement, or otherwise, in connection with the execution and delivery of the CDG Asset Purchase Agreement and the consummation of the transactions contemplated thereby have been obtained irrevocably in writing on or before July 15, 2000, the transactions contemplated by the CDG Asset Purchase Agreement will be consummated on the terms and subject to the conditions set forth therein as soon as practicable and the CDG Share Purchase Agreement will thereafter have no force or effect. If all such consents and agreements are not obtained from KCPL on or before July 15, 2000, the transactions contemplated by the CDG Share Purchase Agreement will be consummated on the terms and subject to the conditions set forth therein as soon as practicable and the CDG Asset Purchase Agreement will thereafter have no force or effect.

8.13 TRADEMARK LICENSE AGREEMENT. (a) If, prior to the Closing Date, either of the retailers Mervyn's or Sears (i) changes its format of distribution to that of a lower-level retailer which is unacceptable to CP; (ii) changes its format of distribution in any other manner that is unacceptable to CP; (iii) ceases carrying or selling Licensed Products that display the Licensed Trademarks or, in the reasonable opinion of CP, changes its format of distribution resulting in a material decrease in the amount of selling space dedicated to men's sportswear and dress shirts; (iv) is liquidated; or (v) reduces the number of Mervyn's or Sears stores, respectively, in operation by more than twenty-five (25) percent of the number of such stores in operation as of the date hereof (any one of which shall constitute either a "PRE-CLOSING MERVYN'S EVENT" or a "PRE-CLOSING SEARS EVENT"); then the Parties shall revise the form of Trademark License Agreement attached hereto as Exhibit 2.4(b) prior to the Closing Date as follows: (x) except upon the occurrence of an event described in clause (v) hereof (such an event, an "EXCEPTED EVENT"), the definition of the "Base Amount" as set forth in Section 4(a)(i) of the form of Trademark License Agreement attached hereto as Exhibit 2.4(b) shall be revised to equal \$5 million MINUS the greater of (A) any royalties that would be payable as determined pursuant to the method of formulation set forth in Section 4(a)(ii) thereof based on net sales by Licensor (as defined in the form of Trademark License Agreement attached hereto as Exhibit 2.4(b)) to Mervyn's (in the event of a Pre-Closing Mervyn's Event) or Sears (in the event of a Pre-Closing Sears Event) during the four calendar quarters immediately preceding the date upon which such Pre-Closing Mervyn's Event or Pre-Closing Sears Event occurs or (B) any royalties that would be payable as determined pursuant to the method of formulation set forth in Section 4(a)(ii) thereof based on net sales of Licensor to Mervyn's (in the event of a Pre-Closing Mervyn's Event) or Sears (in the event of a Pre-Closing Sears Event) during the 1999 calendar year (the greater of which shall be

referred to as the "AMOUNT OF ROYALTIES"); in the event that an Excepted Event occurs, the definition of the "Base Amount" as set forth in Section 4(a)(i) of the form of Trademark License Agreement attached hereto as Exhibit 2.4(b) shall be revised to equal \$5 million MINUS a percentage of the Amount of Royalties, which percentage shall be the percentage of Mervyn's or Sears stores closed that is referred to in clause (v) hereof; (y) in the event that a Pre-Closing Mervyn's Event other than one based on clause (v) hereof occurs, the text of Section 4(d) shall be stricken and replaced with the following text: "[This section purposely left blank]"; and (z) in the event that a Pre-Closing Mervyn's Event other than one based on clause (v) hereof occurs, the term "Readjusted Minimum Royalty" shall be stricken in its entirety.

- (b) In the event that a Pre-Closing Mervyn's Event or a Pre-Closing Sears Event is based on the occurrence of an event described in clause (v) hereof, on each anniversary of the Readjustment Date for such Pre-Closing Mervyn's Event or Pre-Closing Sears Event, Licensor shall further adjust the Readjusted Minimum Royalty to take into account any additional decrease during the previous year in the number of Mervyn's stores in operation.
- (c) Notwithstanding the foregoing, for purposes of this Agreement, the provisions of Section 8.13(a) and (b) shall not apply to a Pre-Closing Mervyn's Event or Pre-Closing Sears Event that occurs as a result of the purchase by either Mervyn's or Sears of Licensed Products from Buyer or any one of Buyer's affiliates instead of from CP.

ARTICLE 9.

CONDITIONS OF CLOSING

- 9.1 MUTUAL CONDITIONS. The obligations of each Party and each of the Minority CAT Shareholders to consummate the transactions contemplated by this Agreement and the Ancillary Agreements are, unless waived by each Party and the Minority CAT Shareholders, subject to the fulfillment on or before the Closing, of each of the following conditions:
 - (a) No statute, rule, regulation, injunction or restraining order shall be in effect to forbid or enjoin the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements:
 - (b) The transactions contemplated by this Agreement and the Ancillary Agreements to be completed before the Closing shall have been consummated upon the terms and subject to the conditions set forth therein; and
 - (c) All waiting periods under the HSR $\operatorname{\mathsf{Act}}$ shall have expired or been terminated.

- 9.2 ADDITIONAL CONDITIONS OF CP. The obligations of CP to consummate the transactions contemplated by this Agreement and the Ancillary Agreements are, unless waived by CP, subject to the fulfillment, on or before the Closing of each of the following additional conditions:
 - (a) Buyer and each other party thereto other than CP and its affiliates shall have executed and delivered each of the Ancillary Agreements to which it is a party; and
 - (b) All covenants of Buyer under this Agreement and the Ancillary Agreements to be performed prior to the Closing shall have been performed in all material respects, and the representations and warranties of Buyer contained in this Agreement and the Ancillary Agreements shall be true and correct on and as of the Closing in all material respects (except to the extent qualified by materiality, in which event such representations and warranties shall be true and correct) with the same effect as though such representations and warranties had been made on and as of such date.
- 9.3 ADDITIONAL CONDITIONS OF BUYER. The obligations of Buyer to consummate the transactions contemplated by this Agreement and the Ancillary Agreements are, unless waived by Buyer, subject to the fulfillment, on or before the Closing, of each of the following additional conditions:
 - (a) CP and each other party thereto other than Buyer shall have executed and delivered each of the Ancillary Agreements to which it is a party;
 - (b) All covenants of CP, CP Canada, FS and the Minority CAT Shareholders under this Agreement and the Ancillary Agreements to be performed prior to the Closing shall have been performed, and the representations and warranties of CP contained in this Agreement and the Ancillary Agreements shall be true and correct on and as of the Closing in all material respects (except to the extent qualified by materiality in which event such representations and warranties shall be true and correct) with the same effect as though such representations and warranties had been made on and as of such date; and
 - (c) There shall not have occurred since the date hereof any damage to, or destruction or loss of, Arrow Inventory that constitutes a material adverse change to the Arrow Inventory.

ARTICLE 10.

TERMINATION; SURVIVAL

10.1 TERMINATION BY BUYER OR CP. Anything herein or elsewhere to the contrary notwithstanding, this Agreement may be terminated and the transactions contemplated hereby abandoned at any time prior to or at the Closing by:

- (a) mutual written consent of CP and Buyer;
- (b) CP or Buyer, by written notice to the other, if the transactions contemplated hereby are not consummated on or before July 31, 2000 and if the failure to consummate such transactions on or before such date did not result from a breach prior to or on the Closing Date of any representation, warranty or covenant of the Party seeking such termination; provided that if the condition set forth in Section 9.1(c) has not been satisfied prior to July 31, 2000, CP may in its reasonable discretion elect to extend the July 31, 2000 deadline referred to above to October 31, 2000; and provided further that Buyer may not terminate this Agreement and abandon the transactions contemplated hereby because the condition set forth in Section 9.3(c) has not been satisfied (i) until the twelve month anniversary of the material adverse change referred to in Section 9.3(c) and (ii) unless CP and its affiliates have not cured such material adverse change on or prior to such twelve month anniversary; and provided further that if CP extends the July 31, 2000 deadline referred to above, it must similarly extend the corresponding deadlines in the CDG Asset Purchase Agreement and the CDG Share Purchase Agreement.
- (c) CP or Buyer, by written notice to the other, if a court of competent jurisdiction or governmental, regulatory or administrative agency or commission shall have issued an order, decree or ruling or taken any other action, in each case, permanently restraining, enjoining or otherwise prohibiting the consummation of the transactions contemplated hereby and such order, decree, ruling or other action shall have become final and nonappealable; or
- (d) either Party, by written notice to the other Party, if the other Party is in breach in any material respect of any of its representations or warranties made in this Agreement or the Ancillary Agreements, or is in violation or default in any material respect of any of its covenants or agreements in this Agreement or the Ancillary Agreements, if such breach, violation or default is not cured within twenty (20) business days after written notice by the notifying Party to the other Party;
- 10.2 SURVIVAL. If this Agreement is terminated pursuant to Section 10.1 hereof, this Agreement shall become void and of no further force and effect, except for the provisions of Article 4 and Sections 12.1(a), 14.1, 14.2 and 16.8 hereof; PROVIDED that such termination shall not relieve any party for liability for Damages resulting from its breach of this Agreement. Promptly

following termination of this Agreement each party will destroy or return to the other parties all documents received from such parties in connection with the contemplated transaction, except documents which have been publicly distributed or publicly filed.

ARTICLE 11.

CLOSING

11.1 CLOSING. The closing of the transactions contemplated by this Agreement (the "Closing") shall take place at the offices of Simpson Thacher & Bartlett, 425 Lexington Avenue, New York, New York as soon as practicable after all the conditions to Closing set forth in Article 9 hereof shall be satisfied or duly waived, or at such other time and place as Buyer and CP may mutually agree (the date of the Closing referred to herein as the "CLOSING DATE").

 $\,$ 11.2 CP, CP CANADA AND FS OBLIGATIONS AND CLOSING DELIVERIES. At the Closing, CP, CP Canada and FS shall deliver, or cause to be delivered, to Buyer:

- (a) Bill of Sale and Assignment substantially in the form attached as Exhibit 2.4(c) hereto executed by CP, CP Canada and FS;
- (b) Such other instruments of sale, transfer, conveyance and assignment, in form and substance reasonably satisfactory to Buyer's counsel, as shall be effective to vest in Buyer good title, rights and interest to the Assets (the instruments referred to in Sections 10.2(a) and 10.2(b) being collectively referred to herein as the "ASSIGNMENT AND ASSUMPTION INSTRUMENTS") and such other documents to facilitate the transfer of the Real Property as Buyer or Buyer's counsel may reasonably request;
- (c) Each of the other Ancillary Agreements (other than the Assumption Agreement) executed by each of CP, FS and CP Canada, as applicable, and, in the case of the Trademark License Agreement, Resources and, in the case of the Waivers and Consents, CP and its contractors, subcontractors and warehousemen that agree to sign after CP uses its reasonable efforts to cause them to sign;
- (d) Written receipt executed by CP of payment of the portion of the Purchase Price paid at Closing;
- (e) Certificate of the President or any Vice President of CP that the conditions set forth in Section 9.3(b) have been satisfied; and

- (f) the certificates representing CP's CAT Shares, duly endorsed or accompanied by stock powers duly executed and with all necessary tax stamps affixed thereto, as described in Article 2 hereof.
- 11.3 BUYER'S OBLIGATIONS AND CLOSING DELIVERIES. At the Closing, Buyer shall deliver, or cause to be delivered:
 - (a) to CP, the Purchase Price (except as expressly provided otherwise in Section 3.2);
 - (b) to CP, each of the Ancillary Agreements (other than the Bill of Sale and Assignment) executed by Buyer;
 - (c) to the Escrow Agent, the Inventory Escrow;
 - (d) to CP, a Certificate of the President or any Vice President of Buyer to the effect that the conditions set forth in Section 9.2(b) have been satisfied.
- 11.4 MINORITY CAT SHAREHOLDER'S OBLIGATIONS AND CLOSING DELIVERIES. At the Closing, the Minority CAT Shareholders shall deliver, or cause to be delivered, to Buyer:
 - (a) the certificates representing the Minority CAT Shareholders' CAT Shares, duly endorsed or accompanied by stock powers duly executed and with all necessary tax stamps affixed thereto, as described in Article 2 hereof.

ARTICLE 12.

EXPENSES AND POST CLOSING OBLIGATIONS

12.1 TAXES AND OTHER CHARGES. (a) Except as specifically provided for otherwise in Section 3.2(c), and regardless of whether or not the transactions contemplated hereby are consummated each party to this Agreement shall pay all expenses incurred by it or on its behalf in connection with the preparation, authorization, execution and performance of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby, including, but not limited to, all fees and expenses of all consultants, brokers, investment bankers, agents, representatives, counsel and accountants engaged by such party. Any liability for all sales, documentary stamp, transfer and similar Taxes (including, without limitation, all recording fees and related charges and real property gains tax) arising from or attributable or related to the sale, transfer or assignment to Buyer of any of the Assets or CAT Shares shall be paid by the party that is primarily responsible for such Taxes under applicable law and such party shall, at its expense, file all Tax Returns and other documentation with respect to

all such Taxes. Buyer shall provide to CP any resale certificate required with respect to the transfer of the Assets to Buyer.

- (b) CP and Buyer shall cooperate regarding the filing of any Tax Returns with respect to Taxes addressed in Section 12.1(a) relating to the Assets and the CAT Shares that cover a period which includes the date of the Closing. Property Tax Returns will be filed by the party which owns the property subject to the return on the assessment date for the property tax.
- (c) CP, at its option, shall have sole control over any contest or other proceeding (including claims for refund or challenges of tax assessments) which relates to liability for taxes for periods or portions thereof ending on or prior to the Closing Date, subject to approval of Buyer (not to be unreasonably withheld) relating to any Liability reasonably likely to be imposed on or otherwise likely to be the sole responsibility of Buyer, provided, that the approval of Buyer shall not be required if CP is obliged, and acknowledges its obligation, to indemnify Buyer therefor pursuant hereto. In any event, any refunds of taxes which become available on or after the date of the Closing but which relate to tax periods or portions thereof ending on or prior to the Closing shall belong to CP.
- 12.2 FURTHER ASSURANCES. At any time after the Closing, the Parties and the Minority CAT Shareholders agree to cooperate with one another to execute and deliver such other documents, instruments of transfer or assignment, files, books and records and do all such further acts and things as may be reasonably required to carry out the transactions contemplated by this Agreement or the Ancillary Agreement.
- 12.3 ACCESS TO BOOKS, RECORDS AND FACILITIES. CP, CP Canada and FS agree that on and after the Closing it will permit Buyer and its representatives, during normal business hours and working days and upon reasonable advance notice, (i) to have access to and to examine and make copies of all books and records of CP, CP Canada and FS (except books and records protected by attorney-client or other privilege which any of CP, CP Canada and FS may be entitled to assert against Buyer in any pending or threatened proceeding, suit or action) which relate to the Assets, the CAT Shares, CAT, the conduct of the Honduran Business or the operations of the U.S. Business that support the Assets to the extent that the events reflected therein relate to transactions or events occurring prior to the Closing or to transactions or events occurring subsequent to the Closing which arise out of transactions or events occurring prior to the Closing and (ii) to have access to and to examine and make copies of all documents listed in the Schedules attached hereto and all files, records and papers of any and all proceedings and matters listed in the Schedules attached hereto. All books and records of CP, CP Canada and FS relating to the Assets, the CAT Shares, CAT, the conduct of the Honduran Business or the operations of the U.S. Business that support the Assets will be preserved by CP, CP Canada and FS in accordance with their records retention policy, but in no event for a period of less than three years following the Closing. Prior to any destruction or disposition by any of CP, CP Canada and FS of any such books and records, one of them will notify Buyer in writing and Buyer shall have the right to receive and retain such books and records at its expense. Buyer agrees that, after the

Closing, it will permit, and will cause its subsidiaries to permit, CP, CP Canada and FS and their representatives full access during normal business hours and working days and upon reasonable advance notice to have access to the books and records of Buyer and its subsidiaries (except records protected by attorney-client or other privilege which Buyer or its affiliates may be entitled to assert against any of CP, CP Canada and FS in any pending or threatened proceeding, action or suit), relating to the Assets, the CAT Shares, CAT, the conduct of the Honduran Business or the operations of the U.S. Business that support the Assets, to the extent that any of the foregoing relates to periods prior to the Closing, and is reasonably necessary in connection with any then pending or threatened litigation, claim, liability, or judicial or administrative matters in which any of CP, CP Canada and FS is involved.

ARTICLE 13.

BULK SALES LAW

13.1 WAIVER. Buyer hereby waives compliance by CP, CP Canada and FS with any bulk sales laws (including any applicable bulk sales provision of any state sales tax law) which may be applicable. CP agrees to indemnify and hold Buyer harmless against any and all claims, losses, damages, liabilities, costs and expenses (including, without limitation, any Taxes) incurred by Buyer or any of its affiliates as a result of any failure to comply with any such bulk sales laws.

ARTICLE 14.

PUBLICITY, CONFIDENTIALITY

14.1 PUBLICITY. The Parties and the Minority CAT Shareholders agree that no publicity, release or announcement concerning (a) the execution of this Agreement or the Ancillary Agreements, (b) any of the provisions of this Agreement or the Ancillary Agreements or the transactions contemplated hereby or thereby or (c) the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements shall be issued without the advance written approval of the form and content of the same by the Parties and the Minority CAT Shareholders; provided, however, that no such consent shall be required when such disclosure is required by applicable law, although the Parties and the Minority CAT Shareholders agree to consult with each other as to the content of any release so required and consider in good faith the comments of the other thereon.

14.2 CONFIDENTIALITY. Buyer and CP confirm that they and their affiliates are bound by the terms of the confidentiality agreement, dated December 9, 1998, as amended (the "CONFIDENTIALITY AGREEMENT"), between Cluett American Investment Corp. and Phillips-Van Heusen Corporation and that they will keep and treat the evaluation material and all other items of

confidential information exchanged hereunder by Buyer and its affiliates, on the one hand, and CP and its affiliates, on the other hand, in accordance with the terms of that confidentiality agreement.

ARTICLE 15.

NOTICES

15.1 NOTICES. Any notices or communications permitted or required hereunder shall be deemed sufficiently given if hand-delivered, or sent by (i) registered or certified mail return receipt requested, (ii) telecopy or other electronic transmission service (to the extent receipt is confirmed) or (iii) by overnight courier, in each case to the Parties at their respective addresses and telecopy numbers set forth below, or to such other address of which any Party may notify the other Party in writing.

If to Buyer, to

Phillips-Van Heusen Corporation 200 Madison Avenue New York, New York 10016 Attention: Chief Executive Officer Phone: (212) 381-3501

Fax: (212) 381-3979

With a copy to:

Phillips-Van Heusen Corporation 200 Madison Avenue New York, New York 10016 Attention: General Counsel Phone: (212) 381-3509 Fax: (212) 381-3970

If to CP, FS, CP Canada or any Minority CAT Shareholder, to

Cluett, Peabody & Co., Inc. 48 West 38th Street New York, New York 10018 Attention: Chief Executive Officer

Phone: (212) 984-8915 Fax: (212) 984-8925

with copies to:

Vestar Capital Partners 245 Park Avenue 41st Floor New York, New York 10167 Attention: General Counsel Phone: (212) 351-1630 Fax: (212) 808-4922

and

Simpson Thacher & Bartlett 425 Lexington Avenue New York, New York 10017 Attention: Peter J. Gordon Phone: (212) 455-2605 Fax: (212) 455-2502

ARTICLE 16.

MISCELLANEOUS

16.1 BINDING EFFECT; ASSIGNMENT. This Agreement shall be binding upon, and inure to the benefit of, all the Parties and Minority CAT Shareholders and their respective successors, legal representatives and assigns permitted in accordance with this Section 16.1. Nothing herein shall create or be deemed to create any third party beneficiary rights in any person or entity not a party hereto. No assignment of this Agreement or of any rights or obligations hereunder may be made by any Party or Minority CAT Shareholder (by operation of law or otherwise) without the prior written consent of the other Parties and Minority CAT Shareholders, and any attempted assignment without the required consents shall be void; provided, however, that no such consent shall be required for Buyer to assign part or all of its rights and obligations under this Agreement prior to the Closing to one or more subsidiaries of Buyer, or after the Closing to any third party.

16.2 EXHIBITS AND SCHEDULES. All Exhibits and Schedules attached hereto and the documents and agreements referred to herein to be delivered and the acts to be performed at or subsequent to the Closing (collectively, the "ITEMS") are incorporated herein and expressly made a part of this Agreement as fully as though completely set forth herein.

16.3 SPECIFIC PERFORMANCE. CP, CP Canada, FS, the Minority CAT Shareholders, on the one hand, and Buyer, on the other hand, acknowledge that the other will have

no adequate remedy at law if it fails to perform any of its obligations under this Agreement and the Ancillary Agreements. In such event, the performing party shall have the right, in addition to any other rights it may have, to seek specific performance of this Agreement and the Ancillary Agreements and/or injunctive relief.

- 16.4 COUNTERPARTS. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all, of the Parties and the Minority CAT Shareholders. In pleading or proving any provision of this Agreement, it shall not be necessary to produce more than one such counterpart.
- 16.5 HEADINGS; INTERPRETATION. (a) The headings contained in this Agreement are inserted for convenience of reference only and shall not otherwise affect the meaning or interpretation or be deemed a substantive part of this Agreement.
- (b) Except to the extent that the context otherwise requires "include," "includes" and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of like import.
- (c) The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement will refer to this Agreement as a whole, including the Exhibits and Schedules attached hereto, and not to any particular provision of this Agreement, and section and subsection references are to this Agreement unless otherwise specified.
- (d) The meanings given to terms defined herein will be equally applicable to both the singular and plural forms of such terms.
- (e) All matters to be agreed to by any party hereunder must be agreed to in writing by such party unless otherwise indicated herein.
- 16.6 WAIVER. The failure of any Party or Minority CAT Shareholder at any time or times to enforce or require performance of any provision hereof shall in no way operate as a waiver or affect the right of such Party or Minority CAT Shareholder at a later time to enforce the same. No waiver by any Party or Minority CAT Shareholder of any condition or the breach of any term, covenant, representation or warranty contained in this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such condition or breach, or a waiver of any other condition or of any term, covenant, representation or warranty contained in this Agreement. Any agreement on the part of a Party or Minority CAT Shareholder hereto to a waiver shall be valid only if set forth in an instrument in writing signed by such Party or Minority CAT Shareholder.

16.7 SEVERABILITY. If any provision of this Agreement shall hereafter be held to be invalid or unenforceable for any reason, that provision shall be reformed to the maximum extent permitted to preserve the original intent of the Parties and the Minority CAT Shareholders, failing which, it shall be severed from this Agreement with the balance of this Agreement continuing in full force and effect. Such occurrence shall not have the effect of rendering the provision in question invalid in any other jurisdiction or in any other case or circumstances, or of rendering invalid any other provisions contained herein to the extent that such other provisions are not themselves actually in conflict with any applicable law.

16.8 GOVERNING LAW AND FORUM. This Agreement and the Ancillary Agreements and the rights and duties of the Parties and the Minority CAT Shareholders hereunder and thereunder shall be governed by, and construed in accordance with, the law of the State of New York applicable to contracts made and to be performed therein. Each of the Parties and the Minority CAT Shareholders waives any objection that it may have to the venue of any suit, action or proceeding with respect to this Agreement and the Ancillary Agreements or the transactions contemplated hereby or thereby in the courts of the State of New York or the courts of the United States of America, in each case located in the Borough of Manhattan, City of New York and State of New York, or that such suit, action or proceeding brought in the courts of the State of New York or the courts of the United States of America, in each case located in the Borough of Manhattan, City of New York and State of New York, was brought in an inconvenient court and agrees not to plead or claim the same. Each of the Parties and the Minority CAT Shareholders agrees to service of process by registered mail, return receipt requested, or by any other manner provided by New York law.

 $\hbox{IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their duly authorized representatives as of the day and year}\\$ first above written.

PHILLIPS-VAN HEUSEN CORPORATION

By: /s/ Mark D. Fischer

------Name: Mark D. Fischer Title: Vice President

CLUETT, PEABODY & CO., INC.

By: /s/ Bryan P. Marsal

> Name: Bryan P. Marsal Title: Chairman

CLUETT, PEABODY CANADA, INC.

/s/ Bryan P. Marsal By:

Name: Bryan P. Marsal

Title: Chairman

ARROW FACTORY STORES INC.

/s/ Bryan P. Marsal By:

Name: Bryan P. Marsal

Title: Chairman

CLUETT DESIGNER GROUP, INC.

/s/ Bryan P. Marsal By:

Name: Bryan P. Marsal

Title: Chairman

CONSUMER DIRECT CORPORATION

By: /s/ Bryan P. Marsal

Name: Bryan P. Marsal Title: Chairman

CLUETT, PEABODY HOLDING CORP.

By: /s/ Bryan P. Marsal

Name: Bryan P. Marsal Title: Chairman

GUARANTEE

Cluett American Corp., a Delaware corporation ("CAC"), hereby irrevocably and unconditionally guarantees the payment of any amount payable by Cluett, Peabody & Co., Inc., a Delaware corporation ("CP"), under the Purchase and Sale Agreement (defined below) to Phillips-Van Heusen Corporation, a Delaware corporation ("Buyer"), which is not paid by CP when due. For purposes of this Guarantee, the "PURCHASE AND SALE AGREEMENT" refers to the Purchase and Sale Agreement dated as of June 12, 2000 among CP, Cluett, Peabody Canada, Inc., a Canadian corporation, Arrow Factory Stores Inc., a Delaware corporation, Cluett Designer Group, a Delaware corporation, Consumer Direct Corporation, a Delaware corporation, Cluett Peabody Holding Corp., and Buyer. Notwithstanding the foregoing, if (i) CAC and its affiliates cease to be affiliates of CP, (ii) the entity that controls CP upon the occurrence of the event described in clause (i) executes and delivers to Buyer a guarantee identical in all material respects to this Guarantee (the "SUBSTITUTE GUARANTEE") and (iii) the entity referred to in clause (ii) is not less creditworthy than CAC, Buyer shall accept the Substitute Guarantee and this Guarantee thereafter shall have no force or effect.

CLUETT AMERICAN CORP.

By: /s/ Bryan P. Marsal

Name: Bryan P. Marsal Title: Chairman

SHARE PURCHASE AGREEMENT

Among

CLUETT, PEABODY & CO., INC.,

BIDERMANN TAILORED CLOTHING INC.,

CLUETT DESIGNER GROUP INC.,

and

PHILLIPS-VAN HEUSEN CORPORATION

dated as of June 28, 2000

TABLE OF CONTENTS

ARTICLE	1.
1.1	DEFINITIONS. 2 Definitions. 2
ARTICLE	2.
2.1 2.2 2.3 2.4	PURCHASE AND SALE OF SHARES. 4 Transfer by Sellers of Shares. 4 Other Transactions. 4 Obtaining Permits and Licenses 5 Further Assurances. 5
ARTICLE :	3.
3.1 3.2	CONSIDERATION. 5 Purchase Price 5 Inventory. 5
ARTICLE 4	4.
4.1 4.2 4.3	INDEMNITIES. 8 Indemnity. 8 Indemnification Procedure. 9 Other Indemnification Provisions. 10
ARTICLE !	5.
5.1	EMPLOYEES AND EMPLOYEE BENEFITS. 10 Employee Relations and Benefits. 10
ARTICLE	6.
6.1 6.2 6.3	REPRESENTATIONS AND WARRANTIES OF SELLERS

PAGE

6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 6.12 6.13 6.14	Due Authorization; No Breach. 14 Title and Condition of Assets. 14 Consents. 15 Compliance With Laws. 15 Permits and Licenses. 15 Contracts. 15 Tax Matters. 15 Finders; Brokers. 17 Litigation, Claims and Proceedings. 17 Employee Matters. 17 Survival of Representations and Warranties. 18
ARTICLE	7.
7.1 7.2 7.3 7.4 7.5 7.6 7.7	REPRESENTATIONS AND WARRANTIES OF BUYER. 18 Corporate Existence. 18 Due Authorization; No Breach. 18 Consents. 19 Purchase for Investment. 19 Finders; Brokers. 20 Availability of Funds. 20 No Other Representations or Warranties 20 Survival of Representations and Warranties 20
ARTICLE	8.
8.1 8.2 8.3 8.4 8.5 8.6 8.7	COVENANTS. 20 Conduct of the Business. 20 Best Efforts and Alternative Proposals. 22 Notification. 22 Letter of Credit. 23 Returned Inventory. 24 Accounts Receivable. 24 Name Change. 25
ARTICLE	9.
9.1	CONDITIONS OF CLOSING

	PAGE
9.3	Additional Conditions of Buyer
ARTICLE	10.
10.1 10.2	TERMINATION; SURVIVAL
ARTICLE	11.
11.1 11.2 11.3	CLOSING
ARTICLE	12.
12.1 12.2 12.3 12.4	EXPENSES AND POST-CLOSING OBLIGATIONS
ARTICLE	13.
13.1 13.2	PUBLICITY, CONFIDENTIALITY
ARTICLE	14.
14.1	NOTICES
ARTICLE	15.
15.1 15.2 15.3	MISCELLANEOUS

15.4	Counterparts	3
	Headings; Interpretation	
	Waiver	
	Severability	
15.8	Governing Law and Forum	3

PAGE

EXHIBITS

Exhibit 2.2(a) Escrow Agreement
Exhibit 2.2(c) Waiver and Consent
Exhibit 2.2(d) Hosiery Agreement

٧

SCHEDULES

Schedule 4.1(a)	_	Assumed Liabilities
Schedule 5.1(a)(i)	_	CDG Employees
` , ` ,	-	• •
Schedule 5.1(a)(ii)	-	Severance Benefits
Schedule 5.1(h)	-	Vacation Policy
Schedule 6.1	-	Jurisdiction Where Qualified to do Business
Schedule 6.6	-	CDG Required Consents
Schedule 6.7	-	Compliance with Laws
Schedule 6.8	-	Permits and Licenses
Schedule 6.9	-	Contracts in Default
Schedule 6.10(a)	-	Tax Filings
Schedule 6.10(b)	-	Tax Defaults
Schedule 6.10(c)	-	Tax Extensions
Schedule 6.10(d)	-	Tax Suits
Schedule 6.10(e)	-	Tax Liens
Schedule 6.12	-	Litigation, Claims and Proceedings
Schedule 6.13(d)	-	Employee Benefit Plans
Schedule 7.2	-	Due Authorization; No Breach
Schedule 7.3	-	Buyer's Required Consents
Schedule 8.2	-	Assets to be Assigned

SHARE PURCHASE AGREEMENT

This Share Purchase Agreement (this "AGREEMENT"), dated as of June 2000, is entered into by and among CLUETT, PEABODY & CO., INC., a Delaware corporation ("CP"), CLUETT DESIGNER GROUP INC., a Delaware corporation ("CDG"), BIDERMANN TAILORED CLOTHING INC., a Delaware corporation ("BTC", and collectively with CP, the "SELLERS"), and PHILLIPS-VAN HEUSEN CORPORATION, a Delaware corporation ("BUYER").

WHEREAS, CDG is engaged, among other things in the business of designing, marketing and selling dress shirts with the Kenneth Cole, Kenneth Cole New York, Kenneth Cole Collection and Kenneth Cole Reaction labels (the "BUSINESS") pursuant to a license agreement made as of August 1, 1997 between K.C.P.L. Inc., a Delaware corporation ("KCPL"), and CDG, as amended (the "KCPL SHIRT LICENSE AGREEMENT");

WHEREAS, CP, Cluett, Peabody Canada, Inc., a Canadian corporation, Arrow Factory Stores Inc., a Delaware corporation ("FS"), CDG, Consumer Direct Corporation, a Delaware corporation, Cluett Peabody Holding Corp., a Delaware corporation, and Buyer have entered into a purchase and sale agreement (the "PURCHASE AGREEMENT") which, among other things, contemplates that the parties will enter into this Agreement;

WHEREAS, pursuant to the Purchase Agreement, CP, Buyer and Cluett Peabody Resources Corporation, a Delaware corporation ("RESOURCES"), will enter into the Trademark License Agreement (as defined in the Purchase Agreement);

WHEREAS, pursuant to the Purchase Agreement, CP, Buyer and FS will enter into the Transition Services Agreement (as defined in the Purchase Agreement);

WHEREAS, Sellers collectively own, beneficially and of record, all of the issued and outstanding equity securities of CDG (the "SHARES"); and

WHEREAS, Buyer desires to purchase from Sellers, and Sellers desire to sell to Buyer, all of the Shares upon the terms and conditions set forth herein, for purposes of acquiring the Business and CDG's rights under the KCPL Shirt License Agreement.

NOW, THEREFORE, in consideration of the foregoing premises and the mutual covenants contained herein, the parties hereto agree as follows:

ARTICLE 1.

DEFINITIONS

1.1 DEFINITIONS. The following terms are defined in the Section indicated:

TERM	SECTION
Agreement. Alternative Proposal. Ancillary Agreements. Arbiter. Assets. BTC. Business. Buyer. Buyer Indemnified Parties. Buyer Plans. Buyer's 401(k) Plan. Capital Stock Equivalents. CDG. CDG's Cost. CDG Employees. Claim. Closing. Closing Date.	
Confidentiality Agreement	
Contract	

Encumbrances2.1
ERISA5.1(d)
Escrow Agreement2.2(a)
Estimated Inventory Purchase Price
Final Inventory Purchase Price3.2(d)
FSRecitals
Governmental Authority6.10(i)
HSR Act6.6
Indemnified Party4.2
Indemnifying Party4.2
Issuing Bank8.4
Items
July 9 Count
KCPLRecitals
KCPL Inventory
KCPL Inventory Closing Date Statement
KCPL Inventory Escrow3.2(a)
KCPL Letter of Credit8.4
KCPL Shirt License AgreementRecitals Liability4.1(a)
Pension Plan
Permits6.8
Purchase Agreement
Purchase Price
Resources
Returned Inventory8.5
Securities Act
Seller Indemnified Parties4.1(b)
20110. 1

SellersIntroduc	tion
Sellers' Knowledge	.6.9
SharesReci	tals
Sourcing Commitments	.8.4
Tax Returns6.1	9(g)
Taxes6.1	9(h)
Transferred Employees5.:	1(a)
Vacation Policy5.	1(h)
WARN	1(i)

ARTICLE 2.

PURCHASE AND SALE OF SHARES

- 2.1 TRANSFER BY SELLERS OF SHARES. On the Closing Date (as defined in Section 2.3) and subject to the terms and conditions set forth in this Agreement, Sellers will sell, assign, transfer and deliver to Buyer the Shares, free and clear of all options, pledges, mortgages, security interests, liens, restrictions on voting or transfer or other encumbrances of any nature ("ENCUMBRANCES").
- 2.2 OTHER TRANSACTIONS. In addition to the transactions contemplated above, the following acts or transactions shall also occur at or before the Closing:
 - (a) Buyer, Sellers and The Bank of New York shall each execute and deliver an Escrow Agreement substantially in the form attached hereto as Exhibit 2.2(a) (the "ESCROW AGREEMENT");
 - (b) Sellers shall deliver UCC-3 partial releases and/or termination statements and such other instruments as are required to release all Encumbrances of the lenders and other creditors of CDG on CDG's assets;
 - (c) CDG and, to the extent relevant, its affiliates (including the Sellers) shall execute and deliver, and CDG and Sellers shall use their reasonable efforts to cause CDG's contractors, subcontractors and warehousemen to deliver, a Waiver and Consent substantially in the form attached hereto as Exhibit 2.2 (c); and

(d) Buyer shall execute and deliver a Hosiery Agreement substantially in the form attached hereto as Exhibit 2.2(d).

The agreements set forth in paragraphs (a) through (c) of this Section 2.2 are hereinafter referred to as the "ANCILLARY AGREEMENTS."

- 2.3 OBTAINING PERMITS AND LICENSES. Buyer shall use all reasonable efforts to obtain as of the Closing or as soon thereafter as may be practicable all permits and licenses required by any governmental agency with respect to the Business. Sellers shall provide any cooperation reasonably requested by Buyer to assist Buyer in obtaining such permits and licenses.
- 2.4 FURTHER ASSURANCES. After the Closing, each party hereto shall from time to time, at the request of any other party and without further cost or expense to such other party, execute and deliver such other instruments of conveyance and transfer and take such other actions as such other party may reasonably request in order to more effectively consummate the transactions contemplated hereby and to vest in Buyer good and valid title to the Shares.

ARTICLE 3.

CONSIDERATION

- 3.1 PURCHASE PRICE. Buyer shall pay Sellers for the Shares an amount equal to the Final Inventory Purchase Price (as defined in Section 3.2(d)). All payments made to Sellers pursuant to this Section 3.1 (all such payments, collectively, the "PURCHASE PRICE") shall be made by wire transfer of immediately available funds to an account designated in writing by Sellers at least five business days prior to the Closing Date. All payments made to Sellers pursuant to this Section 3.1 shall be made at Closing, except as expressly provided otherwise in Section 3.2.
- 3.2 INVENTORY. (a) Ten days prior to the Closing Date, Sellers shall deliver to Buyer their good faith estimate, prepared on a basis consistent with CDG's past practices, of CDG's Cost of the KCPL Inventory (defined below) as of the close of business on the Closing Date. The estimate referred to in the preceding sentence is referred to as the "ESTIMATED INVENTORY PURCHASE PRICE". Buyer shall pay Sellers at Closing 95% of the Estimated Inventory Purchase Price. The balance of the Estimated Inventory Purchase Price (the "KCPL INVENTORY ESCROW") shall be deposited in escrow by Buyer in accordance with the Escrow Agreement. For purposes hereof, "CDG'S COST" shall mean CDG's cost as calculated in accordance with generally accepted accounting principles applied in a manner consistent with the preparation of the audited financial statements of Cluett American Corp. for the fiscal year ended December 31, 1999. For purposes of this Agreement, "KCPL INVENTORY" means: all raw materials, work in process, trim (including, without limitation, packing, labels, buttons, zippers and hang tags) and finished goods inventory relating to the Business, in each case including, without limitation, any

of such items in transit from manufacturing facilities of CDG and its affiliates or third parties or warehouses of CDG and its affiliates. Notwithstanding anything to the contrary in the foregoing, KCPL Inventory in transit will be treated pursuant to this Section 3.2(a) based on the date of receipt and the condition upon receipt. The parties to this Agreement agree to cooperate in resolving any disputes with third party vendors with respect to such KCPL Inventory.

- (b) As soon as practicable following the Closing Date and in any event within 60 calendar days thereof, Sellers shall prepare and deliver to Buyer a statement of CDG's Cost of the KCPL Inventory as of the close of business on the Closing Date (including the notes thereto, the "KCPL INVENTORY CLOSING DATE STATEMENT"), together with the work papers used by Sellers (and, subject to the execution and delivery by Buyer of a customary accountants' access letter, including all work papers used by Sellers' independent public accountants) in the preparation thereof. There shall be attached to the KCPL Inventory Closing Date Statement an annex setting forth in reasonable detail the computation of CDG's Cost of the KCPL Inventory.
- (c) The KCPL Inventory Closing Date Statement shall be based upon a full physical count of the KCPL Inventory as of the close of business on July 9, 2000, which shall be conducted by CDG and observed and participated in by Buyer and its representatives (the "JULY 9 COUNT"), as adjusted for items sold after the July 9 Count, items received after the July 9 Count and items manufactured after the July 9 Count. Buyer and Sellers shall share equally the cost of the July 9 Count. As promptly as practicable following the July 9, Count, Sellers shall deliver to Buyer copies of all work papers used by Sellers (and, subject to the execution and delivery by Buyer of a customary accountants' access letter, including all work papers used by Sellers' independent public accountants) in connection with the July 9 Count. Sellers shall, and shall cause Sellers' independent public accountants to, be available to Buyer and its representatives to answer questions regarding the July 9 Count or such work papers.
- (d) Unless Buyer delivers written notice to Sellers on or prior to the 45th day following Buyer's receipt of the KCPL Inventory Closing Date Statement and the work papers used in the preparation thereof of Buyer's objection to any amount set forth in the KCPL Inventory Closing Date Statement, specifying in reasonable detail all disputed amounts and the basis therefor, Buyer shall be deemed to have accepted and agreed to the KCPL Inventory Closing Date Statement. If Buyer so notifies Sellers, Sellers and Buyer shall have 15 business days thereafter to resolve their differences, and each of Buyer and Sellers shall make a good faith effort to resolve their differences. If any amount remains in dispute, all such amounts remaining in dispute shall be submitted for resolution to a "Big 5" accounting firm mutually agreed upon by Buyer and each of the Sellers which has not had a material relationship with either Buyer and its subsidiaries or either of the Sellers and their respective subsidiaries within the two years preceding the appointment. If Buyer and Sellers cannot agree on the arbiter, they shall request that their respective independent certified public accountants jointly designate the arbiter, and such appointment shall be binding and conclusive on the parties hereto. The arbiter designated by Buyer and Sellers or their respective independent certified public accountants, as the case may

be, is hereinafter referred to as the "ARBITER." Within 30 business days after the Arbiter's acceptance of its appointment under this Section 3.2(d) as Arbiter, each of the Sellers and Buyer shall submit to the Arbiter (and each other) a single presentation setting forth its proposed resolution of the dispute and the basis therefor and the resulting computation of the KCPL Inventory Closing Date Statement and CDG's Cost of the KCPL Inventory (collectively, the "FINAL INVENTORY PURCHASE PRICE"). Within 20 business days after the earlier to occur of (1) the Arbiter's receipt of the presentations referred to in the preceding sentence or (2) the expiration of the 30 business day period referred to in the preceding sentence, the Arbiter shall accept either Buyer's or Sellers' proposed resolution of the dispute and the resulting computation of the KCPL Inventory Closing Date Statement and CDG's Cost of the KCPL Inventory, and shall notify each of Buyer and Sellers of its decision in writing. Such decision shall be final, binding and conclusive on all parties hereto. The party whose proposed resolution is not accepted shall pay all of the Arbiter's fees and expenses with respect to its engagement under this Section 3.2(d). Each party agrees to execute, if requested by the Arbiter, a reasonable engagement letter with the Arbiter with respect to its engagement under this Section 3.2(d).

(e) If the Final Inventory Purchase Price as determined above equals or exceeds the Estimated Inventory Purchase Price, the KCPL Inventory Escrow shall be delivered to Sellers and Buyer shall pay Sellers by wire transfer of immediately available funds to an account designated in writing by Sellers an amount equal to any excess, together with interest on such excess at an annual rate equal to the reference rate from time to time of The Chase Manhattan Bank from and including the Closing Date to but not including the date of payment. If the Final Inventory Purchase Price is less than the Estimated Inventory Purchase Price but greater than 95% thereof, the portion of the KCPL Inventory Escrow equal to any such excess shall be delivered to Sellers and the balance shall be delivered to Buyer. If the Final Inventory Purchase Price is equal to 95% of the Estimated Inventory Purchase Price, the KCPL Inventory Escrow shall be delivered to Buyer. If the Final Inventory Purchase Price is less than 95% of the Estimated Inventory Purchase Price, the KCPL Inventory Escrow shall be delivered to Buyer, and Sellers shall pay Buyer an amount equal to such difference between 95% of the Estimated Inventory Purchase Price and the Final Inventory Purchase Price by wire transfer of immediately payable funds to an account designated in writing by Buyer, together with interest on such amount at an annual rate equal to the reference rate from time to time of The Chase Manhattan Bank from and including the Closing Date to but not including the date of payment. All Net Investment Proceeds in the Inventory Escrow Fund (as those terms are defined in the Escrow Agreement) shall be disbursed to Buyer and Sellers, respectively, in the same proportion that the KCPL Inventory Escrow is disbursed. Any amounts payable pursuant to this Section 3.2(e) in addition to the Inventory Escrow Fund shall be made on the third business day following (i) the 30th day after the delivery to Buyer of the KCPL Inventory Closing Date Statement and the work papers used in the preparation thereof, to the extent that there shall have been no dispute between Sellers and Buyer with respect to the KCPL Inventory Escrow and the KCPL Inventory Closing Date Statement, and/or (ii) if a dispute with respect to the KCPL Inventory Escrow or the KCPL

Inventory Closing Date Statement is settled by the parties without resort to arbitration, the date of such settlement and/or (iii) the issuance of the Arbiter's written decision under Section 3.2(d).

ARTICLE 4.

INDEMNITIES

- $4.1\ \text{INDEMNITY}.$ (a) Sellers hereby agree, jointly and severally, to indemnify and hold harmless each of Buyer and its affiliates and their respective directors, shareholders, partners, officers, employees, agents, consultants, representatives, successors, transferees and assigns (hereinafter sometimes collectively referred to as "BUYER INDEMNIFIED PARTIES") against any Liabilities of CDG which are not set forth on Schedule 4.1(a), including, without limitation, any Liability under any law, rule or regulation now or hereinafter in effect relating to environmental matters and any liability loss, damage (excluding consequential damages (including, without limitation, lost business profits)) and other out-of-pocket costs (including reasonable fees and expenses of outside counsel) incident to proceedings or investigations or the prosecution or defense of any claim (collectively, "DAMAGES") incurred by any of them relating to any such Liability. The term "LIABILITY" shall mean and include any direct or indirect indebtedness, liability, claim, loss, damage, deficiency, cost of environmental investigation or remediation, obligation or responsibility, of whatever kind or nature, whether known or unknown, fixed or unfixed, choate or inchoate, liquidated or unliquidated, secured or unsecured, accrued or unaccrued, absolute, contingent or otherwise. Liabilities for Taxes are exclusively provided for in Section 12.1.
- (b) Buyer shall indemnify and hold harmless each of the Sellers and their affiliates and the directors, shareholders, partners, officers, employees, agents, consultants, representatives, successors, transferees and assigns of any thereof (hereinafter sometimes collectively referred to as "SELLER INDEMNIFIED PARTIES") against any Damages which are caused by or arise out of any breach of any covenant or agreement of Buyer hereunder or under any Ancillary Agreement. It is agreed and understood that claims under this Section 4.1(b) may be brought by Seller Indemnified Parties at any time and shall not be limited as to dollar amount.
- (c) Sellers shall also, jointly and severally, indemnify and hold harmless each Buyer Indemnified Party against any Damages (i) which are caused by or arise out of any breach of any covenant or agreement of Sellers hereunder or under any Ancillary Agreement or (ii) which are caused by or arise out of any breach of any representation or warranty of Sellers contained in the first and third sentences of Section 6.1 or in Section 6.2, Section 6.3, Section 6.4 or Section 6.5. No claim for breach of a representation or warranty shall be asserted under Section 4.1(c)(ii) after the expiration of the survival period thereof under Section 6.14. The maximum amount recoverable by Buyer Indemnified Parties for indemnification for breaches of representations and warranties under Section 4.1(c)(ii) shall be \$500,000 in the aggregate. It is

agreed and understood that claims under Section 4.1(a) and Section 4.1(c)(i) may be brought by Buyer Indemnified Parties at any time and shall not be limited as to dollar amount.

- (d) Any indemnification amounts payable to Buyer Indemnified Parties or Seller Indemnified Parties hereunder shall be paid to them by wire transfer of immediately available funds to a bank account designated by them in writing.
- (e) The indemnification obligations of Sellers and Buyer set forth in this Section 4.1 shall constitute the sole and exclusive remedy of the Buyer Indemnified Parties and Seller Indemnified Parties for the recovery of any Damages, except for Damages which are caused by or arise out of the fraud of the other party.
- (f) Nothing in this Agreement shall be deemed to limit Buyer's rights under Section 8.11 of the Purchase Agreement if CDG is prevented from exercising its rights as the licensee under the KCPL Shirt License Agreement.
- 4.2 INDEMNIFICATION PROCEDURE. The obligation of a party (the "INDEMNIFYING PARTY") to indemnify any person or entity (the "INDEMNIFIED PARTY") under Section 4.1 hereof is conditioned upon receiving from the Indemnified Party written notice of the assertion or institution of a claim arising from or related to any liability set forth in Section 4.1 hereof (a "CLAIM") or of the occurrence of an event which the Indemnified Party reasonably believes could lead to the assertion of a Claim, specifying in reasonable detail the nature and amount of such Claim, promptly after the Indemnified Party becomes aware of such Claim or event; provided, however, that the failure of the Indemnifying Party to receive such notice on a timely basis shall relieve the Indemnifying Party of its obligation to indemnify hereunder only if and to the extent that such failure is prejudicial to its ability to defend such Claim. Subject to the terms hereof, the Indemnifying Party shall have the absolute right, in its sole discretion and at its sole expense, to elect to defend, settle or otherwise protect against any Claim with legal counsel of its own selection reasonably satisfactory to the Indemnified Party, provided, however, that no Claim may be settled by the Indemnifying Party without the consent of the Indemnified Party, which consent shall not be unreasonably withheld. The Indemnified Party shall have the right, but not the obligation, to participate, at its own expense, in the defense of any Claim through counsel of its own and the fees and expenses of such counsel will be at the expense of such Indemnified Party unless (i) the Indemnifying Party specifically authorized the employment of such counsel and specifically agreed to pay such counsel's fees, (ii) based on the advice of counsel, there is a conflict of interest between the position of the Indemnifying Party on the one hand and the Indemnified Party on the other hand, or (iii) the Indemnifying Party fails to assume the defense or fails to contest such action in good faith, in any which case, if the Indemnified Party notifies the Indemnifying Party that it elects to employ separate counsel, the Indemnifying Party will not have the right to assume the defense of such Claim on behalf of the Indemnified Party and the reasonable fees and expenses of such separate counsel shall be borne by the Indemnifying Party. The Indemnified Party shall, and shall cause its affiliates to, at all times

cooperate in all reasonable ways with, make its relevant files and records available for inspection and copying by, and make (subject to assertion of attorney-client and other applicable privileges) its employees available or otherwise render reasonable assistance to the Indemnifying Party in connection with its defense of any Claim. Subject to the next sentence, if the Indemnified Party, without the prior consent of the Indemnifying Party (which shall not be unreasonably withheld or delayed), makes any settlement with respect to any Claim, the Indemnifying Party shall be discharged from all obligations under Section 4.1 hereof with respect to such Claim. In the event the Indemnifying Party does not undertake the defense against, settlement of or protection against any Claim in accordance with this Section 4.2, the Indemnified Party shall have the right, but not the obligation, to defend, contest, assert cross claims or counterclaims or otherwise protect against the same, to make any compromise or settlement thereof, with the consent of the Indemnifying Party which shall not be unreasonably withheld or delayed, and to recover from the Indemnifying Party and be indemnified by the Indemnifying Party for the entire cost thereof, including, without limitation, legal expenses, disbursements and all amounts paid as a result of such Claim or the compromise or settlement thereof.

4.3 OTHER INDEMNIFICATION PROVISIONS. The parties to this Agreement shall make appropriate adjustments for insurance proceeds actually recovered by or on behalf of an Indemnified Party in respect of an indemnifiable liability in determining the amount of Damages pursuant to any Claims asserted under this Article 4. All indemnification payments made pursuant to this Article 4 shall be considered adjustments to the Purchase Price for federal income tax purposes.

ARTICLE 5.

EMPLOYEES AND EMPLOYEE BENEFITS

- 5.1 EMPLOYEE RELATIONS AND BENEFITS. (a) The parties hereto intend that there shall be continuity of employment for all CDG employees listed on Schedule 5.1(a)(i) ("CDG EMPLOYEES"). Buyer will be responsible for the payment of all severance benefits, if any, consistent with severance practices set forth on Schedule 5.1(a)(ii), and other costs related to any termination of the CDG Employees by Buyer on or after the Closing Date. CDG Employees who are employed by CDG at Closing shall be referred to as "TRANSFERRED EMPLOYEES."
- (b) As of the Closing Date, each Transferred Employee shall participate in employee benefit plans, agreements, programs, policies and arrangements of Buyer that are no less favorable in the aggregate than those that cover similarly situated employees of Buyer ("BUYER PLANS"). Notwithstanding the above, with respect to medical and dental welfare benefits, Transferred Employees will continue to receive coverage under CP's medical and dental plans from the Closing Date through the end of the calendar month in which the Closing Date

occurs, with respect to claims incurred during such period. On the first day of the calendar month immediately following the Closing Date, Transferred Employees shall participate in Buyer's medical and dental plans, and such plans shall be responsible for any claims thereafter.

- (c) CP shall or shall cause the applicable plan to retain responsibility for and continue to pay all medical, life insurance, disability and other welfare plan expenses and benefits for each Transferred Employee with respect to claims incurred by such employees or their covered dependents subject to and in accordance with the terms of the applicable plan for payment of such expenses and benefits prior to the Closing Date. Expenses and benefits with respect to claims incurred by Transferred Employees or their covered dependents on or after the Closing Date shall be the responsibility of Buyer. For purposes of this paragraph, a claim is deemed incurred when the services that are the subject of the claim are performed; in the case of life insurance, when the death occurs, in the case of long-term disability benefits, when the disability occurs and, in the case of a hospital stay, when the employee first enters the hospital.
- (d) With respect to any Buyer Plan that is a "welfare benefit plan" (as defined in Section 3(1) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA")) or any Buyer Plan that would be a "welfare benefit plan" (as defined in Section 3(1) of ERISA) if it were subject to ERISA, Buyer shall (i) cause there to be waived any pre-existing condition applicable to the Transferred Employees, to the extent waived under CDG's similar welfare benefit Plan, (ii) give effect for the year in which the Closing Date occurs, in determining any deductible and maximum out-of-pocket limitations, to claims incurred and amounts paid by, and amounts reimbursed to, such Transferred Employees with respect to similar CDG Plans, as in effect immediately prior to the Closing Date and (iii) recognize all credited service to the same extent such service was recognized under similar CDG Plans immediately prior to the Closing Date.
- (e) As of the Closing Date with respect to each Transferred Employee, CP shall cause the participation by each such employee in the Cluett Retirement Plan (the "PENSION PLAN") to cease and shall cause the accrued benefit of each such Transferred Employee in the Pension Plan to fully vest. No assets or liabilities with respect to the Transferred Employees shall be transferred as a result of this Agreement from the Pension Plan to any plan or arrangement established by Buyer or any other employer for the benefit of the Transferred Employees (other than pursuant to an eligible rollover distribution). Benefits payable to the Transferred Employees under the Pension Plan through the Closing Date shall be payable to such Transferred Employees pursuant to the terms of, and at the time and in the amounts provided under, the Pension Plan based upon such Transferred Employees' years of service with, and compensation received from, CDG through the Closing Date (including periods of employment with any other employer which is taken into account under the Pension Plan). No period of service by any Transferred Employee with Buyer or any other employer and no compensation earned by any Transferred Employee for services performed for Buyer or any other employer shall be taken into account under the Pension Plan as a result of this Agreement except as expressly provided above.

- (f) As of the Closing Date with respect to each Transferred Employee, CP shall (i) cause the account balance of each Transferred Employee in the Cluett Savings Plan ("CP'S 401(K) PLAN") to fully vest, (ii) cause the participation by each Transferred Employee in the CP's 401(k) Plan to cease and (iii) make distributions to such Transferred Employees in accordance with the terms of the CP's 401(k) Plan and applicable law.
- (g) Buyer shall take all steps necessary to permit each such Transferred Employee who has received an eligible rollover distribution (as defined in Section 402(c)(4) of the Code) from the Pension Plan or CP's 401(k) Plan, if any, to roll such eligible rollover distribution into an account under the Buyer's Associate Investment Plan (the "BUYER'S 401(K) PLAN"). Periods of employment by Transferred Employees with CDG for which credit was given under the CP's 401(k) Plan shall be taken into account for purposes of eligibility and vesting (but not benefit accrual) under the Buyer's 401(k) Plan and any Buyer Plan that is a "pension plan" (as defined in section 3(2) of ERISA).
- (h) With respect to any accrued but unused vacation time to which any Transferred Employee is entitled pursuant to the vacation policy applicable to such Transferred Employee immediately prior to the Closing Date (the "VACATION POLICY"), which Vacation Policy is set forth on Schedule 5.1(h), Buyer shall assume the liability for such accrued vacation and allow such Transferred Employee to use such accrued vacation; PROVIDED, HOWEVER, that if Buyer deems it necessary to disallow such Transferred Employee from taking such accrued vacation, Buyer shall be liable for and pay in cash to each such Transferred Employee an amount equal to such accrued but unused vacation time in accordance with the terms of the Vacation Policy; and PROVIDED, FURTHER, that Buyer shall be liable for and pay in cash an amount equal to such accrued but unused vacation time to any Transferred Employee whose employment terminates for any reason subsequent to the Closing Date, subject to the terms of the Vacation Policy. Nothing herein shall be deemed to require Buyer to continue the Vacation Policy on or after the Closing Date for the Transferred Employees.
- (i) Buyer agrees to provide any required notice under the Worker Adjustment and Retraining Notification Act of 1988, as amended ("WARN"), and any other applicable law and to otherwise comply with any such statute with respect to any "plant closing" or "mass layoff" (as defined in WARN) or similar event affecting any Transferred Employees and occurring on or after the date such person becomes a Transferred Employee. Buyer shall indemnify and hold harmless the Seller Indemnified Parties with respect to any liability under WARN or other applicable law arising from the actions (or inactions) of Buyer or its affiliates with respect to the termination of the Transferred Employees by Buyer on or after the Closing Date.
- (j) Nothing herein, expressed or implied, shall confer upon any employee or former employee of CDG, Buyer, Sellers or any of their affiliates (including, without limitation, the Transferred Employees), any rights or remedies (including, without limitation, any right to

employment or continued employment for any specified period) of any nature or kind whatsoever, under or by reason of this Agreement.

(k) CP shall use reasonable efforts to provide to Buyer Corporate Services and Non-Corporate Services (as those terms are defined in the Transition Services Agreement, as amended hereby) relating to the Business pursuant to the Transition Services Agreement on the terms and subject to the conditions set forth therein, as amended hereby.

ARTICLE 6.

REPRESENTATIONS AND WARRANTIES OF SELLERS

Sellers hereby jointly and severally represent and warrant to uyer that:

- 6.1 ORGANIZATION AND CORPORATE POWER. Each of the Sellers and CDG is a corporation duly organized, validly existing and in good standing under the laws of Delaware. CDG has all requisite power and authority to own, lease and operate the properties and assets owned, leased and/or operated by it and to carry on the Business as now being conducted. Each of the Sellers and CDG has full corporate power and authority to enter into and perform this Agreement and each Ancillary Agreement to which it is a party. CDG is duly qualified to do business as a foreign corporation and is in good standing in each jurisdiction where the character of its properties owned or held under lease or the nature of its activities make such qualification necessary, except where the failure to so qualify would not have a material adverse effect on CDG or its ability to consummate the transactions contemplated hereby. All such jurisdictions are listed on Schedule 6.1 hereto. Buyer heretofore has been provided with correct and complete copies of the Certificate of Incorporation and By-Laws of CDG, as in effect on the date hereof, including all amendments thereto through the date hereof. Sellers have heretofore made available to Buyer correct and complete copies of the minute books of CDG. CDG has not made any investments in or advances to, and (other than as disclosed in the Purchase Agreement) does not own any securities of or other interests in, any other person or otherwise control any other person.
- 6.2 CAPITALIZATION. The authorized share capital of CDG consists of 3,000 shares of common stock, par value \$1.00 per share. Of such authorized shares, only the Shares are issued and outstanding and no other shares of capital stock of CDG or securities exercisable or convertible into or exchangeable for shares of capital stock of CDG ("CAPITAL STOCK EQUIVALENTS") are authorized, issued or outstanding. There are no subscriptions, options, warrants, calls, rights, contracts, commitments, understandings, restrictions or arrangements relating to the issuance, sale, transfer or voting of any shares of capital stock, whether issued or unissued, of CDG or Capital Stock Equivalents, including any rights of issuance, conversion or exchange under any outstanding securities or other instruments. There are no bonds, debentures,

notes or other indebtedness having the right to vote on any matter submitted for a vote of CDG's stockholders. All of the Shares have been validly issued and are outstanding and fully paid, nonassessable and free of preemptive rights.

- 6.3 OWNERSHIP OF SHARES. Sellers are the record and beneficial owner of all of the Shares. The Shares are owned free and clear of all Encumbrances. Upon the consummation of the transactions contemplated hereby, Buyer will acquire title to the Shares free and clear of all Encumbrances, except for Encumbrances arising as a result of any action taken by, or failure to act of, Buyer or any of its affiliates.
- 6.4 DUE AUTHORIZATION; NO BREACH. The execution, delivery and performance by each of the Sellers and CDG of this Agreement and each Ancillary Agreement to which any of them is a party and the consummation of the transactions contemplated hereby and thereby have been approved by their respective Boards of Directors and no further corporate action is required to be taken by any of them and their affiliates in order to execute, deliver and perform this Agreement and the Ancillary Agreements to which any of them is a party and to transfer the Shares to Buyer. This Agreement is a valid and legally binding obligation of each of the Sellers and CDG enforceable against each of them in accordance with its terms subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to the effect of general principles of equity (regardless of whether enforcement is considered in a proceeding at law or in equity), and each Ancillary Agreement and other agreement or instrument contemplated by this Agreement to which any of the Sellers and CDG is a party, when executed and delivered by each of the Sellers and CDG that is a party thereto in accordance with the provisions thereof, will be a valid and legally binding obligation of such party, enforceable against such party in accordance with its terms subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to the effect of general principles of equity (regardless of whether enforcement is considered in a proceeding at law or in equity). All persons who have executed this Agreement on behalf of any of the Sellers and CDG or who will execute on behalf of any of the Sellers and CDG any agreement or instrument contemplated by this Agreement, have been duly authorized to do so by all necessary corporate action. Neither the execution and delivery of this Agreement, any Ancillary Agreement and the other agreements and documents to be executed or delivered pursuant hereto, nor the consummation of the transactions contemplated hereby and thereby, will (i) violate, or conflict with, any provision of the certificate of incorporation or by-laws of any of the Sellers and CDG, (ii) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event of default which with notice or lapse of time or both would become a default) under, or result in the termination (or grant a right of termination) of, cancellation, amendment or accelerate (or grant the right to accelerate) the performance required by, or result in the creation of any Encumbrance upon any of the assets of CDG set forth on Schedule 8.2 or the KCPL Inventory (collectively, the "ASSETS") under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license,

agreement, lease, Permit or other material instrument to which any of the Sellers and CDG is a party or by which any of the Sellers and CDG or any of the Assets may be bound or (iii) violate, or conflict with, any order, writ, injunction, arbitration award, judgment or decree of any court, governmental body or arbitrator applicable to CDG or any applicable statute, law, rule or regulation, except, in the case of clause (ii) or (iii), as would not have a material adverse effect on CDG, the Shares, the Assets or the Business.

- 6.5 TITLE AND CONDITION OF ASSETS. As of the Closing Date, CDG will have good and marketable title to the Assets. Except for Encumbrances created by, or arising as a result of the ownership of the Shares by, Buyer, the Assets will be free and clear of all Encumbrances as of Closing.
- 6.6 CONSENTS. Schedule 6.6 hereto sets forth all actions, approvals, permits, consents or authorizations required to be obtained by CDG in order to consummate the transactions contemplated hereby, other than the expiration of any applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder (the "HSR ACT").
- 6.7 COMPLIANCE WITH LAWS. Except as set forth in Schedule 6.7 hereto, CDG is not in default under or in violation of any United States federal, state or local statute, law or regulation which would give rise to a material Liability or which is reasonably likely to materially and adversely affect CDG, the Shares or the Business as presently conducted.
- 6.8 PERMITS AND LICENSES. Schedule 6.8 attached hereto sets forth all governmental licenses, permits and other governmental authorizations (collectively "PERMITS") which are issued to, held or used by CDG in respect of the ownership or use of the Assets as presently used or the conduct of the Business as currently conducted, including the dates of issuance and expiration thereof and, except as set forth on Schedule 6.8 hereto, there are no other governmental licenses, permits, franchises or authorizations which are material to the operation of the Business as presently conducted. Except as set forth on Schedule 6.8 hereto, within the past 24 months, CDG has not received any written warning, notice of violation or probable violation, notice of revocation or other written communication from or on behalf of any governmental entity, which violation has not been corrected or otherwise settled, alleging (i) any material violation of any Permit listed on Schedule 6.8, (ii) that CDG needs a Permit to conduct the Business as currently conducted or to use the Assets as currently used, in each case which Permit is not currently held by CDG or (iii) any current violation, with respect to the conduct of the Business as presently conducted or the use of the Assets as presently used, of any United States federal, state or local statutes or regulations.
- 6.9 CONTRACTS. Except as set forth in Schedule 6.9, (i) each contract, lease, license, commitment, sales order, purchase order, invoice and other agreement set forth on Schedule 4.1(a) (the "CONTRACT") is a valid agreement, in full force and effect and constitutes a

valid and binding obligation of CDG, and, to Sellers' Knowledge, of any other party thereto, and is legally enforceable, (ii) none of Sellers and CDG has received any written notice from any other party to a Contract of the termination thereof or alleging a material default thereunder by CDG and (iii) there is no material default or event which, with notice or lapse of time or both, would constitute a material default on the part of CDG (nor, to Sellers' Knowledge, on the part of any other party thereto) under any Contract. For the purposes of this Agreement, "SELLERS' KNOWLEDGE" shall mean any fact or set of facts of which any director or officer of either Seller or CDG has actual knowledge.

- 6.10 TAX MATTERS. (a) Except as set forth in Schedule 6.10(a), CDG has duly and timely filed all material Tax Returns that are required to be filed in all jurisdictions where such entities are required to file with the appropriate Governmental Authority (as defined below) and has duly, completely and correctly reported all income and all other amounts and information required to be reported thereon and no material fact has been omitted therefrom.
- (b) Except as set forth in Schedule 6.10(b), CDG has duly and timely paid all material Taxes, including all installments on account of material Taxes for the current year, that are due and payable by it.
- (c) Except as set forth on Schedule 6.10(c), CDG has not requested, or entered into any agreement or other arrangement or executed any waiver providing for, any extension of time within which (i) to file any Tax Return covering any Taxes for which it is or may be liable; (ii) to file any elections, designations or similar things relating to Taxes for which it is or may be liable; (iii) it is required to pay or remit any Taxes or amounts on account of Taxes; or (iv) any Governmental Authority may assess or collect Taxes for which it is or may be liable.
- (d) As of the date hereof, there are no actions, suits, proceedings, investigations, audits or claims that have been commenced or, to the best knowledge of Sellers, threatened in writing, against CDG in respect of any Taxes, except as set forth in Schedule 6.10(d).
- (e) There are no Encumbrances for material Taxes on any of the assets of CDG that arose in connection with any failure to pay any Tax or otherwise, except for Encumbrances described on Schedule 6.10(e) other than Encumbrances for current Taxes which are not yet due and payable.
- (f) CDG is not a party to, bound by or liable under any Tax-sharing agreement.
- (g) For purposes of this Agreement, "TAX RETURNS" includes, without limitation, all returns, reports, declarations, elections, notices, filings, information returns and statements filed in respect of Taxes.

- (h) For purposes of this Agreement, "TAXES" includes, without limitation, all taxes, duties, fees, premiums, assessments, imposts, levies and other charges of any kind whatsoever imposed by any Governmental Authority, together with all interest, penalties, fines, additions to tax or other additional amounts imposed in respect thereof, including those levied on, or measured by, or referred to as income, gross receipts, profits, capital, transfer, land transfer, sales, goods and services, use, value-added, excise, stamp, withholding, business, franchising, property, payroll, employment, health, social services, education and social security taxes, all surtaxes, all customs duties and import and export taxes, all license, franchise and registration fees and all unemployment insurance, health insurance and other government pension plan premiums.
- (i) For purposes of this Agreement, "GOVERNMENTAL AUTHORITY" means any government, regulatory authority, governmental department, agency, commission, board, tribunal, crown corporation, or court or other law, rule or regulation-making entity having or purporting to have jurisdiction on behalf of any nation, or province or state or other subdivision thereof or any municipality, district or other subdivision thereof.
- 6.11 FINDERS; BROKERS. With the exception of fees and expenses payable to Bear, Stearns & Co. Inc., which shall be Sellers' sole responsibility, none of CDG and the Sellers is a party to any agreement with any finder or broker, or in any way obligated to any finder or broker for any commissions, fees or expenses in connection with the origin, negotiation, execution or performance of this Agreement or the Ancillary Agreements.
- 6.12 LITIGATION, CLAIMS AND PROCEEDINGS. Except as set forth in Schedule 6.12 attached hereto, as of the date hereof there are no (i) judgments, orders, writs or injunctions of any foreign or U.S. federal, state or local court or governmental authority presently pending or, to Sellers' Knowledge, threatened against CDG, the Business or the Shares or by which any of the Assets or the Shares are or would be bound or (ii) lawsuits, actions, arbitrations, claims, governmental proceedings or notices of violation presently pending or, to Sellers' Knowledge, threatened to which CDG is a party (as plaintiff, defendant or otherwise) which relate to CDG, the Shares or the Business, except in the case of each of clauses (i) and (ii) for Liabilities which are not set forth on Schedule 4.1(a). As of the date hereof, there is no lawsuit, action, arbitration or governmental proceeding pending to which any of CDG and its affiliates is a party which seeks to prevent the consummation, or otherwise challenges the validity, of the transactions contemplated hereby.
- 6.13 EMPLOYEE MATTERS. (a) CDG is not a party to or bound by any labor contract and neither Seller is a party to or bound by any labor contract affecting CDG or the CDG Employees. The CDG Employees are not represented by any union or other employee organization. There has not been, there is not presently pending or existing and, to Seller's Knowledge, there is not threatened (a) any strike, slowdown, picketing, work stoppage or lockout in respect of CDG or any of the CDG Employees or (b) any organizational activity involving

CDG or any of the CDG Employees. To Sellers' Knowledge, CDG is not engaged in any unfair labor practice or discriminatory employment practice and as of the date hereof no complaint of any such practice against CDG has been filed and is pending or is threatened to be filed with or by the National Labor Relations Board, the Equal Employment Opportunity Commission or any other administrative agency that regulates labor or employment practices. As of the date hereof no grievance has been filed or, to Sellers' Knowledge, threatened to be filed against CDG by any employee pursuant to any collective bargaining or other employment agreement.

- (b) Except as set forth on Schedule 4.1(a), CDG is not party to any employment or severance arrangement or agreement that is to be assumed by Buyer. The execution of this Agreement and the consummation of the transactions contemplated hereby will not (x) entitle any current or former employee of CDG to any severance, termination or other payment pursuant to any employment or severance policies or arrangements or (y) accelerate the time of payment or vesting or increase the amount of compensation due any such employee that, in either case, is to be assumed by Buyer.
- (c) To Sellers' Knowledge, all employees of CDG are either United States citizens or resident aliens specifically authorized to engage in employment in the United States.
- (d) Other than as set forth on Schedule 6.13(d), within the six-year period preceding the date hereof CDG has not maintained or had an obligation to contribute to any "employee benefit plans," as defined in Section 3(3) of ERISA or any other employee benefit arrangements or payroll practices.
- 6.14 SURVIVAL OF REPRESENTATIONS AND WARRANTIES. (a) None of the representations and warranties made by Sellers shall survive the Closing, except as provided in this Section 6.14. The representations and warranties contained in the first and third sentences of Section 6.1 and in Section 6.2, Section 6.3, Section 6.4 and Section 6.5 shall survive the Closing until the first anniversary of the Closing Date and shall thereupon expire together with any right to indemnification for breach thereof (except to the extent a claim for indemnification has been made prior to the expiration of such period.)
- (b) EXCEPT AS SET FORTH IN THIS AGREEMENT AND IN OTHER CERTIFICATES DELIVERED PURSUANT TO THIS AGREEMENT, NONE OF SELLERS AND THEIR AFFILIATES MAKES ANY REPRESENTATION OR WARRANTY TO BUYER WHATSOEVER, EXPRESS OR IMPLIED, WITH RESPECT TO CDG, THE SHARES OR THE BUSINESS.

ARTICLE 7.

REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer hereby represents and warrants to Sellers that:

7.1 CORPORATE EXISTENCE. Buyer is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation. Buyer has all requisite power and authority to own the Shares and to carry on the Business as it intends to conduct the Business after the Closing. Buyer is duly authorized, qualified or licensed to do business as a foreign corporation and in good standing in every jurisdiction wherein, by reason of the character of the Business, the failure to be so qualified would have a material adverse effect on the ability of Buyer to consummate the transactions contemplated hereby. Buyer has full corporate power and authority to enter into and perform this Agreement and each Ancillary Agreement to which it is a party.

7.2 DUE AUTHORIZATION; NO BREACH. (a) The execution and performance by Buyer of this Agreement and each Ancillary Agreement to which it is a party and the transactions contemplated hereby and thereby have been approved all necessary corporate action, and no further corporate action is required to be taken by it in order to execute, deliver and perform this Agreement. Each of this Agreement and the Ancillary Agreements to which it is a party is a valid and legally binding obligation of Buyer, and each agreement or instrument contemplated by this Agreement, when executed and delivered by Buyer in accordance with the provisions hereof, will be a valid and legally binding obligation of Buyer in each case enforceable against Buyer in accordance with its terms subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to the effect of general principles of equity (regardless of whether enforcement is considered in a proceeding at law or in equity). All persons who have executed this Agreement on behalf of Buyer or who will execute on behalf of Buyer any agreement or instrument contemplated by this Agreement, have been duly authorized to do so by all necessary corporate action. Except as set forth on Schedule 7.2 hereto, neither the execution and delivery of this Agreement, the Ancillary Agreements and all other agreements and documents to be executed or delivered hereunder, nor the consummation of the transactions contemplated hereby and thereby, will (i) violate, or conflict with, any provision of Buyer's certificate of incorporation or by-laws or other governing documents, (ii) violate, or conflict with, or result in a breach of any provisions of, or constitute a default (or an event of default which with notice or lapse of time or both would because a default) under, or result in the termination (or grant a right of termination) of, cancellation, amendment or accelerate (or grant the right to accelerate) the performance required by, or result in the creation of any lien, security interest, charge or encumbrance upon any of the properties or assets of Buyer under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, agreement, lease, Permit or other material instrument to which Buyer is a party or by which it or any of its properties or assets are bound or (iii) violate, or conflict with, any order, writ, injunction, arbitration award, judgment or decree of any court, governmental body or arbitrator applicable to Buyer or any applicable statute, law, rule or regulation, except, in the case of clause (ii) or (iii), as would not have a material adverse effect

on the ability of Buyer to perform its obligations hereunder or any Ancillary Agreement to which it is a party.

- 7.3 CONSENTS. Buyer is not subject to any order, judgement or decree which would prevent the consummation of the transactions contemplated hereby. No claim, legal action, suit, arbitration, governmental investigation, action, or other legal or administrative proceeding is pending or, to the knowledge of Buyer, threatened against Buyer which would enjoin or delay the transactions contemplated hereby. Except as set forth in Schedule 7.3 hereto, no consent, approval, order or authorization of, license or permit from, notice to or registration, declaration or filing with, any governmental authority or entity, domestic or foreign, or of any third party, is or has been required on the part of Buyer in connection with the execution and delivery of this Agreement or the Ancillary Agreements or any of the transaction documents, or the consummation of the transactions contemplated hereby and thereby, except for such consents, approvals, orders or authorizations of, licenses or permits, filings or notices the failure of which to obtain or make would not have a material adverse effect on the ability of Buyer to consummate the transactions contemplated hereby.
- 7.4 PURCHASE FOR INVESTMENT. Buyer is aware that the Shares have not been registered under the Securities Act of 1933, as amended (the "SECURITIES ACT"), or under any state or foreign securities laws. Buyer is not an underwriter, as such term is defined under the Securities Act, and is purchasing such shares solely for investment, with no present intention to distribute any Shares to any person, and Buyer will not sell or otherwise dispose of Shares except in compliance with the registration requirements or exemption provisions under the Securities Act and the rules and regulations promulgated thereunder, or any other applicable securities laws.
- 7.5 FINDERS; BROKERS. Buyer is not party to any agreement with any finder or broker, or in any way obligated to any finder or broker for any commissions, fees or expenses in connection with the origin, negotiation, execution or performance of this Agreement or the Ancillary Agreements.
- 7.6 AVAILABILITY OF FUNDS. Buyer has (i) in hand or (ii) the right to borrow under legally valid and binding definitive agreements all funds necessary to enable Buyer to consummate the transactions contemplated by this Agreement.
- 7.7 NO OTHER REPRESENTATIONS OR WARRANTIES. Except for the representations and warranties contained in this Article 7, neither Buyer nor any other person makes any other express or implied representation or warranty on behalf of Buyer.
- 7.8 SURVIVAL OF REPRESENTATIONS AND WARRANTIES. None of the representations and warranties contained in this Article 7 shall survive the Closing, except for those representations and warranties contained in Sections 7.2, 7.4 and 7.5 which shall survive until the 12-month anniversary of the Closing.

ARTICLE 8.

COVENANTS

- 8.1 CONDUCT OF THE BUSINESS. Sellers hereby agree that, except (i) as expressly contemplated elsewhere in this Agreement or (ii) as consented to by Buyer, from the date hereof until the Closing or the termination of this Agreement in accordance with Article 10 hereof, Sellers shall:
 - (a) cause CDG to operate the Business only in the usual and ordinary course of business and use all reasonable efforts consistent with good business judgment to (i) maintain the present business organization of CDG, (ii) maintain CDG's books and records in accordance with past practices; (iii) keep available the services of CDG's officers and employees related to or involved with the Business, provided, however, that Buyer acknowledges that the announcement of the transactions contemplated by this Agreement may adversely affect CDG's ability to keep available such services; and (iv) maintain satisfactory relationships with licensors, suppliers, creditors, distributors, customers and others transacting business with CDG relating to or involving or in connection with the Business, provided, however, that Buyer acknowledges that the announcement of the transactions contemplated by this Agreement may adversely affect CDG's ability to maintain such relationships;
 - (b) refrain from taking any action which reasonably could be expected to render any representation or warranty of Sellers contained herein untrue or incorrect in any material respect (except to the extent a representation or warranty is qualified by materiality, in which case Sellers will refrain from taking any action which would render such representation or warranty untrue or incorrect) as of the Closing;
 - (c) cause CDG to comply in all material respects with all laws applicable to the Business;
 - (d) cause CDG to comply in all material respects with all applicable laws;
 - (e) cause CDG to refrain from making any wage or salary increase, or from agreeing to make any such increase, for any of the CDG Employees except as required pursuant to any employment agreement listed on Schedule 4.1(a), as the same is in effect on the date hereof, and except for any such increases that, in the aggregate, do not exceed \$50,000 per year and, with respect to any CDG Employee, individually, do not exceed 15% of such CDG Employee's current base salary;

- (f) cause CDG to refrain from terminating or attempting to terminate any Contract, including the KCPL Shirt License Agreement, other than the termination or attempted termination in the ordinary course of business of a Contract other than the KCPL Shirt License Agreement;
- (g) refrain from amending the Certificate of Incorporation or By-Laws of CDG, merging CDG with or into or consolidating CDG with any other person, subdividing or in any way reclassifying any of CDG's capital stock or changing in any manner the rights of the Shares or the character of the Business;
- (h) cause CDG to refrain from issuing, selling, or purchasing, or issuing options or rights to subscribe for, or entering any contracts or commitments to issue, sell or purchase any shares of CDG's capital stock, the capital stock of others or any bonds, notes, debentures or other evidences of indebtedness of CDG or any other person;
- (i) cause CDG to refrain from declaring or making any dividend payment or distribution to any stockholder of CDG or disposing of any Asset (other than KCPL Inventory in the ordinary course of business) or making any direct or indirect redemption, retirement, purchase or other acquisition of any capital stock of CDG; and
- (j) cause CDG to refrain from making any change in its accounting methods or practices, except as required by law or United States generally accepted accounting practices.
- 8.2 BEST EFFORTS AND ALTERNATIVE PROPOSALS. Each party to this Agreement will use all reasonable best efforts to take all action and to do all things necessary, proper or advisable in order to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements (without payment of money, commencement of litigation, the assumption of any material obligation or the entering of any agreement to divest or hold separate any assets), including using all reasonable best efforts to prevent a change in the current management of CDG referred to in the next sentence. If the parties anticipate a change in the current management of CDG prior to Closing, or if a change of management occurs prior to Closing, in each case within the meaning of Section 16.1(iv) of the KCPL Shirt License Agreement, CDG shall present the names and business qualifications of the new management employees, along with a summary of such individuals' experience and reputation in the garment industry, to KCPL for its review and reasonable approval and take all other reasonable steps to secure KCPL's reasonable approval of such new management employees prior to Closing. All assets of CDG other than the KCPL Inventory and assets set forth on Schedule 8.2 shall be assigned or transferred by CDG to a Seller or one of their affiliates as soon as practicable after the date hereof. Any Liability of CDG which the Buyer Indemnified Parties are indemnified against pursuant to Section 4.1(a) shall be assumed by a Seller or one of their affiliates. Sellers and CDG shall use their reasonable best efforts to effect such assignments, transfers and assumptions prior

to Closing. Sellers agree that they and their affiliates shall (a) not, and shall direct and use reasonable best efforts to cause their officers, directors, employees, agents and representatives (including, without limitation, any investment banker, attorney or accountant retained by it or any of its subsidiaries) not to, initiate, solicit or encourage, directly or indirectly, any inquiries or the making or implementation of any proposal or offer to consummate any transaction similar to the transactions contemplated by this Agreement and the Ancillary Agreements involving the Shares, the KCPL Shirt License Agreement or the Business that would prevent, delay or impede the consummation of the transactions contemplated hereby (any such proposal or offer being hereinafter referred to as an "ALTERNATIVE PROPOSAL") or engage in any negotiations concerning, or provide any confidential information or data to, or have any discussions or otherwise cooperate in any way with, any person relating to an Alternative Proposal, or otherwise facilitate or encourage any effort or attempt to make or implement an Alternative Proposal and (b) upon execution and delivery of this Agreement, (and shall direct and use their best efforts to cause their officers, directors, employees, agents and representatives to) immediately cease and cause to be terminated any existing activities, discussions or negotiations with any parties conducted heretofore with respect to any of the foregoing.

- 8.3 NOTIFICATION. Sellers shall provide Buyer with written notice prior to Closing of any material damage or destruction of the assets used in the Business, material adverse change in the Business, any occurrence that would cause the representation set forth in Section 6.12 hereof to no longer be true without regard to the date as of which such representation was made, or breaches of Article 6 that constitute a failure to satisfy Section 9.3(b), in each case of which Sellers or CDG acquire actual knowledge prior to the Closing Date.
- 8.4 LETTER OF CREDIT. Buyer shall use all reasonable efforts to replace at Closing all outstanding letters of credit pertaining to the orders which CDG has with third party manufacturers for the manufacture of goods for use in the Business ("SOURCING COMMITMENTS") to which CDG is a party and to replace at Closing the outstanding letter of credit provided by CDG for the benefit of KCPL under the KCPL Shirt License Agreement (the "KCPL LETTER OF CREDIT"). If Buyer is unable to replace at Closing any such outstanding letters of credit, in whole or in part, Buyer (i) with respect to the KCPL Letter of Credit, shall and (ii) with respect to all such other letters of credit will use all reasonable efforts to, provide at Closing "back-to-back" letters of credit to each of the banks that issued each of the letters of credit that are not being replaced (each, an "ISSUING BANK") in a form satisfactory to the relevant Issuing Bank (in its sole discretion) issued by a bank satisfactory to the relevant Issuing Bank (in its sole discretion), in an amount equal to 100% of the then undrawn stated amount of the outstanding letters of credit issued by such Issuing Bank that are not being replaced. The notice of reimbursement with respect to any drawdown on any such "back-to-back" letter of credit shall state the amount of the drawdown and the purchase orders to which it relates. If Buyer is unable to provide such "back- to-back" letters of credit, Buyer will reimburse CDG for all amounts drawn down under outstanding letters of credit that are not being replaced and for which "back-to-back" letters of credit are not being provided within five business days after receipt by Buyer of reasonable

evidence of such drawdown. Such evidence shall include the letter of credit number, the latest ship date, product description, product cost, price per unit and a copy of the funding message and such other documentation that Buyer may reasonably request. On the date hereof and at least ten business days prior to Closing CDG will provide Buyer with a list of each outstanding letter of credit, stating the letter of credit number, the issuing bank, the name, address and country of the beneficiary, the purchase orders covered, the expiration date and the amount outstanding on the day the list is compiled with respect to each outstanding letter of credit. If Buyer offers to provide a "back-to-back" letter of credit to an Issuing Bank issued by Chase Manhattan Bank, N.A., Citibank, N.A. or The Bank of New York in customary form in an amount equal to 100% of the then undrawn stated amount of the relevant outstanding letter of credit, and the Issuing Bank states that any such issuing bank or such form is not satisfactory to the Issuing Bank, Buyer shall be deemed to have satisfied its obligations under the second sentence of this Section 8.4 with respect to such outstanding letter of credit.

- 8.5 RETURNED INVENTORY. Buyer and Sellers agree (a) to minimize all returns of KCPL Inventory sold by CDG to third parties prior to Closing and any such return needs to be authorized by each party and (b) if such returns cannot be avoided as a practical matter consistent with the past practice of the Business, Buyer agrees to purchase all such inventory which is returned to Sellers within 180 days after Closing (the "RETURNED INVENTORY"). Buyer shall pay Seller for such Returned Inventory an amount equal to seventy-five percent (75%) of Buyer's actual selling price therefor.
- 8.6 ACCOUNTS RECEIVABLE. (a) After the Closing Date, Sellers shall have the sole right and authority to collect, for their own account and sole benefit, all monies receivable in respect of the Assets for any period prior to the Closing Date, and Buyer shall have the sole right and authority to collect, for its own account and sole benefit, all monies payable in respect of the Assets for any period on or following the Closing Date. Buyer and Sellers shall jointly send a letter on the Closing Date to each customer of the Business announcing the transaction contemplated hereby and describing the customer's obligations to each of Sellers and Buyer, including specific instructions for the remittance of payments to each of Sellers and Buyer. The form and substance of such letter shall be agreed upon at or prior to the Closing.
- (b) If Sellers (or their successors) shall receive any such monies with respect to any period on or following the Closing Date, it shall hold all such monies in trust for the sole benefit of Buyer and its affiliates. If Buyer or any of its affiliates (or their successors) shall receive any such monies with respect to any period prior to the Closing Date, it shall hold all such monies in trust for the sole benefit of Sellers. On the first and fifteenth day of each calendar month (or, if such day is not a business day, the first business day thereafter), Sellers (or their successors) or Buyer and its affiliates (or their successors), as the case may be, shall cause the transfer and delivery to the other of any such monies which they (or their successors) may receive after the Closing Date. No party will take any action that compromises the ability of the other party and its affiliates or their successors to collect, for its own account and sole benefit, all

monies payable to it as provided herein. Each of Sellers and Buyer and their respective affiliates will use all reasonable efforts to collect its own receivables and to assist the other in the collection of their respective accounts receivable; provided, however, that nothing in this Agreement, shall be deemed to require any of them (i) to take any action that such party believes, in its sole discretion, would be detrimental to such party's business relationship with any customer or (ii) to incur any out-of-pocket expenses or allocate any significant employee resources to the collection of the other's accounts receivable.

- (c) All disputed items and deductions taken from payments for invoices, arising from, but not limited to, markdown and other allowances, returns, NSF checks, vendor guide violations, shortages, discounts and price relating to pre-Closing accounts receivable and shipments are the responsibility of Sellers, and Sellers shall have sole authority to make decisions regarding the same. All disputed items and deductions taken from payments for invoices, arising from, but not limited to, markdown and other allowances, returns, NSF checks, vendor guide violations, shortages, discounts and price relating to post-Closing accounts receivable and shipments are the responsibility of Buyer, and Buyer shall have sole authority to make decisions regarding the same. Each of Sellers and Buyer will identify to the other all such disputes and deductions that come to its attention but are the responsibility of the other and provide any documentation in its possession to the responsible entity. In the event that a deduction has been taken against the wrong party, such party shall provide all documentation provided by the customer to the responsible party. Settlement in the form of a cash reimbursement of deductions taken against the wrong party shall be made on the first and fifteenth day of each calendar month (or, if such day is not a business day, the first business day thereafter).
- 8.7 NAME CHANGE. Buyer acknowledges and agrees that neither it nor any of its affiliates shall by virtue of any of the transactions contemplated hereby, or otherwise, obtain any right, title or interest whatsoever in or to the name "Cluett" and any trademark, tradename, symbol, sign or logo bearing the name "Cluett" or suggestive thereof. After the Closing neither Buyer nor any of its affiliates (including CDG) shall utilize in any respect the name "Cluett" or any trademark, tradename, sign, symbol or logotype bearing the name "Cluett" or suggestive thereof, except that CDG may continue to contain the name "Cluett in its name for a period not exceeding five business days from the Closing Date.

ARTICLE 9.

CONDITIONS OF CLOSING

9.1 MUTUAL CONDITIONS. The obligations of each party to this Agreement to consummate the transactions contemplated by this Agreement and the Ancillary Agreements are, unless waived by each party, subject to the fulfillment on or before the Closing, of each of the following conditions:

- (a) No statute, rule, regulation, injunction or restraining order shall be in effect to forbid or enjoin the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements;
- (b) The transactions contemplated by this Agreement and the Ancillary Agreements to be completed before the Closing shall have been consummated upon the terms and subject to the conditions set forth therein;
- (c) All waiting periods under the HSR Act shall have expired or been terminated; and
- (d) The transactions contemplated by the Purchase Agreement shall have been consummated upon the terms and subject to the conditions set forth therein.
- 9.2 ADDITIONAL CONDITIONS OF SELLERS. The obligations of Sellers to consummate the transactions contemplated by this Agreement and the Ancillary Agreements are, unless waived by Buyer, subject to the fulfillment, on or before the Closing of each of the following additional conditions:
 - (a) Buyer and each other party thereto other than CDG and its affiliates shall have executed and delivered each of the Ancillary Agreements to which it is a party; and
 - (b) All covenants of Buyer under this Agreement and the Ancillary Agreements to be performed prior to the Closing shall have been performed in all material respects, and the representations and warranties of Buyer contained in this Agreement and the Ancillary Agreements shall be true and correct on and as of the Closing in all material respects (except to the extent qualified by materiality, in which event such representations and warranties shall be true and correct) with the same effect as though such representations and warranties had been made on and as of such date.
- 9.3 ADDITIONAL CONDITIONS OF BUYER. The obligations of Buyer to consummate the transactions contemplated by this Agreement and the Ancillary Agreements are, unless waived by Buyer, subject to the fulfillment, on or before the Closing, of each of the following additional conditions:
 - (a) Sellers and each other party thereto other than Buyer shall have executed and delivered each of the Ancillary Agreements to which they are a party;
 - (b) All covenants of Sellers under this Agreement and the Ancillary Agreements to be performed prior to the Closing shall have been performed, and the representations and warranties of Sellers contained in this Agreement and the Ancillary Agreements shall be true and correct on and as of the Closing in all material respects

(except to the extent qualified by materiality in which event such representations and warranties shall be true and correct) with the same effect as though such representations and warranties had been made on and as of such date;

- (c) Sellers shall have delivered to Buyer the written resignations of all directors and officers of CDG, in each case effective as of the Closing Date; and
- (d) There shall not have occurred since the date hereof any damage to, or destruction or loss of, KCPL Inventory that constitutes a material adverse change to the KCPL Inventory.

ARTICLE 10.

TERMINATION; SURVIVAL

- 10.1 TERMINATION BY BUYER OR CDG. Anything herein or elsewhere to the contrary notwithstanding, this Agreement shall be terminated and the transactions hereby contemplated shall be abandoned if prior to Closing the transactions contemplated by the CDG Asset Purchase Agreement (as defined in the Purchase Agreement) shall have been consummated and this Agreement may be terminated and the transactions contemplated hereby abandoned at any time prior to Closing by:
 - (a) mutual written consent of Sellers and Buyer;
- (b) Sellers or Buyer, by written notice to the other, if the transactions contemplated hereby are not consummated on or before July 31, 2000 and if the failure to consummate such transactions on or before such date did not result from a breach prior to or on the Closing Date of any representation, warranty or covenant of the party seeking such termination; provided that if any of the conditions set forth in Sections 9.1(c) and 9.1(d) have not been satisfied prior to July 31, 2000, Sellers may in their reasonable discretion elect to extend the July 31, 2000 deadline referred to above to October 31, 2000; and provided further that Buyer may not terminate this Agreement and abandon the transactions contemplated hereby because the condition set forth in Section 9.3(d) has not been satisfied (i) until the twelve month anniversary of the material adverse change referred to in Section 9.3(d) and (ii) unless CDG and its affiliates have not cured such material adverse change on or prior to such twelve month anniversary:
- (c) Sellers or Buyer, by written notice to the other, if a court of competent jurisdiction or governmental, regulatory or administrative agency or commission shall have issued an order, decree or ruling or taken any other action, in each case, permanently restraining, enjoining or otherwise prohibiting the consummation of the transactions

contemplated hereby and such order, decree, ruling or other action shall have become final and nonappealable;

- (d) Sellers or Buyer, by written notice to the other, if the other party(ies) is in breach in any material respect of any of its representations or warranties made in this Agreement or the Ancillary Agreements, or is in violation or default in any material respect of any of its covenants or agreements in this Agreement or the Ancillary Agreements, if such breach, violation or default is not cured within twenty (20) business days after written notice by the notifying party to the other party(ies);
- (e) Sellers or Buyer, by written notice to the other, if the Purchase Agreement, is terminated in accordance with its terms;
- (f) Sellers, by written notice to Buyer, if the \$7 million fee referred to in Section 8.11 of the Purchase Agreement (i) becomes payable and is not irrevocably waived in writing by Buyer or (ii) is paid; or
- (g) Sellers or Buyer, by written notice to the other on or before July 15, 2000, if all consents and agreements required to be obtained from KCPL and its affiliates under the KCPL Shirt License Agreement, the Assignment of License Agreement referred to in the Purchase Agreement, the Termination of License Agreement referred to in the Purchase Agreement, or otherwise, in connection with the execution and delivery of the CDG Asset Purchase Agreement and the consummation of the transactions contemplated thereby have been obtained irrevocably in writing.
- 10.2 SURVIVAL. If this Agreement is terminated pursuant to Section 10.1 hereof, this Agreement shall become void and of no further force and effect, except for the provisions of Article 4 and Sections 12.1(a), 13.1, 13.2 and 15.8 hereof; PROVIDED that such termination shall not relieve any party for liability for Damages resulting from its breach of this Agreement. Promptly following termination of this Agreement each party will destroy or return to the other parties all documents received from such parties in connection with the contemplated transaction, except documents which have been publicly distributed or publicly filed.

ARTICLE 11.

CLOSING

11.1 CLOSING. The closing of the transactions contemplated by this Agreement (the "CLOSING") shall take place at the offices of Simpson Thacher & Bartlett, 425 Lexington Avenue, New York, New York as soon as practicable after all the conditions to Closing set forth

in Article 9 hereof shall be satisfied or duly waived, or at such other time and place as Buyer and Sellers may mutually agree (the date of the Closing referred to herein as the "CLOSING DATE").

- 11.2 SELLERS' OBLIGATIONS AND CLOSING DELIVERIES. At the Closing, Sellers shall deliver, or cause to be delivered, to Buyer:
 - (a) each of the Ancillary Agreements other than the Hosiery Agreement, executed by Sellers or CDG, as the case may be and, in the case of the Waivers and Consents, CDG and, to the extent relevant, its affiliates (including the Sellers) and its contractors, subcontractors and warehousemen that agree to sign after CDG and the Sellers use their reasonable efforts to cause them to sign;
 - (b) written receipt executed by each of the Sellers of payment of the portion of the Purchase Price paid at Closing;
 - (c) certificate of the President or any Vice President of CDG that the conditions set forth in Section 9.3(b) have been satisfied;
 - (d) certificates evidencing the Shares, which certificates shall be properly endorsed for transfer or accompanied by duly executed stock powers, in either case executed in blank or in favor of Buyer, with all necessary stock transfer stamps attached thereto and cancelled, and otherwise in a form acceptable for transfer on the books of CDG;
 - (e) resignations of directors of CDG as contemplated by Section 9.3(c) hereof; and
 - (f) such instruments of sale, transfer, conveyance, assignment and assumption, in form and substance reasonably satisfactory to Buyer's counsel, as shall be effective to effect the assignments, transfers and assumptions contemplated by Section 8.2.
- 11.3 BUYER'S OBLIGATIONS AND CLOSING DELIVERIES. At the Closing, Buyer shall deliver, or cause to be delivered:
 - (a) to Sellers, the Purchase Price (except as expressly provided otherwise in Section 3.2);
 - (b) to Sellers, the Ancillary Agreement referred to in Sections 2.2 (a) and 2.2 (d), executed by Buyer;
 - (c) to the Escrow Agent, the KCPL Inventory Escrow;

(d) to Sellers, a Certificate of the President or any Vice President of Buyer to the effect that the conditions set forth in Section 9.2(b) have been satisfied.

ARTICLE 12.

EXPENSES AND POST-CLOSING OBLIGATIONS

- 12.1 TAXES AND OTHER CHARGES. (a) Except as specifically provided for otherwise in Sections 3.2(c) and (d) and Section 12.1(d), and regardless of whether or not the transactions contemplated hereby are consummated, each party to this Agreement shall pay all expenses incurred by it or on its behalf in connection with the preparation, authorization, execution and performance of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby, including, but not limited to, all fees and expenses of all consultants, brokers, investment bankers, agents, representatives, counsel and accountants engaged by such party. Any liability for all sales, documentary stamp, transfer and similar Taxes (including, without limitation, all recording fees and related charges) arising from or attributable or related to the sale, transfer or assignment to Buyer of any of the Shares shall be paid by the party that is primarily responsible for such Taxes under applicable law and such party shall, at its expense, file all Tax Returns and other documentation with respect to all such Taxes.
- (b) Notwithstanding anything else in this Agreement to the contrary, Sellers will indemnify and hold Buyer harmless from and against any and all Liabilities for Taxes imposed on CDG (i) for taxable periods ending on or before the Closing Date and (ii) in the case of Taxable periods beginning before and ending after the Closing Date, that are attributable to that portion of the Taxable period beginning before the Closing Date through the Closing Date. For purposes of clause (ii) of this Section 12.1(b), the Taxes based on income or gross receipts attributable to such pre-closing Tax period shall be determined by closing the books of CDG as of the close of the Closing Date, and treating the portion of the Taxable period through and including the Closing Date and after the Closing Date as if each were separate Taxable periods, except that exemptions, allowances and deductions calculated on an annual or similar basis shall be apportioned on a per diem basis.
- (c) CDG shall, as instructed by Sellers, prepare and file, or cause to be prepared and filed, any and all Tax Returns required to be filed by CDG (after giving effect to any valid extensions of the due date for filing any such Tax Returns) in respect of periods, or portions thereof, ending on or prior to the Closing Date. All such Tax Returns shall be prepared in a manner consistent with the prior Tax Returns of CDG, unless otherwise required under applicable law. CDG, as instructed by Buyer, shall prepare and file, or cause to be prepared and filed any and all other Tax Returns required to be filed by CDG. Without prejudice to its right to indemnification under this Agreement, Buyer shall cause CDG to pay all Taxes shown as due and

owing on all such Tax Returns. Sellers and Buyer shall reasonably cooperate, and shall cause their respective affiliates, officers, employees, agents, auditors and other representatives reasonably to cooperate, in preparing and filing all Tax Returns, including maintaining and making available to the other parties all records necessary in connection with Taxes and in resolving all disputes and audits with respect to all taxable periods relating to Taxes.

- (d) If Sellers and Buyer disagree as to the amount for which Buyer and CDG or Sellers are liable in respect of Taxes under this Agreement, CP and Buyer shall promptly consult with each other in an effort to resolve such dispute. If any such point of disagreement cannot be resolved within 30 days of the date of consultation, such dispute shall be submitted to a Tax adviser which is mutually acceptable to Buyer and each of the Sellers to act as an arbitrator to resolve all points of disagreement concerning Taxes with respect to this Agreement; provided that Buyer and Seller shall share equally the cost of such Tax advisor.
- (e) Sellers, at their option, shall have sole control over any contest or other proceeding (including claims for refund or challenges of Tax assessments) which relates to liability for Taxes for periods or portions thereof ending on or prior to the Closing Date. In any event, any refunds of Taxes which become available on or after the date of the Closing but which relate to Tax periods or portions thereof ending on or prior to the Closing shall belong to Sellers.
- 12.2 FURTHER ASSURANCES. At any time after the Closing, Sellers and Buyer agree to cooperate with one another to execute and deliver such other documents, instruments of transfer or assignment, files, books and records and do all such further acts and things as may be reasonably required to carry out the transactions contemplated by this Agreement or the Ancillary Agreement.
- 12.3 ACCESS TO BOOKS, RECORDS AND FACILITIES. (a) Sellers agree that on and after the Closing they will permit Buyer and its representatives, during normal business hours and working days and upon reasonable advance notice, (i) to have access to and to examine and make copies of all books and records of CDG (except books and records protected by attorney-client or other privilege which Sellers may be entitled to assert against Buyer or its affiliates in any pending or threatened proceeding, suit or action) which relate to the Shares or the operations of the Business to the extent that the events reflected therein relate to transactions or events occurring prior to the Closing or to transactions or events occurring subsequent to the Closing which arise out of transactions or events occurring prior to the Closing and (ii) to have access to and to examine and make copies of all documents listed in the Schedules attached hereto and all files, records and papers of any and all proceedings and matters listed in the Schedules attached hereto. During the period commencing on the date hereof and ending on the Closing Date, CDG will, and Sellers will cause CDG to, afford to Buyer and its counsel, accountants and other authorized representatives access at all reasonable times upon reasonable advance notice to the officers, directors, employees, accountants and other advisors and agents, properties, books, records and contracts, of CDG (excluding those relating to CDG's businesses other than the

Business, CDG's assets other than the Assets, or Liabilities of CDG not set forth on Schedule 4.1(a)), and the right to make copies and extracts from such books, records and contracts, and to furnish Buyer with all financial, operating and other data and information concerning CDG and the Business (excluding those relating to CDG's businesses other than the Business, CDG's assets other than the Assets, or Liabilities of CDG not set forth on Schedule 4.1(a)) as Buyer and its advisors may reasonably request. All books and records of CDG relating to the Shares, the Assets or the operations of the Business will be preserved by CDG in accordance with CDG's records retention policy, but in no event for a period of less than three years following the Closing. Prior to any destruction or disposition by CDG of any such books and records, Sellers will notify Buyer in writing and Buyer shall have the right to receive and retain such books and records at its expense.

- (b) Except in respect of Liability for Taxes, in which event the commitment shall continue until the expiration of Sellers' obligations therefor, Buyer agrees that from the Closing Date and until the fifth anniversary of the Closing Date, Buyer will permit, and will cause its subsidiaries to permit, Sellers and their representatives full access during normal business hours and working days and upon reasonable advance notice to have access to the books and records of Buyer and its subsidiaries (except records protected by attorney-client or other privilege which Buyer or its affiliates may be entitled to assert against Sellers or their affiliates in any pending or threatened proceeding, action or suit), relating to the Shares, the Assets or the operations of the Business, to the extent that any of the foregoing relates to periods prior to the Closing, and is reasonably necessary in connection with any then pending or threatened litigation, claim, liability, or judicial or administrative matters in which any of Sellers or their affiliates is involved.
- 12.4 RECISSION. If after the Closing the \$7 million fee referred to in Section 8.11 of the Purchase Agreement (i) becomes payable and is not irrevocably waived in writing by Buyer or (ii) is paid, the parties hereto shall use their best efforts to (1) unwind and rescind all transactions consummated pursuant hereto, (2) restore the status quo as at immediately prior to the Closing and (3) terminate this Agreement to the extent provisions hereof have not been performed and do not relate to transactions consummated pursuant hereto that are not unwound and rescinded; provided that, in any event, the provisions of Article 4 and Section 12.1(a) (in each case, as modified to reflect the actions referred to in clauses (1) and (2)) and Sections 13.1, 13.2 and 15.8 shall remain in effect, and provided further that such termination shall not relieve any party for Damages resulting from its breach of this Agreement. To the extent that the parties hereto cannot effect any of the actions referred to in clauses (1) and (2) of the preceding sentence, they shall use their best efforts to enter into any reasonable and lawful arrangement designed to put the parties in the same or comparable economic position that they would have been in had such actions been effected.

ARTICLE 13.

PUBLICITY, CONFIDENTIALITY

- 13.1 PUBLICITY. Sellers and Buyer agree that no publicity, release or announcement concerning (a) the execution of this Agreement or the Ancillary Agreements, (b) any of the provisions of this Agreement or the Ancillary Agreements or the transactions contemplated hereby or thereby or (c) the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements shall be issued without the advance written approval of the form and content of the same by the other parties; provided, however, that no such consent shall be required when such disclosure is required by applicable law, although the parties agree to consult with each other as to the content of any release so required and consider in good faith the comments of the other
- 13.2 CONFIDENTIALITY. Buyer and Sellers confirm that they and their affiliates are bound by the terms of the confidentiality agreement, dated December 9, 1998, as amended (the "CONFIDENTIALITY AGREEMENT"), between Cluett American Investment Corp. and Phillips-Van Heusen Corporation and that they will keep and treat the evaluation material and all other items of confidential information exchanged hereunder by Buyer and its affiliates, on the one hand, and Sellers and their affiliates, on the other hand, in accordance with the terms of the Confidentiality Agreement.

ARTICLE 14.

NOTICES

14.1 NOTICES. Any notices or communications permitted or required hereunder shall be deemed sufficiently given if hand-delivered, or sent by (i) registered or certified mail return receipt requested, (ii) telecopy or other electronic transmission service (to the extent receipt is confirmed) or (iii) by overnight courier, in each case to the parties at their respective addresses and telecopy numbers set forth below, or to such other address of which any party may notify the other party in writing.

If to Buyer, to

Phillips-Van Heusen Corporation 200 Madison Avenue New York, New York 10016 Attention: Chief Executive Officer

Phone: (212) 381-3501 Fax: (212) 381-3970

With a copy to:

Phillips-Van Heusen Corporation 200 Madison Avenue New York, New York 10016 Attention: General Counsel Phone: (212) 381-3509 Fax: (212) 381-3970

If to Sellers, to

Cluett, Peabody & Co., Inc. 48 West 38th Street New York, New York 10018

Attention: Chief Executive Officer Phone: (212) 984-8915

Fax: (212) 984-8925

with copies to:

Vestar Capital Partners 245 Park Avenue 41st Floor New York, New York 10167 Attention: General Counsel Phone: (212) 351-1630 Fax: (212) 808-4922

and

Simpson Thacher & Bartlett 425 Lexington Avenue New York, New York 10017 Attention: Peter J. Gordon

Phone: (212) 455-2605 Fax: (212) 455-2502

ARTICLE 15.

MISCELLANEOUS

- 15.1 BINDING EFFECT; ASSIGNMENT. This Agreement shall be binding upon, and inure to the benefit of, all the parties to this Agreement and their respective successors, legal representatives and assigns permitted in accordance with this Section 15.1. Nothing herein shall create or be deemed to create any third party beneficiary rights in any person or entity not a party hereto. No assignment of this Agreement or of any rights or obligations hereunder may be made by any party (by operation of law or otherwise) without the prior written consent of the other parties to this Agreement, and any attempted assignment without the required consents shall be void; provided, however, that no such consent shall be required for Buyer to assign part or all of its rights and obligations under this Agreement prior to the Closing to one or more subsidiaries of Buyer, or after the Closing to any third party.
- 15.2 EXHIBITS AND SCHEDULES. All Exhibits and Schedules attached hereto and the documents and agreements referred to herein to be delivered and the acts to be performed at or subsequent to the Closing (collectively, the "ITEMS") are incorporated herein and expressly made a part of this Agreement as fully as though completely set forth herein.
- 15.3 SPECIFIC PERFORMANCE. Sellers and Buyer acknowledge that the other will have no adequate remedy at law if it fails to perform any of its obligations under this Agreement and the Ancillary Agreements. In such event, the performing party shall have the right, in addition to any other rights it may have, to seek specific performance of this Agreement and the Ancillary Agreements and/or injunctive relief.
- 15.4 COUNTERPARTS. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all, of the parties to this Agreement. In pleading or proving any provision of this Agreement, it shall not be necessary to produce more than one such counterpart.
- 15.5 HEADINGS; INTERPRETATION. (a) The headings contained in this Agreement are inserted for convenience of reference only and shall not otherwise affect the meaning or interpretation or be deemed a substantive part of this Agreement.
- (b) Except to the extent that the context otherwise requires "include," "includes" and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of like import.
- (c) The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement will refer to this Agreement as a whole, including the Exhibits and

Schedules attached hereto, and not to any particular provision of this Agreement, and section and subsection references are to this Agreement unless otherwise specified.

- (d) The meanings given to terms defined herein will be equally applicable to both the singular and plural forms of such terms.
- (e) All matters to be agreed to by any party hereunder must be agreed to in writing by such party unless otherwise indicated herein.
- 15.6 WAIVER. The failure of any party at any time or times to enforce or require performance of any provision hereof shall in no way operate as a waiver or affect the right of such party at a later time to enforce the same. No waiver by any party of any condition or the breach of any term, covenant, representation or warranty contained in this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such condition or breach, or a waiver of any other condition or of any term, covenant, representation or warranty contained in this Agreement. Any agreement on the part of a party hereto to a waiver shall be valid only if set forth in an instrument in writing signed by such party.
- 15.7 SEVERABILITY. If any provision of this Agreement shall hereafter be held to be invalid or unenforceable for any reason, that provision shall be reformed to the maximum extent permitted to preserve the original intent of the parties, failing which, it shall be severed from this Agreement with the balance of this Agreement continuing in full force and effect. Such occurrence shall not have the effect of rendering the provision in question invalid in any other jurisdiction or in any other case or circumstances, or of rendering invalid any other provisions contained herein to the extent that such other provisions are not themselves actually in conflict with any applicable law.
- 15.8 GOVERNING LAW AND FORUM. This Agreement and the Ancillary Agreements and the rights and duties of the parties hereunder and thereunder shall be governed by, and construed in accordance with, the law of the State of New York applicable to contracts made and to be performed therein. Each of the parties waives any objection that it may have to the venue of any suit, action or proceeding with respect to this Agreement and the Ancillary Agreements or the transactions contemplated hereby or thereby in the courts of the State of New York or the courts of the United States of America, in each case located in the Borough of Manhattan, City of New York and State of New York, or that such suit, action or proceeding brought in the courts of the State of New York or the courts of the United States of America, in each case located in the Borough of Manhattan, City of New York and State of New York, was brought in an inconvenient court and agrees not to plead or claim the same. Each of the parties agrees to service of process by registered mail, return receipt requested, or by any other manner provided by New York law.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their duly authorized representatives as of the day and year first above written.

CLUETT, PEABODY & CO., INC.

By: /s/ BRYAN P. MARSAL

Name: Bryan Marsal Title: Chairman

BIDERMANN TAILORED CLOTHING INC.

By: /s/ BRYAN P. MARSAL

Name: Bryan Marsal Title: Chairman

CLUETT DESIGNER GROUP INC.

By: /s/ BRYAN P. MARSAL

Name: Bryan Marsal Title: Chairman

PHILLIPS-VAN HEUSEN CORPORATION

By: /s/ MARK D. FISCHER

Name: Mark D. Fischer Title: Vice President

GUARANTEE

Cluett American Corp., a Delaware corporation ("CAC"), hereby irrevocably and unconditionally guarantees the payment of any amount payable by either of Bidermann Tailored Clothing Inc., a Delaware corporation ("BTC"), or Cluett, Peabody & Co., Inc., a Delaware corporation ("CP"), under the Share Purchase Agreement (defined below) to Phillips-Van Heusen Corporation, a Delaware corporation ("BUYER"), which is not paid by BTC or CP when due. For purposes of this Guarantee, the "SHARE PURCHASE AGREEMENT" refers to the Purchase and Sale Agreement dated as of June 28, 2000 among Cluett Designer Group Inc., BTC, CP and Buyer. Notwithstanding the foregoing, if (i) CAC and its affiliates cease to be affiliates of CDG, (ii) the

entity that controls CDG upon the occurrence of the event described in clause (i) executes and delivers to Buyer a guarantee identical in all material respects to this Guarantee (the "SUBSTITUTE GUARANTEE") and (iii) the entity referred to in clause (ii) is not less creditworthy than CAC, Buyer shall accept the Substitute Guarantee and this Guarantee thereafter shall have no force or effect.

CLUETT AMERICAN CORP.

By: /s/ BRYAN P. MARSAL -----

Name: Bryan Marsal Title: Chairman

PHILLIPS-VAN HEUSEN, CLUETT AMERICAN COMPLETE TRANSACTIONS

NEW YORK -- (July 24, 2000) -- Cluett American Corp. said today that it has completed the licensing of its U.S. Arrow trademark for men's and boys' dress shirts and sportswear to the Phillips-Van Heusen Corporation (NYSE: PVH). Cluett also said that it completed the sale of its wholly-owned subsidiary, the Cluett Designer Group Inc., to Phillips- Van Heusen. This unit is the domestic licensee for the design, sourcing and distribution of dress shirts bearing the Kenneth Cole trademark.

The two transactions were initially announced on June 13, 2000. Cluett currently licenses the Arrow brand name to 33 separate licensees in 75 countries worldwide.

Cluett American Corp. is a leading designer, manufacturer and marketer of men's, women's and children's socks under the company-owned Gold Toe brand, as well as licensed brands Nautica, Jockey and North Face in the United States. The company is also the domestic licensee and manufacturer of Kenneth Cole men's socks and is the worldwide licensor/brand manager for the Arrow trademark.

Phillips-Van Heusen Corporation is one of the leading apparel and footwear companies in the world. Its roster of nationally recognized brands includes Van Heusen, Izod, G.H. Bass & Co., Geoffrey Beene and now Arrow. In addition, the company licenses DKNY, FUBU, Regis by The Van Heusen Company, John Henry and Manhattan for dress shirts.

###

Contacts: Emanuel Chirico

Executive Vice President & Chief Financial Officer

Phillips-Van Heusen Corporation

(212) 381-3503 www.pvh.com

Bryan P. Marsal

Chairman and Chief Executive Officer

Cluett American Corp.

(212) 984-8915